CANADIAN THESES ON MICROFICHE

THÈSES CANADIENNES SUR MICROFICHE

∎�

National Library of Canada Collections Development Branch

Canadian Theses on Microfiche Service

Ottawa, Canada K1A 0N4 Bibliothèque nationale du Canada Direction du développement des collections

Service des thèses canadiennes sur microfiche

NOTICE

The quality of this microfiche is heavily dependent upon the quality of the original thesis submitted for microfilming. Every seffort has been made to ensure the highest quality of reproduction possible.

If pages are missing, contact the university which granted the degree

Some pages may have indistinct print especially if the original pages were typed with a poor typewriter ribbon or if the university sent us an inferior photocopy

Previously copyrighted materials (journal articles, published tests, etc.) are not filmed,

Reproduction in full or in part of this film is governed by the Canadian Copyright Act, R.S.C. 1970, c. C-30. Please read the authorization forms which accompany this thesis.

\cdot AVIS

La qualité de cette microfiche dépend grandement de la qualité de la thèse soumise au microfilmage. Nous avons tout fait pour assurer une qualité supérieure de reproduction.

S'il manque des pages, veuillez communiquer avec l'université qui a conféré le grade

La qualité d'impression de certaines pages peut laisser à désirer, surtout si les pages originales ont été dactylographiées à l'aide d'un ruban usé ou si l'université nous a fait parvenir une photocopie de qualité inférieure.

Les documents qui font déjà l'objet d'un droit d'auteur (articles de revue, examens publiés, etc.) ne sont pas microfilmés.

La reproduction, même partielle, de ce microfilm est soumise à la Loi canadienne sur le droit d'auteur, SRC 1970, c. C-30. Veuillez prendre connaissance des formules d'autorisation qui accompagnent cette thèse.

THIS DISSERTATION HAS BEEN MICROFILMED EXACTLY AS RECEIVED

 LA THÈSE A ÉTÉ MICROFILMÉE TELLE QUE NOUS L'AVONS REÇUE



NO 11 12 Bibliothèque nationale **National Library** du Canada of Canada Canadian Theses Division Division des thèses canadiennes Ottawa, Canada K1A 0N4 67377 PERMISSION TO MICROFILM - AUTORISATION DE MICROFILMER Please print or type -- Écrire en lettres moulées ou dactylographier Full Name of Author - Nom complet de l'auteur TOSEPH WAL HUNG SO Date of Birth – Date de naissance Country of Bifth - Lieu de naissance Le of Birth - Date de naissance MAY 28,1933 rmanent Address - Résidence fixe DEPAR 7 MENT OF MAINENTING, UNIVERSITY OF ALBERTA EPMONTUN, ALBERTA, (ANADA, ICG 261 1 Permanent Address — Résidence fixe Title of Thesis - Titre de la thèse A SIUDY OF SOME OAC- AND TWO- LUCI PREPATOR - PRLY INTERACTION MODELS University - Université UNIVERSITY OF ALBERTA Degree for which thesis was presented Grade pour lequel cette these fut presentee Ph. D Name of Supervisor - Nom du directeur de thèse Year this degree conferred -- Année d'obtention de ce grade DR HERBERT I FREEDMAN 1984 L'autorisation est, par la présente, accordée à la BIBLIOTHÈ-Permission is hereby granted to the NATIONAL LIBRARY OF QUE NATIONALE DU CANADA de microfilmer cette thèse et de CANADA to microfilm this thesis and to lend or sell copies of prêter ou de vendre des exemplaires du film the film L'auteur se réserve les autres droits de publication; ni la thèse The author reserves other publication rights, and neither the ni de longs extraits de celle-ci ne doivent être imprimés ou thesis nor extensive extracts from it may be printed or otherautrement reproduits sans l'autorisation écrite de l'auteur. wise reproduced without the author's written permission

		``	·
Date		Signature	
	OCTOBER 11, 1984	Aordai He	t

THE UNIVERSITY OF ALBERTA

A STUDY ON SOME ONE- AND TWO- LOCI PREDATOR-PREY INTERACTION MODELS

BΥ

JOSEPH WAI HUNG SO

A THESIS

SUBMITTED TO THE FACULTY OF GRADUATE STUDIES AND RESEARCH IN PARTIAL FULFILLMENT OF THE REQUIREMENTS FOR THE DEGREE OF DOCTOR OF PHILOSOPHY

DEPARTMENT OF MATHEMATICS

EDMONTON, ALBERTA -

FALL, 1984

THE UNIVERSITY OF ALBERTA

RELEASE FORM

NAME OF AUTHOR: TITLE OF THESIS:

JOSEPH WAI HUNG SO A STUDY OF SOME ONE- AND TWO- LOCI PREDATOR-PREY INTERACTION MODELS

DEGREE FOR WHICH THESIS PRESENTED: DECTOR OF PHILOSOPHY YEAR THIS DEGREE GRANTED: 1984

Permission is hereby granted to THE UNIVERSITY OF ALBERTA LIBRARY to reproduce single copies of this thesis and to lend or sell such copies for private, scholarly or scientific research purposes only.

The author reserves other publication rights, and neither the thesis nor extensive extracts from it may be printed or otherwise reproduced without the author's written permission.

(Signed) . . Ao Wai Hung

PERMANENT ADDRESS: . Yo Dept. of Math . Univ of Alberta Edmonton . . 766261

DATED: October 11, 1984

Κ.

THE UNIVERISTY OF ALBERTA

FACULTY OF GRADUATE STUDIES AND RESEARCH

The undersigned certify that they have read, and recommend to the Faculty of Graduate Studies and Research, for acceptance, a thesis entitled A STUDY ON SOME ONE- AND TWO- LOCI PREDATOR-PREY INTERACTION MODELS submitted by JOSEPH WAI HUNG SO in partial fulfilment of the requirements for the degree of Doctor of Philosophy.

· Lige

Supervisor to 7 Millar

. J. Butler.

K. P. Hadily

External Examiner

Date . Oct 1/84



iv

ABSTRACT

Four continuous models of population genetics with nonlinear birth and death processes are considered. The first model is concerned with a one-locus, two-allele problem in which fertility / viability is incorporated. Global dynamics of this three population system is The second model is a predator-prey interaction model in obtained. which the prey population is subdivided into three genotypes and its dynamics is given by the first model. The trade-offs between the fertility/viability of the prey genotypes and their predation functional responses relating to questions of convergence to pure strains and persistence are discussed. The third and fourth models are the analogues of the first two models but for a two-locus, two-allele problem and with no fertility / viability differences. Global convergence to the Hardy-Weinberg manifold is shown for the third model. Some results relating to the question of convergence to pure strains as well as to the question of persistence are obtained for the fourth model.

ACKNOWLEDGEMENTS

 e_{2}

Ś

I wish to express my deep gratitude to my thesis supervisor, Herb Freedman, for his guidance and assistance throughout this research.

۱

I would also like to thank Geoff Butler, Curt Strobeck and Paul Waltman for valuable discussions at various stages of this research.

Financial assistance was provided in part by NSERC, grant number A4823 (Principle investigator: Dr. H.I. Freedman).

Ŋ

C

-

53

TABLE OF CONTENTS

Ð

A

\$

د	••			()												<u>P</u>	age,
DEDICATIC	DN .		•••	, .	•	•	•	, * .	•	•		•	•	•	•	•	iv
ABSTRACT	• •		•••		•	•	•	÷		•	•	•	•	•	•	•	v
ACKNOWLED	GEMENT	. 1			•	•	•	•	•	•		•	•	•	• .	•	vi
							,- ,		[.]								
CHAPTĘR) ~				-								•
1	INTRO	DUCTION	•	_) • •	•	•		•	•	•	•	•	•	•	•	•]
2	MATHEN	ATICAL	PREL	IMINA	RIES	•	•	•	•		•	•	•	•	•	•	5
	2.1	Standa	rd Re	sults	•	•	•	• 、	•		•	•	•	•	•	•	5
	2.2	Conven	tions	and	Noṫa	tio	ns	•	•	•	•	•		•	•	•	6
	2.3	Positi	ve In	varia	ince	•	•	•	•	•	•	•	•	•	•	•	8
	2.4	Bounde	dneşs	•••		•	•	•	•	•	•	•	•	•	•	•	10
	2.5	Persis	tence	• •	•	•	•	•	۴ •.	•	•	•	•	•	•	•	13
	2.6	Wazews	ki Se	ts .	•	•	•	•		•	•	•	•	•	•	•	14
	2.7	Chain	Recur	rence	· ·	•	•	•	•	•	•	•	• .	•	•	•	16
	2.8	Invari	ant M	anifo	olds			•		•	•	•	•	•	•	٠	17
	2.9	Linear	izati	on Th	neore	ms	•	•		•	•	•	•	•	•	•	21 .
	2.10	Sadd1e	s Nod	es.	•		•	• .		•	•	•	•	•	•	•	24
	2.11	Omega-	Limit	Sets	5 ;	•	•	•	•	•	•	•	•	•	•	•	27
3	A ONE-	-LOCUS							, СЕС		`						31
		FERTIL	111/	A PART		וט	FFE	KEN	ίĽΟ	•	•	•	• .	•	•	•	51
ł	3.1	Introd	uctio	n anc	1 the	Мо	del	•	٠	•	•	•	•	•	•	•	31
	3.2	The As	socia	ted F	low	•	•	•	•	•	•	•	•	•	•	•	35
	3.3	The Re	st Po	int S	Set.	•	•	•	•	•	•	•	•	•	•	•	37
	3.4	Projec	tive	Flow		•	•	•	.•	•	•	•	•	•	•	•	41
	3.5	Global	Dyna	mics.	•	•	•	•	•	•	•	•	•	•	•	<i>"</i> •	50 _j ,
	3.6	Discus	sion	• •	•	•	•	·	÷	•	•	•	•	•	•	•	60 ·

vii

ÇHAPTER

ł

À

(?)

G

8

١

Page

1						*	
(4	A PREI	DATOR-PREY MODEL CONSISTING OF THREE PREY	-1			
, ۱		964 14	GENOTYPES WITH FERTILITY / VIABILITY DIFFERENC	E,	•	•	62
		4.1	Introduction and the Model	•	•	•	62
		4.2	The Associated Flow	•	•	•	64
		4.3	Boundary Rest Points and Their Stability .	•	•	•	67
		4.4	Evolution to Pure Strains	•	•	•	73
		4.5	Persistence and Non-Persistence Results .	•	•	•	79
		4.6	Discussion.	•	• ^ ħ	•	82
ţ							.
	5	Α ΤΨΟ	-LOCUS TWO-ALLELE MODEL	•	•	•	84
$\langle \cdot \rangle$		5.1	Introduction and the Model	• -	•	•	84
		5.2	The Associated Flow	•	•	•	87
Ĵ		5.3	The Rest Point Set	•	o	•	89
		5.4	Global Convergence to the Hardy-Weinberg Mani	fol	d.	•	94
		5.5	Discussion	•	•	•	103
	6		DATOR-PREY MODEL IN WHICH THE PREY POPULATION	ĪS			
	0		SUBDIVIDED INTO NINE GENOTYPES CORRESPONDING	Τ0	А		
			TWO-LOCUS TWO-ALLELE PROBLEM	•	•	• ,	105
	r				_	Ç	105
		6.1	Introduction and the Model	•	•	•	107
		6.2	The Associated Flow	• v	•	•	110
		6.3	Some Boundary Rest Points and Their States	y •	•	•	
		6.4	Some Conditions Which Lead to Evolution to				116
			Pure Strains	•	•	•	119
		6.5	Some Fersisterce Results	•	•	• •	121
		6.6	Discussion	•	•	•	121
	7	∕⊂CON¢C	_UDING DISCUSSION AND REMARKS	•	•	•	122
<i>6</i> :				_		•	132
	BIBLIO	RAPHY	· · · · · · · · · · · · · · · ·	•	•	L	
*	APPENDI	X 1.		٠	•.	•	138
	APPEND	IX 2.		,	•	•	140

viii

CHAPTER 1

INTRODUCTION

Population genetics is the study of how genetic principles apply to a population and is principally concerned with the genetic and phenotypic properties of the members of a population. Ecology, on the other hand, is more concerned with population size and discribution as well as population interactions such as competition, predation and mutualism. Even though these two areas of population biology are closely related, most mathematical studies tend to specialize more in one of these areas than another. The purpose of this dissertation is to study a number of mathematical models which both the genetic and the ecological components are present.

The usual setting for problems in population generation difference equations rather than differential equations. Difference equations in population ecology appropriate for problems of non-overlapping generations. When organisms reproduce continuously the differential equation approach seems more appropriate. Since environmental factors act continuously, it seems more appropriate to integrate the genetic and ecological components in a continuous formulation. This is the approach undertaken in this thesis.

Even among continuous models, it is convenient, for our purpose, to classify them into three types. First, there are the pure genetic models, representatives of which are the ones studied in Hadeler-Liberman (1975), Butler-Freedman-Waltman (1982), and, Hadeler-Glas (1983). In these models,

the birth and death rates are taken to be constants. In some cases the death rate is taken to be zero, and it is more natural to talk about frequency of a genotype rather than the number of organisms of that genotype, since the number of organisms grows unboundedly. It should be pointed out that many of the continuous models in population genetics are of this type. Among them two famous ones are the Fisher-Wright-Haldane model for one locus problems (see Crow-Kimura (1970), Akin-Hofbauer (1982), and, Losert-Akin (1983)) and the Kimura model for two loci problems (see Crow-Kimmura (1970), Kimura (1958), and, Akin (1979, 1983)). The second class of models is the ones that incorporate nonlinear birth and death processes. Representatives of this class are: Freedman-Waltman (1978, 1982), and, Beck (1982). In this class of models the birth and the death functions are singled out, and the coefficient associated with the birth function determines the mating structure and selection. The third class of models is the ones that incorporate environmental effects, as well as nonlinear birth and death processes. Representatives of this class are: Freedman-Waltman (1978, 1982), Beck-Keener-Ricciardi (1982, 1984), and, Beck (1984).

The models (3.1) and (5.1) to be studied in this thesis belong to the second class and models (4.1) and (6.1) belong to the third class. These models attempt to extend the model studied in Freedman-Waltman (1978, 1982) to include fertility / viability differences and to two loci problems. It turns out that even for the case studied in Freedman-Waltman (1978, 1982), we are able to generalize and unify their results as one single theorem (Theorem 4.6).

The genetic component of all the models studied in the papers mentioned under class 2 and 3 can be derived from the models of Nagylaki-

٠2

Ś

Crow (1974). This derivation is carried out in Waltman (1984). The model (3.1) (one-locus, two-allele with fertility / viability differences) and (5.1) (two-locus, two-allele with equal fertility / viability) studied in this thesis are extensions of these models. A different derivation can also be found in Appendix 1 and 2. The main theme for models (3.1) and (5.1) is global convergence to equilibrium. The recent work of Akin mentioned previously shows that even though global convergence to equilibrium is to be expected for one locus problems, periodic solutions (through Hopf bifurcation) are possible in the two loci problems if selection and recombination are incorporated.

In model (4.1) (resp. (6.1)); an ecological component in the form of a predator is added to model (3.1) (resp. (5.1)). Thus in model (4.1) (resp. (6.1)) the prey population is modelled by (3.1) (resp. (5.1)). To model the predator-prey interaction, the standard generalized model of Gause (1934) for predator-prey interactions is employed. There is a vast literature on the generalized Gause model. Of particular importance to the investigations we are undertaking in this thesis are: the local stability criteria of Rosenzweig-MacArthur (1963) and Gause-Smaragdova-Witt (1936), the phase-plane analysis of Freedman (1976), and the global stability criteria of Hsu (1978) and Cheng-Hsu-Lin (1981). For model (4.1), there are two types of selection forces at work. On the one hand, there is the selection due to fertility/viability differences, and, on the other hand, there is the selection due to differential predation functional responses. The main theme here is to determine the interplay of these two selection forces in relation to the questions of persistence and non-persistence. It should be pointed out that the recent work of Freedman-So-Waltman (1984) shows that even in the absence of fertility/viability differences, it is possible for model (4.1) to possess periodic solutions. These periodic solutions are, of course, predator mediated. In model (6.1), we attempt to discover some simplarities and differences between the one locus and the two loci theories in relation to the question of persistence and non-perfistence.

The dissertation is organized as follows. In Chapter 2, a number of mathematical results used in the main body of this thesis will be described. Proofs are provided for those results that appear to be new. In the next four chapters, four continuous models in population genetics are discussed. Chapter 3 is concerned with a one-locus, <code>_two-allele model with nonl-inear birth and death processes as well as fer-</code> tility/viability differences. It is shown that the dynamics of this model is trivial (that is all solutions converge to some equilibrium). In Chapter 4, the model studied in Chapter 3 is extended to include a predator. We obtain conditions under which only one of the prey gamete types survives as well as conditions under which the system persists. Of particular interest and importance is the interplay between the fertility / viability and the predator functional responses of the prey genotypes. In Chapter 5, the model studied in Chapter 3 is extended to the two-locus, two-allele case but with no fertility / viability differences. Again the dynamics of this model is shown to be trivial and the analogue of the Hardy-Weinberg equilibrium relation for discrete models is obtained. In Chapter 6, the model studied in Chapter 5 is extended to include a preda-Same questions as in Chapter 4 are raised. It is shown that the tor. predator functional response of the double heterozygote can change a nonpersistent system (when considered as one-locus problems) to that of persistence. Finally, a concluding discussion is given in Chapter 7.

CHAPTER 2

MATHEMATICAL PRELIMINARIES

In this chapter, a number of concepts and results from the theory of autonomous differential equations and dynamical systems will be discussed. Some of the following sections also contain a number of simple but useful technical results which are new.

2.1. Standard Results.

The portion of the mathematical theory which is considered as standard or well-known will be listed below along with references where they can be found.

- (i) Existence, uniqueness and continuous dependence on initial conditions and parameters theorem: Chapter 1 of Coddington -Levinson (1955).
- (ii) Differential inequalities: Section I.6 of Hale (1969) and
 Section I.4 of Coppel (1965).
- (iii) Flows, local flows and semi-flows: Chapter 2 of LaSalle (1976),
 Chapter II of Sell (1971), and, Chapter 1 of Bhatia-Szegö (1967).
- (iv) Deriving a flow from an autonomous differential equation:Chapter 2 of LaSalle (1976) and Section 1.7 of Hale (1969).

Ø

(v) Poincaré - Bendixson theory and the theory of index for planar
 systems: Chapter 16 of Coddington - Levinson (1955) and Chapter
 VII of Hartman (1982).

- (vi) Omega limit sets, alpha limit sets and their well-known properties: Section I.8 of Hale (1969) and Chapter 2 of LaSalle (1976).
- (vii) Liapure and global stability for sets: Chapter 2 of LaSalle (1976) and Chapters 1 and 2 of Bhatia - Szegö (1967).
- (viii) Flow equivalence and C^k equivalence: Chapter 2 of Irwin (1980).

2.2. Conventions and Notations.

In this section, some conventions and notations that are used in the main body of the thesis will be discussed.

A. Conventions.

- (i) In all our applications, the flows are defined by autonomous differential equations on subsets of R^n .
- (ii) By boundedness of an orbit (or solution) is meant boundedness for all positive time.

6

Ľ

1

(i) R_{+}^{n} = the positive cone in R^{n} (ii) $cl(R_{+}^{n})$ = the closure of R_{+}^{n} in R^{n} = the non-negative cone in R^{n}

(iii)
$$b(R_+^n) = the boundary of R_+^n in R^n$$

(iv) For $1 \le i_1 < \cdots < i_k \le n$, let $I = \{i_1, \cdots, i_k\}$ and $I' = \{1, \cdots, n\} \setminus I$.

$$\begin{split} & \bigwedge_{i_1}^{n}, \cdots, x_{i_k}^{n} &= \{x \in cl(\mathbb{R}^n_+) : x_i > 0 \text{ for } i \text{ in } I, \\ & x_i = 0 \text{ for } i \text{ in } I'\} \\ & cl(H_{x_{i_1}}, \cdots, x_{i_k}^{n}) = closure \text{ of } H_{x_{i_1}}, \cdots, x_{i_k}^{n} \text{ in } \mathbb{R}^n \\ &= \{x \in cl(\mathbb{R}^n_+) : x_i = 0 \text{ for } i \text{ in } I'\} . \end{split}$$

- (v) For x in \mathbb{R}^n and $1 \le i \le n$, x_i or $(x)_i$ denotes the i^{th} component of x.
- (vi) d denotes the Euclidean metric on Rⁿ or any given metric of a metric space.
- (vii) Solutions of a differential equation are sometimes denoted by x_0 .

١

-11

(viii) If a flow (X,ϕ) comes from a differential equation

 $\dot{\mathbf{x}} = \mathbf{f}(\mathbf{x})$ $(\mathbf{x} \in \mathbb{R}^n)$

and there is a need to emphasize f, then ϕ is also denoted by ϕ_f .

(ix) Lip denotes the class of locally Lipschitz continuous functions.

(x) w(x) denotes the omega - limit set of x_{∞} .

(xi) O(x) (resp. $O^+(x)$) denotes the orbit through x (resp. the non-negative semi-orbit through x).

2.3. Positive Invariance.

Consider the differential equation

 $\dot{\mathbf{x}} = \mathbf{f}(\mathbf{x})$ $(\mathbf{x} \in \mathbf{R}^n)$

where $f \in Lip$. If this system is to model interactions of populations, we should expect $cl(R_+^n)$ to be positively invariant. A necessary and sufficient condition for this is given by the following proposition.

Proposition 2.1: $cl(R_{+}^{n})$ is positively invariant under the flow defined by f if and only if for all $1 \le i \le n$, we have,

 $f_{i}(x_{1}, \dots, x_{i-1}, 0, x_{i+1}, \dots, x_{n}) \geq 0$

for all
$$(x_1, \dots, x_{i-1}, 0, x_{i+1}, \dots, x_n)$$
 in $cl(R_+^n)$.

<u>Proof</u>: First suppose there exists $1 \le i_0 \le n$ and $x^* = (x_1^*, \cdots, x_{i_0-1}^*, 0, x_{i_0+1}^*, \cdots, x_n^*)$ in $cl(R_+^n)$ such that $f_{i_0}(x^*) < 0$. Let x(t) be the solution such that $x(0) = x^*$. Then $x_{i_0}(t)$ is strictly decreasing for t small enough. Hence, $x_{i_0}(t) < 0$ for t > 0 small enough, contradicting the positive invariance of $cl(R_+^n)$.

Now consider the sequence of differential equations

 $\dot{x} = f(x) + \frac{1}{N} = f^{N}(x)$, for N = 1,2,...

Since $f_i^N(x_1, \dots, x_{i-1}, 0, x_{i+1}, \dots, x_n) > 0$ for all $(x_1, \dots, x_{i-1}, 0, x_{i+1}, \dots, x_n)$ in $cl(R_+^n)$ and $1 \le i \le n$, by looking at the flow on $b(R_+^n)$ and using an argument similar to that above, we can show that $cl(R_+^n)$ is positively invariant under the flow ϕ_N defined by f^N for all $f^N = 1, 2, \dots$. Next, we suppose that $cl(R_+^n)$ is not positively invariant under the flow ϕ_f defined by f. Then there exists an $1 \le i_0 \le n$, a $t^* > 0$ and an $x^* = (x_1^*, \dots, x_{i_0-1}^*, 0, x_{i_0+1}^*, \dots, x_n^*)$ in $cl(R_+^n)$ such that

 $d(\phi_{f}(x^{*},t^{*}), cl(R^{n}_{+})) = d > 0$.

Now, by continuous dependence on parameters, (see Theorem 7.4 on p. 29 of Coddington-Levinson (1955)),

 $\phi_{f^{N}}(x^{\star},t^{\star})$ converges to $\phi_{f}(x^{\star},t^{\star})$ as $N \rightarrow +\infty$.

But this leads to a contradiction because

ď

$$d(\phi_{f^N}(x^*,t^*),cl(R^n_+)) = 0$$
 for all $N = 1,2,\cdots$

completing the proof of the proposition.

2.4. Boundedness.

Let (X, φ) be a semi-flow and let $A \subset X$.

Definition: (X,ϕ) is said to be <u>A-dissipative</u> if

(i) A is compact and positively invariant, and,

q

(ii) for all $x \in X$, $\phi(x,t) \rightarrow A$ as $t \rightarrow +\infty$.

<u>Remark</u>: (X,ϕ) is A-dissipative implies that all solutions are bounded.

The following is a situation which is applicable to all the models we study here and for which we can show the flow is A-dissipative.

$$\dot{x}_{i} = f_{i}(x_{1}, \dots, x_{n}, y)$$
 (i = 1, ..., n)
(2.1)

 $y = g(x_1, \dots, x_n, y)$

where $f_i, g \in Lip$ and $cl(R_+^{n+1})$ is positively invariant. Let $x = x_1 + \cdots + x_n$ and suppose that

q.e.d.

$$(2.2)$$

$$kx + y \leq kf(x) - sy$$
for some k,s > 0, where f \in Lip satisfies: f(0) = 0, f'(0) > 0,
there exists a unique K > 0 such that f(K) = 0, and, f'(K) < 0.
Proposition 2.2: Let A = {(x₁,...,x_n,y) \in cl(R₊ⁿ⁺¹) : x \leq K and
 $kx + y \leq \frac{k}{s}$ (M+sK)} where M = max {f(x) : x \in [0,K]}. Then system
(2.1) when considered as a flow on cl(R₊ⁿ⁺¹) is A-dissipative.

 $\dot{x} \leq f(x)$

ł

Ĩ

11

Ć

<u>Proof</u>: Consider the scalar differential equation $\dot{z} = f(z)$. By the properties of f, we have:

(i) if $z_0 \in (0,K]$, then $z(t) \rightarrow K$ as $t \rightarrow +\infty$, and,

(ii) if $z_0 \in [K, +\infty)$, then $z(t) \rightarrow K$ as $t \rightarrow +\infty$.

Using a standard differential inequality argument (see Theorem 6.1 on p. 31 of Hale (1969)) and the fact that $cl(R_+^{n+1})$ is positively invariant, we can show that

(iii) if
$$x_0 \in [0,K]$$
, then $x(t) \in [0,K]$ for all $t \ge 0$, and,

(iv) if
$$x_0 \in cl(R_+^1)$$
, then $\overline{\lim_{t \to +\infty} x(t)} \leq K$.

From (2.2), we have,

(v) $kx + y \le -s(kx + y) + k(f(x) + sx)$.

We first show that A is positively invariant. For if we let $(x_{10}, \dots, x_{n0}, y_0) \in A$, then by (iii), $x(t) \leq K$ for all $t \geq 0$. On the other hand, by (v), we have,

$$kx(t) + y(t) \le e^{-st}(kx_0 + y_0) + ke^{-st}(M + sK) \int_0^t e^{s\tau} d\tau$$
$$= \frac{k}{s} (M + sK) - e^{-st}(\frac{k}{s} (M + sK) - (kx_0 + y_0)) .$$

Therefore, if $kx_0 + y_0 \le \frac{k}{s} (M + sK)$, then

$$kx(t) + y(t) \le \frac{k}{s} (M + sK)$$
 for all $t \ge 0$.

We now show that for all $(x_{10}, \dots, x_{n0}, y_0) \in cl(R_+^{n+1})$,

$$(x_1(t), \dots, x_n(t), y(t)) \rightarrow A$$
 as $t \rightarrow +\infty$.

By (iv), it remains to prove that for all $\epsilon>0$, there exists T>0 - such that

 $kx(t) + y(t) \leq \frac{k}{s} (M + sK) + \varepsilon$ for all $t \geq T$.

By (iv), there exists $T_1 > 0$ such that $x(t) \le K + \frac{\varepsilon}{3k}$ for all $t \ge T_1$. Let $K' = \max \{x(t) : t \in [0,T_1]\}$. By (v), for all $t \ge T_1$ we have

$$\begin{split} kx(t) + y(t) &\leq e^{-st}(kx_0 + y_0) + ke^{-st} \int_0^{T_1} [f(x(\tau)) + sx(\tau)] e^{s\tau} d\tau \\ &+ ke^{-st} \int_{T_1}^{t} [f(x(\tau)) + sx(\tau)] e^{s\tau} d\tau \end{split}$$

12

Ê

Ê

Since $f(x) \leq M$ for all $x \in cl(R_+^1)$, second term $\leq \frac{k}{s}e^{-s(t-T_1)}(M+sK')$, and third term $\leq \frac{k}{s}(M+sK) + \frac{\varepsilon}{3}$. We can choose $T \geq T_1$ so that for all $t \geq T$, the first and second terms are $<\frac{\varepsilon}{3}$. Then for all $t \geq T$,

$$kx(t) + y(t) \leq \frac{k}{s} (M + sK) + \varepsilon$$

as desired.

q.e.d.

<u>Remark</u>: The above proposition also shows that solutions can be continued for all positive time and hence system (2.1) defines a semi-flow on $cl(R_{+}^{n+1})$.

Ô

2.5. Persistence.

Consider

$$\dot{x} = f(x) \quad (x \in R^n)$$
(2.3)

where x_i (i = 1,...,n) can be thought of as the number/density/biomass of the organisms in the ith population and that the interaction within and between the populations is described by system (2.3). Assume that $cl(R_+^n)$ is positively invariant and all solutions initiating from $cl(R_+^n)$ can be continued for all positive time. <u>Definition</u>: (i) System (2.3) is said to be <u>persistent</u> if for all $x \sim$ in R_{+}^{n} , $\frac{\lim_{t \to +\infty}}{t \to +\infty} x_{i}(t) > 0$ for all $i = 1, \dots, n$.

(ii) System (2.3) is said to be <u>non-persistent</u> if it is not persistent. That is, there exists an $x_0 \in R_+^n$ and i_0 $(1 \le i_0 \le n)$ such that $\lim_{t \to +\infty} x_{i_0}(t) = 0$. Equivalently, the omega-limit set of x_0 has non-empty intersection with the boundary $b(R_+^n)$.

<u>Remarks</u>: (i) It should be pointed out that there are many mathematical definitions of the concept of persistence. For example, in McGehee - Armstrong (1977) the authors defined persistence to mean the existence of an attractor in R_{+}^{n} . The one we choose to use here is the one used in Freedman-Waltman (1984).

(ii) According to the above definition of non-persistence, system (2.3) will be non-persistent if there exists an asymptotically stable rest point on $b(R_+^n)$.

2.6. <u>Wazewski Sets</u>.

In this section the basic properties of a Wazewski set.will be discussed. More information can be found on pp. 24-25 of Conley (1978) and on Sections X.2 and X.3 of Hartman (1982).

Definition: Let (X,ϕ) be a continuous semi-tipw and let $A \subset X$.

(i) The eventual exit set, A^{0} , of A is defined as

 $A^{\circ} = \{x \in A : \phi(x,t) \notin A \text{ for some } t > 0\}$.

(ii) The immediate exit set, A^{-} , of A is defined as

$$A^{-} = \{x \in A : \phi(x, [0,t)) \notin A \text{ for all } t > 0\}$$

(iii) A is called a Wazewski set if it satisfies:

(a) A is flow closed, i.e., if $x \in A$ and $\phi(x, [0, \pm)) \subseteq A$, then $\phi(x, [0, \pm]) \subseteq A$, and,

(b) A^{-} is closed relative to A^{0} .

(iv) The time of exit map $T : A^0 \rightarrow [0, +\infty)$ is defined by

 $T(x) = \sup \{t \ge 0 : \phi(x,[0,t]) \subset A\}$

for all $x \in A^0$.

Theorem 2.3. (Wazewski): If A is a Wazewski set, then A⁻ is a strong deformation retract of A⁰ and A⁰ is open relative to A.

<u>Proof</u>: The statement of the theorem as given on p. 24 of Conley (1977) is for flows but the proof remains valid without modification in the case of semi-flows.

q.e.d.

Another useful property of Wazweski sets is the following.

E.S.

Corollary 2.4: If A is a Wazewski set, then the time of exit map is continuous.

<u>Proof</u>: This is contained in the proof of the theorem. In fact, conditions (a) and (b) are provided to ensure the time of exit map is upper semi-continuous and lower semi-continuous respectively.

q.e.d.

2.7 Chain Recurrence.

 $\{\cdot\}$

In this section the notion of chain recurrence, first used by C. Conley and R. Bowen will be introduced. (See pp. 36-38 of Conley (1978)). This concept turns out to be a very useful kind of recurrence from the application standpoint.

Let (X, ϕ) be a flow and let S be a compact (Hausdorff) invariant subset of X .

Definition: Let U be an open cover of S. Let $x_i, y \in S$ and t > 0. A (U,t)-chain from x to y means a sequence

 $x = x_1, \dots, x_{n+1} = y; t_1, \dots, t_n$ $(t_i \ge t)$

for all $i = 1, \dots, n$ and such that for each pair $(\phi(x_i, t), x_{i+1})$ $(i = 1, \dots, n)$ there is an element of the cover *U* containing both members of the pair. <u>Definition</u>: The <u>chain recurrent</u> set R(S) of $S \cdot is$ defined to be the set of all $x \in S$ such that for any choice of cover U of S and any t > 0 there exists a (U,t) - chain from x to x.

<u>Definition</u>: S is called <u>chain recurrent</u> if R(S) = S. It is called <u>strong gradient like</u> if R(S) is totally disconnected (and consequently is equal to its rest point set).

<u>Proposition 2.5</u>: Given $x \in S$. Then w(x), the omega-limit set of x, is chain recurrent.

Proof: (See p. 38 of Conley (1978).) -

q.e.d.

2.8. Invariant Manifolds.

In this section the invariant manifold theory for rest points will be discussed. Some general references for this topic are: Chapter IX of Hartman (1982) and Section 9.2 of Chow-Hale (1982).

<u>Definition</u>: Let (X,ϕ) be a flow on a smooth manifold X with metric d. (Think of X as an open subset of \mathbb{R}^n .) Given a rest point x^* of ϕ , the <u>stable set</u>, $W^S(x^*)$, of x^* is defined as the set

 $\{x \in X : d(\phi(x,t),x^*) \rightarrow 0 \text{ as } t \rightarrow +\infty\}.$

The <u>unstable set</u>, $W^{U}(x^{*})$, of x^{*} is defined analogously by reversing time t.

If the rest point x^* is hyperbolic (i.e. all eigenvalues of the linearization (Jacobian matrix) have non-zero real parts), we have the following stable manifold theorem.

Theorem 2.6 (global stable manifold theorem): If the flow (X,ϕ) is C^{k} $(k \ge 1)$, and x^{*} is a hyperbolic rest point, then $W^{S}(x^{*})$ (resp. $W^{u}(x^{*})$) are C^{k} immersed submanifolds of X, tangent at x^{*} to the generalized eigenspace corresponding to those eigenvalues of the linearization which have negative real parts (resp. positive real parts).

Proof: (See Theorem 6.17 on p. 152 of Irwin (1980).)

q.e.d.

 \Diamond

<u>Remark</u>: If x^* is hyperbolic, $W^{S}(x^*)$ (resp. $W^{U}(x^*)$) is also known as the stable manifold (resp. unstable manifold) of x^* .

When some of the eigenvalues of the linearization at x have zero real parts, there are a number of locally invariant manifolds through x*.

Theorem 2.7 (strong stable manifold theorem): Consider

 \bigcirc

$$\dot{\mathbf{x}} = \mathbf{A}\mathbf{x} + \mathbf{u}(\mathbf{x}) \qquad (\mathbf{x} \in \mathbf{R}^n)$$

where $u \in C^1$, u(0) = 0 and Du(0) = 0. If the n by n matrix A possesses s (s > 0) eigenvalues having negative real parts, then the set

$$W^{SS}(0) = \{x \in X : \frac{1}{1 + \infty} = \frac{1}{t} \log d(\phi(x,t),0)\} < +\infty \}$$

is a locally invariant C^1 manifold of dimension s .

Proof: (See Theorem 6.1 on p. 242 of Hartman (1982).)

q.e.d.

 \mathfrak{D}

<u>Definition</u>: The set $W^{SS}(0)$ is called the <u>strong stable manifold</u> of x = 0. The <u>strong unstable manifold</u>, $W^{SU}(0)$, can be defined in a similar manner.

V

Theorem 2.8 (center manifold theorem): Consider

$$\dot{\mathbf{x}} = \mathbf{A}\mathbf{x} + \mathbf{u}(\mathbf{x},\mathbf{y}) \qquad (\mathbf{x} \in \mathbf{R}^{\mathsf{m}})$$
$$\dot{\mathbf{y}} = \mathbf{B}\mathbf{y} + \mathbf{v}(\mathbf{x},\mathbf{y}) \qquad (\mathbf{y} \in \mathbf{R}^{\mathsf{n}})$$

<u>near</u> (x,y) = (0,0), where all the eigenvalues of A have zero real parts and all the eigenvalues of B have non-zero real parts, $u,v \in C^{k}$ $(k \ge 1)$, u(0,0) = 0, v(0,0) = 0, Du(0,0) = 0, and, Dv(0,0) = 0. Then there exists a C^{k} function $h : (R^{m},0) \rightarrow (R^{n},0)$ such that

(i) the set

$$W^{C}(0) = \{(x,y) \in (R^{m+n}, (0,0)) : y = h(x)\},\$$

is a locally invariant Ck manifold in R^{m+n}, and,

(ii) $W^{C}(0)$ consists exactly of those solutions (x(t),y(t)) for

which |y(t)| (t $\in \mathbb{R}$) is small.

Proof: (See Theorem 2.1 on p. 313 of Chow-Hale (1982.)

Definition: $\mathcal{W}^{C}(0)$ is called the <u>center manifold</u> of (x,y) = (0,0).

Theorem 2.9 (center stable manifold theorem): Consider

$$\dot{\mathbf{x}} = \mathbf{A}\mathbf{x} + \mathbf{u}(\mathbf{x},\mathbf{y}) \qquad (\mathbf{x} \in \mathcal{R}^{m})$$
$$\dot{\mathbf{y}} = \mathbf{B}\mathbf{y} + \mathbf{v}(\mathbf{x},\mathbf{y}) \qquad (\mathbf{y} \in \mathcal{R}^{n})$$

<u>near</u> (x,y) = (0,0), where all the eigenvalues of A have non-positive real parts, all the eigenvalues of B have positive real parts, $u,v \in C^k$ $(k \ge 1)$, u(0,0) = 0, v(0,0) = 0, Du(0,0) = 0, and, Dv(0,0) = 0. Then there exists a C^k function $g : (R^m,0) \rightarrow (R^n,0)$ such that

(i) the set

$$W^{CS}(0) = \{(x,y)_{1} \in (R^{m+n}, (0,0)) : y = g(x)\}$$

is a locally invariant C^k manifold, and, (ii) $W^{CS}(0)$ consists of exactly those solutions (x(t),y(t)) for which |y(t)| $(t \ge 0)$ is small.

Proof: (See Theorem 2.11 on p. 319 of Chow-Hale (1982).)

q.e.d.

<u>Definition</u>: $W^{CS}(0)$ is called the <u>center stable manifold</u> of (x,y) = 0. Similarly, one can define the <u>center unstable manifold</u>, $W^{CU}(0)$, of (x,y) = (0,0).

<u>Remark</u>: It should be pointed out that these invariant manifolds are tangent to the respective generalized eigenspaces of the linearization.

2.9. Linearization Theorems.

Linearization theorems allow us to say something about the local structure of a flow near a point. To start with there is the socalled "straightening-out" theorem which states that near a regular point (i.e. not a rest point) of a C^2 flow, C^1 change of coordinates near the point can be made so that locally the flow becomes a parallel flow (see Theorem 21.6 on p. 56 of Abraham-Robbin (1967)). In the case of a hyperbolic rest point of a C^1 flow, we have the well-known theorem of Hartman and Grobman which says that locally (i.e. near the rest point), there is a flow equivalence between the original system and the linear system that arises from linearizing the original system at the rest point S(see Theorem 7.1 on p. 244 of Hartman (1982) or Theorem 7.2 on p. 110 of Chow-Hale (1982)). In the general case, we have the following theorem due to Palis and Takens:

Theorem 2.10 (Palis-Takens): Consider

σ

 $\dot{\mathbf{x}} = A\mathbf{x} + \mathbf{u}(\mathbf{x},\mathbf{y}) \qquad (\mathbf{x} \in \mathbb{R}^{m})$ $\dot{\mathbf{y}} = B\mathbf{y} + \mathbf{v}(\mathbf{x},\mathbf{y}) \qquad (\mathbf{y} \in \mathbb{R}^{n})$

(2.4)

21

à

<u>near</u> (x,y) = (0,0), <u>where</u> $u,v \in C^k$ $(k \ge 1)$, u(0,0) = 0, v(0,0) = 0, Du(0,0) = 0, and, Dv(0,0) = 0. Assume that all eigenvalues of A have zero real parts and all eigenvalues of B have non-zero real parts. Then system (2.4) is locally C^0 - equivalent to the decoupled system

$$\dot{w} = Aw + u(w,h(w))$$
 ($w \in R^{m}$)
 $\dot{z} = Bz$ ($z \in R^{n}$)
(2.5)

near (w,z) = (0,0), where h defines the center manifold of (0,0).

<u>Proof</u>: See the second theorem on p. 341 of Palis-Takens (1977). This fact is also mentioned in Remark 2.16 on pp. 322-323 of Chow-Hale (1982).

Using Hartman-Grobman's theorem, we can prove the following proposition which tells us when the stable manifold of a boundary hyperbolic rest point will have empty intersection with the positive cone, R_{+}^{n} .

Proposition 2.11: Consider

¢

 $\dot{\mathbf{x}} = \mathbf{f}(\mathbf{x})$ $(\mathbf{x} \in \mathbb{R}^n)$

and assume that $cl(R_{+}^{n})$ is positively invariant. Let $x_{0} \in b(R_{+}^{n})$ be a hyperbolic rest point. If $W^{u}(x_{0}) \cap (R^{n} \setminus cl(R_{+}^{n})) \neq \emptyset$, then $W^{s}(x_{0}) \cap R_{+}^{n} = \emptyset$.

22

1×

<u>Proof</u>: Suppose not, that is suppose $W^{S}(x_{0}) \cap R_{+}^{n} \neq \emptyset$. If we can show that for fixed $x_{1} \in W^{S}(x_{0})$, $x_{2} \in W^{U}(x_{0})$ where $x_{1}, x_{2} \neq x_{0}$, there is a point x which is arbitrarily close to x_{1} and a time t > 0 so that $\phi(x,t)$ is arbitrarily close to x_{2} , then this will contradict the positive invariance of $cl(R_{+}^{n})$. Since this can essentially be reduced to a local problem, it suffices to establish the following.

Consider

$$\dot{\mathbf{x}} = \mathbf{A}\mathbf{x} \qquad (\mathbf{x} \in \mathbf{R}^{\mathsf{T}})$$
$$\dot{\mathbf{y}} = \mathbf{B}\mathbf{y} \qquad (\mathbf{y} \in \mathbf{R}^{\mathsf{T}})$$

where all the eigenvalues of A have negative real parts and all the eigenvalues of B have positive real parts. Fix $(x^*,0)$ and $(0,y^*) \in \mathbb{R}^{m+n}$ with $x^* \neq 0$, $y^* \neq 0$. Then there exists $(x_N,y_N) \in \mathbb{R}^{m+n}$ and $t_N > 0$ such that $(x_N,y_N) \rightarrow (x^*,0)$ and $\phi((x_N,y_N),t_N) \rightarrow (0,y^*)$ as $N \rightarrow +\infty$. For a proof, let $t_N = N$, $x_N = x^*$, and, $y_N = e^{-t_N B} y^*$. Then clearly $x_N \rightarrow x^*$ and $y_N \rightarrow 0$, since all eigenvalues of B have positive real parts. Furthermore we have,

$$\phi((x_N, y_N), t_N) = (e^{t_N A} x_N, e^{t_N B} y_N) = (e^{t_N A} x^*, y^*) \rightarrow (0, y^*)$$

as $N \rightarrow +\infty$ as desired.



In this section a sufficient condition for the existence of saddle nodes and some related calculations will be discussed. (See pp. 323-329 of Chow-Hale (1982).)

Consider the system

$$\dot{y}_{1} = P(Y)$$

 $\dot{\overline{y}} = A\overline{y} + Q(Y)$ (2.6)

where $\overline{y} = (y_2, \dots, y_n) \in \mathbb{R}^{n-1}$, $Y = (y_1, \overline{y}) \in \mathbb{R}^n$, P and Q are of order two or higher in Y and all the eigenvalues of the (n-1) by (n-1) matrix A have negative real parts. Clearly Y = 0 is a rest point of system (2.6) and the variational matrix of (2.6) at Y = 0 has 0 as a simple eigenvalue and the remaining (n-1) eigenvalues have negative real parts.

<u>Theorem 2.12</u> (saddle-node): Let $\overline{y} = \psi_1(y_1)$ with $\psi_1(0) = 0$ be the solution of

$$A\overline{y} + Q(y_1, \overline{y}) = 0$$
 (2.7)

near $(y_1, \overline{y}) = (0, 0)$ and let

$$\psi_2(y_1) = P(y_1, \psi_1(y_1))$$
 (2.8)

If

$$\psi_2(y_1) = \eta y_1^2 + 0(|y_1|^3)$$

and $n \neq 0$ then Y = 0 is a saddle node: the strong stable manifold of Y = 0 is $(n-1) - dimensional and is tangent to the hyperplane <math>y_1 = 0$, the 1 - dimensional center manifold of Y = 0 is tangent to the $y_1 - axis$, and, Y = 0 is stable on one side of its strong stable manifold and unstable on the other side.

<u>Proof</u>: Refer to Theorem 65 on pp. 340-346 of Andronov-Leontovich-Gordon-Maier (1975) for the case when n = 2 and pp. 326-327 of Chow-Hale (1982) for the general case.

Remarks: (i) It turns out that

$$n = \frac{1}{2} \frac{\partial^2 P}{\partial y_1^2} (0)$$
 (2.9)

(ii) If $\eta > 0$, then the domain of stability of Y = 0 is on the side of the strong stable manifold for which $y_1 < 0$; if $\eta < 0$, it is just the reverse.

The above theorem is stated in "canonical" form. In applications, it is often necessary to transform the problem into this form. Some of that calculation will be carried out in the following.

Consider the system

$$\dot{Z} = G(Z)$$
 (2.10)

where $Z \in \mathbb{R}^n$, G(0) = 0, and the variational matrix of system (2.10) at Z = 0 has 0 as a simple eigenvalue and the remaining (n-1) eigenvalues have negative real parts. Let v_1 be an eigenvector corresponding to the eigenvalue 0 and let v_2, \dots, v_n be a basis of the generalized eigenspace corresponding to the remaining (n-1) eigenvalues. Let $N = [v_1, \dots, v_n]$ be the n by n matrix whose ith column is v_i and let $Y = N^{-1}Z$: Under this transformation, (2.10) takes the form (2.6). The $y_1 = 0$ hyperplane is transformed back to the hyperplane, in the Zcoordinates, spanned by the vectors v_2, \dots, v_n (denoted by $\langle v_2, \dots, v_n \rangle$), the $y_1 - axis$ is transformed back to the line spanned by the vector v_1 (denoted by $\langle v_1 \rangle$), and the positive y_1 axis is transformed back to the half line

$$\{\lambda v_1 \in R^n : \lambda > 0\}$$

Let
$$N = [n_{ij}]_{1 \le i, j \le n}$$
, $N^{-1} = [n_{ij}]_{1 \le i, j \le n}$
G(Z) = (G₁(Z),...,G_n(Z)), and

$$\hat{g}_{i}(z) = \sum_{\substack{1 \le i_{1} + \dots + i_{n} \le 2}} g_{i}^{(i_{1}, \dots, i_{n})} z_{1}^{i_{1}} \dots z_{n}^{i_{n}} + O(|z|^{3})$$

 $(i = 1, \dots, n)$. Also let $I_h = (0, \dots, 0, 2, 0, \dots, 0)$ with a "2" in the h^{th} column and the rest are all 0's, and, $I_{h,k} = (0, \dots, 0, 1, 0, \dots, 0, 1, 0, \dots, 0)$ with 1's in the h^{th} and k^{th} columns and the rest are all 0's. Then

$$n = \sum_{i=1}^{n} \frac{1}{n_{1i}} \left(\sum_{h=1}^{n} g_i^{h} n_{h1}^2 + \sum_{i=1 \le h \le k \le n} g_i^{h,k} n_{h1}^{h} n_{k1}^{h} \right).$$
Hence, if $n \neq 0$, Z = 0 will be a saddle node. The strong stable manifold $W^{SS}(0)$ of Z = 0 is tangent to $\langle v_2, \cdots, v_n \rangle$ and the center manifold $W^C(0)$ is tangent to $\langle v_1 \rangle$.

2.11. Omega-Limit Sets.

In this section a number of simple but useful technical results relating to omega-limit sets will be presented. These results will be used repeatedly in the main body of the thesis.

Proposition 2.13: Consider

$$\dot{\mathbf{x}} = \mathbf{f}(\mathbf{x}) \qquad (\mathbf{x} \in \mathbb{R}^n) \tag{2.11}$$

where $f \in Lip$. Assume that $cl(R_{+}^{n})$ is positively invariant, let x^{*} <u>be a point on</u> $b(R_{+}^{n})$ with $x_{i}^{*} = 0$ and such that $f_{i}(x^{*}) > 0$. Then $x^{*} \notin w(x)$ for any $x \in cl(R_{+}^{n})$.

Proof: Since x^* is a regular point, if f is C^2 we can use the "straightening-out theorem" to construct a C^1 change of coordinates near x^* so that locally the flow becomes a parallel flow. From this and the positive invariance of $cl(R_+^n)$, we can easily see that $x^* \notin w(x)^{-1}$ for any $x \in cl(R_+^n)$. In the general case when $f \in Lip$, $f_i(x^*) > 0$ implies that $x_i(t)$ is strictly increasing near t = 0, where x(t) denotes the solution of system (2.11) such that $x^*(0) = x^*$. Since $x_i^* = 0$, for sufficiently small t < 0, $x(t) \notin cl(R_+^n)$. On the other hand, if $x \in w(x)$ for some $x \in cl(R_+^n)$ then $x(t) \in w(x)$ also. By choosing t < 0 sufficiently small, this implies $w(x) \notin cl(R_+^n)$, contradicting the positive invariance of $cl(R_+^n)$.

Lemma 2.14 (Butler-McGehee): Consider

$$\dot{\mathbf{x}} = \mathbf{f}(\mathbf{x})$$
 $(\mathbf{x} \in \mathbb{R}^n)$

where $f \in C^1$. Let $x^* \in R^n$ be a hyperbolic rest point. If $x^* \in w(x)$ for some $x \in R^n$, then either $w(x) = \{x^*\}$ for both $w(x) \cap W^S(x^*)$ and $w(x) \cap W^U(x^*)$ contain points other than x^* .

Proof: (Refer to Appendix 1 in Freedman-Waltman (1984).)

q.e.d.

 \mathbb{P}

<u>Remark</u>: This lemma together with Proposition 2.11 provide us with a useful tool for proving persistence.

The following proposition shows that under certain conditions O is not in the omega-limit set of any orbit. The point here is that the flow is only continuous and we may not be able to use the Butler-McGehee lemma as stated.

Proposition 2.15: Consider

 $x_{i} = f_{i}(x_{1}, \dots, x_{n}, y) \quad (i = 1, \dots, n) \leq 1$

28

$$\dot{y} = g(x_1, \cdots, x_n, y)$$

where $f_i, g \in Lip, cl(R_+^{n+1})$ is positively invariant, and 0 is a rest point. Let $x = x_1 + \cdots + x_n$ and suppose that there is a neighbourhood U of 0 in $cl(R_+^{n+1})$ such that:

$$\dot{x} \ge k_1 x$$
 and $\dot{y} \le -k_2 y$ (2.12)

in U, for some $k_1, k_2 > 0$. Moreover, assume that the non-negative y axis is invariant and 0 is globally stable over it. Then $0 \notin w(x_1, \dots, x_n, y)$ for all $(x_1, \dots, x_n, y) \in \mathbb{R}^n_+$ for which $0^+(x_1, \dots, x_n, y)$ is bounded.

of: Let $\varepsilon > 0$ be such that (2.12) is valid on

$$V = \{(x_1, \dots, x_n, y) \in cl(R_+^{n+1}) : x \le \varepsilon, y \le \varepsilon\}.$$

$$x(t) \ge x(0) e^{k_1 t} \quad e^{-k_2 t}, \text{ the eventual exit set}$$

 V^{O} and the immediate exit set V^{-} are given by:

Since

 $V^{0} = \{(x_{1}, \dots, x_{n}, y) \in V : x > 0\}$ $V^{-} = \{(x_{1}, \dots, x_{n}, y) \in V : x = \epsilon\}$

refore V is a Wazewski set and hence the time of exit map is continuous. Based on this, we can complete the proof using an argument similar to the proof of the Butler-McGehee lemma.

The following is an extension of the Butler-McGehee lemma to general (not necessarily hyperbolic) rest points.

Proposition 2.16: Consider

 $\dot{\mathbf{x}} = \mathbf{f}(\mathbf{x}) \qquad (\mathbf{x} \in \mathbb{R}^n)$

where $f \in C^2$. Let x^* be a rest point. If $x^* \in w(x)$ for some $x \in R^n$ and $x \notin W^{CS}(x^*)$, then $w(x) \cap W^{CS}(x^*)$ contains a point other than x^* .

<u>Proof</u>: Use Palis-Takens linearization theorem to construct a closed neighbourhood U of x^* such that $V = U \setminus W^{CS}(x^*)$ is a Wazewski set with $V^0 = V$. We can then repeat the argument as in the proof of the Butler-McGehee lemma to complete the proof.

CHAPTER 3

A ONE-LOCUS, TWO-ALLELE MODEL WITH FERTILITY / VIABILITY DIFFERENCES

In this chapter, the global convergence to rest points of a continuous, one-locus, two-allele genetic model that incorporates nonlinear birth and death rates as well as unequal genotypic fertilities/ viabilities will be analyzed. This model will be derived in Appendix 1.

3.1. Introduction and the Model.

This chapter is concerned with a dynamical system of the form

$\hat{x}_{1} = \frac{u^{2}}{(u+v)^{2}} B(x) - \frac{x_{1}}{x} D(x)$	•	
$\dot{x}_2 = \frac{2uv}{(u+v)^2} B(x) - \frac{x_2}{x} D(x)$		(3.1)
$\dot{x}_3 = \frac{v^2}{(u+v)^2} B(x) - \frac{x_3}{x} D(x)$		

where

 $u = f_{1}x_{1} + \frac{1}{2}f_{2}x_{2}$ $v = f_{3}x_{3} + \frac{1}{2}f_{2}x_{2}$

(3.2)

$$x = x_1 + x_2 + x_3$$
 (3.3)

$$f_1, f_2, f_3 > 0$$
 (3.4)

and

5

• = $\frac{d}{dt}$ = derivative with respect to time t.

Here x_1 , x_2 , and x_3 denote the number of organisms of genotype AA, Aa, and aa respectively of a one-locus, two-allele problem where A and a stand for the two allele types. System (3.1) is a generalization of the model (without predator) studied in Freedman-Waltman (1978, 1982). In these two papers, the f_i 's are taken to be unity. In order to preserve some consistency, the same notations as in these two papers will be adopted. Models similar to system (3.1) have been used by K. Beck in her study of the population genetics of cystic fibrosis, Beck (1982). (See also Beck-Keener-Ricciardi (1982, 1984) for related models.) In Section 3.1 of Waltman (1984), there is a nice discussion on the derivation of these types of models. The effect of the introduction of a predator population to the persistence of the system for which the prey population (in the absence of the predator) is modelled by system (3.1) will be studied in Chapter 4. Other related works of interest, besides the ones mentioned above are: Akin (1979, 1983), Aronson-Weinberger (1975, 1978), Beck (1984), Butler-Freedman-Waltman (1982), Christiansen-Fenchel (1977), Conley-Fife (1982), Hadeler-Glas (1983), Hadeler-Liberman (1975), Losert-Akin (1983), Nagylaki (1977), Nagylaki-Crow (1974), and, Shahshahani (1979).

In system (3.1), B(x)/x and D(x)/x are the natural, intrinsic birth and death rates respectively of the entire population x. The f_i's can be thought of as corresponding to fertility or viability of the genotypes x_i 's. (Refer to Appendix 1 for the derivation of the coefficients of B(x).) We will make the following assumptions on the functions B and D:

(H1) B,D :
$$cl(R_{+}^{1}) \rightarrow cl(R_{+}^{1})$$
 are C^{2} ,
(H2) B(0) = D(0) = 0 ,

(H3) B'(0) > D'(0),

(H4) B'(x), D'(x) > 0 for $x \in cl(R^{1}_{+})$,

(H5) there exists a unique K > 0 ("carrying capacity") such that B(K) = D(K), and,

(H6) B'(K) < D'(K)

where ' = $\frac{d}{dx}$ = derivative with respect to x.

Assumption (H1) is sufficient to ensure existence and uniqueness of initial value problems for all $t \ge 0$ and for the invariant manifold and saddle node calculations to follow. Assumption (H2) implies that there can be no birth nor death when there is zero population. (H4) states that the birth and death functions are increasing functions of the entire population. (H3) implies that for small populations, the birth rate increases more rapidly than the natural death rate. (H5) states that there is a carrying capacity of the environment, at which point, birth rate equals natural death rate. (H6) says that at the carrying capacity the birth rate increases less rapidly than the natural death rate.

In Section 3.2, we will show that system (3.1) defines a continuous semi-flow ϕ on $cl(R_+^3)$ and ϕ is a smooth (C²) local flow on R_+^3 . Moreover, all solutions $\phi(x_0,t)$ with initial conditions

$$X_{0} = (x_{10}, x_{20}, x_{30}) \in c1(R_{+}^{3}) \setminus \{(0, 0, 0)\}$$

approach the closed, positively invariant simplex

$$c1(U) = \{X \in c1(R_{+}^{3}) : x = K\}$$
 (3.5)

where $U = \{X \in R_+^3 : x = K\}$. In Section 3.3, we discuss the rest points of the flow ϕ . A complete global picture of the flow ϕ on cl(U) is described in Section 3.4 by means of the "projective" flow $\tilde{\phi}$ of ϕ on

$$cl(S) = \{(x_1, x_3) \in cl(R_+^2) : x_1 + x_3 \leq K\}$$
 (3.6)

where

$$S = \{(x_1, x_3) \in R_+^2 : x_1 + x_3 < K\}$$

Using these results, we show in Section 3.5 the main result of the chapter: for all $X \in cl(R_+^3)$, $\phi(X,t)$ converges to some rest point of ϕ , as t tends to $+\infty$. It turns out that which rest point $\phi(X,t)$ converges to is predictable and these results are directly analogous to the discrete case (see Chapter 3 of Roughgarden (1979)). Finally, Section 3.6 is a short discussion of the results.

A.1

3.2. The Associated Flow.

Let us define the right hand side of system (3.1) to be (0,0,0) when $X = (x_1, x_2, x_3) = (0,0,0)$. That is, define

٩

$$F = (F_1, F_2, F_3) : c1(R_+^3) \to R^3$$

$$F(X) = (F_1(x_1, x_2, x_3), F_2(x_1, x_2, x_3), F_3(x_1, x_2, x_3))$$

where

bу

$$F_{1}(X) = \begin{cases} \frac{u^{2}}{(u+v)^{2}} B(x) - \frac{x_{1}}{x} D(x) & \text{if } X \in c1(R_{+}^{3}) \setminus \{(0,0,0)\} \\ 0 & \text{if } X = (0,0,0) \end{cases}$$

$$F_{2}(X) = \begin{cases} \frac{2uv}{(u+v)^{2}} B(x) - \frac{x}{x} D(x) & \text{if } X \in cl(R_{+}^{3}) \setminus \{(0,0,0)\} \\ 0 & \text{if } X = (0,0,0) \end{cases}$$
(3.7)

$$F_{3}(X) = \begin{cases} \frac{v^{2}}{(u+v)^{2}} B(x) - \frac{x_{3}}{x} D(x) & \text{if } X \in cl(R_{+}^{3}) \setminus \{(0,0,0)\} \\ 0 & \text{if } X = (0,0,0) \end{cases}$$

0

where u, v and x are given by (3.2) and (3.3). The system (3.1) becomes

$$\dot{X} = F(X) \tag{3.8}$$

,

•

Clearly, F is locally Lipschitz. For example, if $X \in cl(R_+^3)$ with $0 < |X| \le r$,

$$|F(X) - F(0)| < 3[B(x) + D(x)] < C |x| \le C' |X|$$

for some constants C,C' > 0 depending on r only. Also, F(0) = 0.

We will summarize some of the properties of system (3.8) in the following proposition.

Proposition 3.1: System (3.8) defines a continuous positive semi-flow ϕ on cl(R_{+}^{3}) which becomes a smooth local flow when restricted to R_{+}^{3} . The simplex cl(U) defined in (3.5) is closed, positively invariant under ϕ and is globally stable on cl(R_{+}^{3}) \ {(0,0,0)}. Furthermore, the vector field F points into R_{+}^{3} on b(R_{+}^{3}) \ cl(H_{x1}UH_{x3}), H_{x1} and H_{x3}. are positively invariant and (0,0,0) is a repelling rest point of ϕ .

<u>Proof</u>: The definition of the flow ϕ by (3.8) and its smoothness properties are clear. It is also easy to see that F points into R_{+}^{3} on $b(R_{+}^{3}) \wedge cl(H_{x_{1}} \cup H_{x_{3}})$. For example for $X \in H_{x_{2}}$, we have, $x_{1} = x_{3} = 0$, so that $F_{1}(X) = F_{3}(X) = \frac{1}{4} B(X) > 0$. Consequently F(X) points into R_{+}^{3} . From this and from the fact that $H_{x_{1}}$ and $H_{x_{3}}$ are invariant, it follows from Proposition 2.1 that $cl(R_{+}^{3})$ is positively invariant.

On the other hand, since.

$$\dot{x} = B(x) - D(x)$$
 (3.9)

the assumptions (H1) - (H6) in Section 3.1 imply that x = K is a sink for system (3.9) and for all $x(0) = x_0 \in R_+^1$, we have, $0 < x(t) \neq K$, as $t \neq +\infty$. This shows that cl(U) is globally stable over $cl(R_+^3) \setminus \{(0,0,0)\}$.

<u>Remark</u>: Let $V : cl(R_+^3) \setminus \{(0,0,0)\} \neq R_+^1$ be defined by V(X) = x. Then for all $K' \in R_+^1$, $V^{-1}(K')$ is a closed simplex and for all $K' \neq K$, the vector field F on $V^{-1}(K')$ points towards $V^{-1}(K) = cl(U)$.

3.3. The Rest Point Set.

By Proposition 3.1, all rest points on ϕ other than (0,0,0) must lie on cl(U). Hence (K,0,0) and (0,0,K) are the only rest points in $b(R_+^3)$ besides (0,0,0).

Proposition 3.2: Define the curve $H : R_{+}^{1} \rightarrow U$ by

$$H(c) = \left(\frac{c^{2}K}{(1+c)^{2}}, \frac{2cK}{(1+c)^{2}}, \frac{K}{(1+c)^{2}}\right)$$
(3.10)

for all $c \in R_{+}^{1}$. Then all rest points of ϕ in R_{+}^{3} must lie on $H(R_{+}^{1})$.

ر تمسر

٠E

Proof: Let
$$X^{*} = (x_{1}^{*}, x_{2}^{*}, x_{3}^{*}) \in \mathbb{R}_{+}^{3}$$
 be a rest point for ϕ and let
 $c = \frac{x_{2}^{*}}{2 x_{3}^{*}}$. Then $c > 0$ and $x^{*} = x_{1}^{*} + x_{2}^{*} + x_{3}^{*} = K$. Let
 $u^{*} = f_{1}x_{1}^{*} + \frac{1}{2}f_{2}x_{2}^{*}$
(3.11)
 $v^{*} = f_{3}x_{3}^{*} + \frac{1}{2}f_{2}x_{2}^{*}$

According to (3.1), we have,

 $\frac{\frac{u^{*2}}{(u^{*}+v^{*})^{2}} B(K) - \frac{x_{1}}{K} D(K) = 0}{\frac{2u^{*}v^{*}}{(u^{*}+v^{*})^{2}} B(K) - \frac{x_{2}}{K} D(K) = 0}{\frac{v^{*2}}{(u^{*}+v^{*})^{2}} B(K) - \frac{x_{3}}{K} D(K) = 0},$

so that,

$$u^{*2} K = x_1^* (u^* + v^*)^2$$
 (3.12)

$$2u^{*}v^{*}K = x_{2}^{*}(u^{*}+v^{*})^{2}$$
(3.13)

$$v^{*2} K = x_3^* (u^* + v^*)^2$$
 (3.14)

since B(K) = D(K). Therefore

$$\frac{(3.13)}{(3.12)} \cdot \frac{(3.13)}{(3.14)}$$

gives

$$4 = \frac{x_{2}^{*2}}{x_{1}^{*}x_{3}^{*}} \text{ or } x_{1}^{*} = c^{2} x_{3}^{*}.$$

Now, $x^* = K$ implies $(c^2 + 2c + 1)x_3^* = K$ and hence

$$X^{\star} = \left(\frac{c^2 K}{(1+c)^2}, \frac{2cK}{(1+c)^2}, \frac{K}{(1+c)^2}\right)$$

as desired.

q.e.d.

Remarks: (i)
$$cl(H(R_{+}^{1})) = H(R_{+}^{1}) \cup \{(K_{3},0,0),(0,0,K)\}$$
 (3.15)
It will be referred to as the Hardy-Weinberg manifold. It is a closed
segment of a parabolic curve in $cl(U)$.

(ii)
$$\lim_{c \to 0} H(c) = (0,0,K)$$
 and $\lim_{c \to +\infty} H(c) = (K,0,0)$. Let
 $a_1 = f_1 - f_2$ and $a_3 = f_3 - f_2$ (3.16)

Proposition 3.3:

(i) If
$$(a_1, a_3) = (0, 0)$$
, then $H(R_+^1)$ is the set of rest
points of ϕ in R_+^3 .
(ii) If $a_1, a_3 > 0$ or if $a_1, a_3 < 0$, then ϕ has exactly
one rest point in R_+^3 , namely, $H(c)$ with $c = a_3/a_1$
(iii) If $a_1 \ge 0$, $a_3 \le 0$ and $(a_1, a_3) \ne (0, 0)$ or if $a_1 \le 0$

$$a_3 \ge 0$$
 and $(a_1, a_3) \ne (0, 0)$, then ϕ has no rest points in R_+^3 .

<u>Proof</u>: By the previous proposition, all rest points X^* of ϕ in R_+^3 are of the form H(c) given by (3.10) for some $c \in R_+^1$. According to (3.11),

$$u^{*} = \frac{cK}{(1+c)^{2}} (cf_{1} + f_{2})$$

$$v^{*} = \frac{K}{(1+c)^{2}} (cf_{2} + f_{3}) .$$
(3.17)

Substituting this into (3.12) gives, after some simplifications,

С

 $(1+c)^{2}(cf_{1}+f_{2})^{2} = (c^{2}f_{1}+2cf_{2}+f_{3})^{2}$

or

1

$$f_1 + f_2 = cf_2 + f_3$$
 (3.18)
 $c_1 = a_3 / a_1$ (3.19)

provided $a_1 \neq 0$.

Thus, (3.12) imposes no restrictions on c in case (i), it requires c to be a_3 / a_1 in case (ii), and in case (iii) no $c \in R_+^1$ will satisfy (3.12). To complete the proof, all we need is to verify that any $c \in R_+^1$ satisfying (3.12) also satisfies (3.13) and (3.14). But this follows from a straightforward calculation. For example, (3.13) simplifies to (3.12) using (3.18).

<u>Remark</u>: In case (i) we have a 1-dimensional manifold of rest points. In general, a manifold of rest points (critical points) is referred to as a critical manifold (see Conley-Fife (1982)).

3.4. Projective Flow.

In order to study the dynamics of ϕ on $cl(R_+^3)$, we will first study the global dynamics of ϕ on the globally stable positively invariant simplex cl(U). This is equivalent to investigate the phase portrait of the "projective" semi-flow ϕ of ϕ on cl(S) (defined in (3.6)). Here $\tilde{\phi}$ is defined by (3.1) with x_2 being replaced by $K - x_1 - x_3$ throughout. That is, we have,

$$\dot{x}_{1} = \frac{u^{2}}{(u+v)^{2}}L - \frac{x_{1}}{K}L$$
(3.20)
$$\dot{x}_{3} = \frac{v^{2}}{(u+v)^{2}}L - \frac{x_{3}}{K}L$$

where, by (3.2),

$$u = fK + (a_{1} + f)x_{1} - fx_{3}$$

$$v = fK - fx_{1} + (a_{3} + f)x_{3}$$
(3.21)

and

$$L = B(K) = D(K)$$
 (3.22)

$$f = f_2 / f$$
 (3.23)

By lifting the flow $\tilde{\phi}$ back to cl(U) (that is, if $(x_1(t), x_3(t))$ is a solution of (3.20) in cl(S), then $(x_1(t), x_2(t), x_3(t))$ with $x_2(t) = K - x_1(t) - x_3(t)$ will be a solution of (3.1) in cl(U)), we obtain the flow ϕ on cl(U). Clearly, by Proposition 3.1 or by a direct verification, the vector field defined by system (3.20) points into S on b(S) \ {(K,0)',(0,K)} where b(S) denotes the boundary of S in the $x_1 - x_3$ plane and that (K,0) and (0,K) are the only rest points of $\tilde{\phi}$ in b(S). Hence S is positively invariant under $\tilde{\phi}$. According to Proposition 3.3,

٥

i A.

$$(x_1^{\star}, x_3^{\star}) = \left(\frac{c^2 K}{(1+c)^2}, \frac{K}{(1+c)^2}\right)$$
 with $c = a_3 / a_1$ (3.24)

is the only rest point of $\tilde{\phi}$ in S, provided $a_1, a_3 > 0$ or $a_1, a_3 < 0$; and when $a_1 \ge 0$, $a_3 \le 0$, $(a_1, a_3) \ne (0, 0)$ or when $a_1 \le 0$, $a_3 \ge 0$, $(a_1, a_3) \ne (0, 0)$, $\tilde{\phi}$ has no rest points in S.

For the remainder of this section, we will restrict our attention to the case when $(a_1,a_3) \neq (0,0)$. The case when $(a_1,a_3) = (0,0)$ for the "full" system (3.1) will be discussed in Section 3.5.

The variational matrix $M(x_1,x_3)$ of (3.20) at an arbitrary point (\mathcal{R}_1,x_3) in cl(S) is given by:

$$M(x_{1},x_{3}) = L \begin{bmatrix} \frac{2u(fu+a_{1}v+fv)}{(u+v)^{3}} - \frac{1}{K} & -\frac{2u(a_{3}u+fu+fv)}{(u+v)^{3}} \\ -\frac{2v(fu+a_{1}v+fv)}{(u+v)^{3}} & \frac{2v(a_{3}u+fu+fv)}{(u+v)^{3}} - \frac{1}{K} \end{bmatrix}$$
(3.25)

in particular, $\begin{bmatrix} - & \frac{a_1}{f_1} \end{bmatrix}$

 $M(K,0) = \frac{L}{K}$

 $\frac{2(a_3+f)}{f_1}$

$$M(0,K) = \frac{L}{K} \begin{bmatrix} -1 & 0 \\ \\ \\ -\frac{2(a_1+f)}{f_3} & -\frac{a_3}{f_3} \end{bmatrix}$$

-1

0

(3.27)

(3.26)

and, if (x_1^*, x_3^*) exists,

$$M(x_{1}^{*}, x_{3}^{*})^{\circ} = \frac{L}{K} \begin{bmatrix} \frac{2a_{3}e_{1}}{RW} - 1 & -\frac{a_{3}e_{3}}{RW} \\ -\frac{a_{1}e_{1}}{RW} & -\frac{2a_{1}e_{3}}{RW} - 1 \end{bmatrix}$$
(3.28)

where

$$R = a_1 + a_3$$
, $W = f_1 f_3 - f_2^2$, (3.2)

$$e_1 = a_1^2 + fR$$
, $e_3 = a_3^2 + fR$. (3.30)

Note that under these notations,

٩

$$x_1^{\star} = \frac{a_3^2 K}{R^2}$$
, $x_3^{\star} = \frac{a_1^2 K}{R^2}$

$$u^* = \frac{a_3 KW}{R^2}$$
, $v^* = \frac{a_1 KW}{R^2}$, and, $u^* + v^* = \frac{KW}{R}$

X

Proposition 3.4: The linearized stability properties of the rest points of $\tilde{\phi}$ in cl(S) are given by the following diagram:

Parameter Region		Rest Points		
al	^a 3	(K,0)	(0,K)	(x [*] ₁ ,x [*] ₃)
+	÷	-,-	-,-	-,+
-	-	-,+	-,+	-,-
+	-	-,-	-,+	*
-	+	-,+	-,-	*
+	0	-,-	- ,0	*
-	0	-,+	-,0	*
0,	+	-,0	-,-	*
0	-	-,0	-,+	*

where a_i (i=1,3) is + (resp. -,0) means that a_i is positive (resp. negative, zero), "-,-" under a rest point means that both eigenvalues of the variational matrix M at that rest point are negative, "-,+" means that one eigenvalue is negative and the other is positive, "-,0" means that one eigenvalue is negative and the other is zero, and, "*" indicates that the rest point does not exist in that parameter region.

<u>Proof</u>: The eigenvalues of M(K,0) are: $-\frac{a_1L}{f_1K}$ and $-\frac{L}{K}$, those of M(0,K) are: $-\frac{a_3L}{f_3K}$ and $-\frac{L}{K}$ and those of M(x_1^*, x_3^*), if (x_1^*, x_3^*) exists, are: $\frac{a_1a_3L}{WK}$ and $-\frac{L}{K}$. To complete the proof, all we need is to note that if $a_1, a_3 > 0$, then W > 0 and if $a_1, a_3 < 0$, then W < 0. q.e.d. Next, we will investigate the phase portrait of $\,\widetilde{\varphi}\,$ in a neighbourhood of each rest point.

Proposition 3.5: We have the following cases, depending on the parameter regions: (i) $a_1 > 0$, $a_3 > 0$: (K,0) and (0,K) are sinks, (x_1^*, x_3^*) is a saddle; (ii) $a_1 < 0$, $a_3 < 0$: (K,0) and (0,K) are saddles whose stable manifolds have trivial intersections with cl(S), namely at (K,0) and (0,K) respectively, and whose unstable manifolds intersect cl(S) at a whole separatrix (out of four for a saddle), (x_1^*, x_3^*) is a sink;

(iii) $a_1 > 0$, $a_3 < 0$:

(K,O) is a sink, (O,K) is a saddle whose local phase portrait is similar to that of case (ii);

(iv) a₁ < 0 , a₃ > 0 : ``

(O,K) is a sink, (K,O) is a saidle whose local phase portrait is similar to that of case (ii);

$$(v) a_1 > 0, a_3 = 0$$
:

(K,0) is a sink, (O,K) is a saddle node: it has a neighbourhood in R^2 which is the union of one parabolic sector, two hyperbolic sectors and three separatrices, (0,K) is stable within the parabolic sector, the union of the two hyperbolic sectors covers a neighbourhood of (0,K) in cl(S) with the separtrix dividing the hyperbolic sectors lying wholly inside cl(S);

$$(vi) a_1 < 0, a_3 = 0$$
:

1

(K,0) is a saddle whose local phase portrait is similar to that of case (ii), (0,K) is a saddle node: it is stable within the parabolic sector which covers a neighbourhood of (0,K) in cl(S);

$$(vii) a_1 = 0, a_3 > 0$$
:

(0,K) is a sink, (K,O) is a saddle node whose local phase
portrait is similar to that of (0,K) in case (v);

(viii)
$$a_1 = 0$$
, $a_3 < 0$:

(0,K) is a saddle whose local phase portrait is similar to that of case (ii), (K,0) is a saddle node whose local phase portrait is similar to that of (0,K) in case (vi).

<u>Proof</u>: When $a_1 < 0$ (consequently (K,0) is a saddle), $\left(\frac{2f_3-f_2}{f_2}, 1\right)$ is an eigenvector corresponding to the negative eigenvalue $-\frac{L}{K}$ of M(K,0). Thus the corresponding 1-dimensional eigenspace has slope $\frac{f_2}{2f_3-f_2}$. (It is vertical if $2f_3-f_2=0$). Now since the line $x_1 + x_3 = K$ has slope -1 and $\frac{f_2}{2f_3-f_2} \notin [-1,0]$, therefore the 1-dimensional stable manifold $W^{S}(K,0)$ of (K,0) has trivial intersection with cl(S), namely at (K,0) itself (because of the stable manifold theorem, Theorem 2.6). On the other hand (1,0) is an eigenvector corresponding to the positive eigenvalue $-\frac{a_1L}{f_1K}$ of M(K,0). Therefore the unstable manifold $W^{U}(K,0)$ of (K,0) has slope 0 and has non-trival intersection with cl(S), namely in a whole separatrix (out of four for a saddle point), due to the positive invariance of cl(S) and the Hartman-Grobman theorem (see Section 2.9). Similar arguments apply to the case when $a_3 < 0$. This discussion and Proposition 3.4 settle the cases (i) - (iv).

Next consider the case when $a_1 = 0$. We will use the procedure described in Section 2.10 to show that (K,0) is a saddle node. Let

 $z_{1} = x_{1} - K$ $z_{2} = x_{3}$ $N = \begin{bmatrix} 1 & \frac{2f_{3} - f_{2}}{f_{2}} \\ 0 & 1 \end{bmatrix},$

and $Y = N^{-1}Z$. Then (3.20) takes the form (2.6). Moreover, by a straightforward calculation, the n in (2.9) is equal to $\frac{a_3L}{2f_1K^2}$. Since, as noted above, $\frac{f_2}{2f_3-f_2} \notin [-1,0]$, the assertions made under the cases (vii) and (viii) are easily verified. The case when $a_3 = 0$ is similar. q.e.d.

Knowing the local behavior of $\,\widetilde{\varphi}\,$ near each rest point, we can now describe the global dynamics of $\,\widetilde{\varphi}\,$ on cl(S) .

<u>Theorem 3.6</u>: For all $X \in cl(S)$, $\tilde{\phi}(X,t)$ <u>converges to a rest point of</u> $\tilde{\phi}$ <u>in</u> cl(S). More precisely, we have the following cases depending on the parameter regions:

(i)
$$a_1 > 0$$
, $a_3 > 0$:

ł

cl(S) is the disjoint union of the three stable manifolds:

$$W^{S}(K,0)$$
, $W^{S}(0,K)$, and $W^{S}(x_{1}^{*},x_{3}^{*})$. $W^{S}(K,0)$ and $W^{S}(0,K)$
are 2-dimensional, and $W^{S}(x_{1}^{*},x_{3}^{*})$ is 1-dimensional which acts
as a separatrix dividing the two regions of stability: $W^{S}(K,Q)$
and $W^{S}(0,K)$ of the rest points (K,0) and (0,K);

(ii)
$$a_1 < 0, a_3 < 0$$
:

 (x_1^*, x_3^*) is globally stable on cl(S) { {(K,0),(0,K)}; (iii) $a_1 > 0$, $a_3 < 0$:

(K,O) is globally stable on
$$cl(S) \setminus \{(0,K)\}$$
;

$$(iv) a_1 < 0, a_3 > 0$$
:

(0,K) is globally stable on
$$cl(S) \setminus \{(K,0)\}$$
;

$$(v) [a_1 > 0, a_3 = 0]$$
:

(K,0) is globally stable on cl(S)
$$\{(0,K)\}$$
;
(vi) $a_1 < 0$, $a_3 = 0$:

$$(0,K) \quad \underline{is \ globally \ stable \ on} \quad cl(S) \setminus \{(K,0)\};$$

$$(vii) \quad a_{1} = 0, \ a_{3} > 0;$$

$$(0,K) \quad \underline{is \ globally \ stable \ on} \quad cl(S) \setminus \{(K,0)\};$$

$$(viii) \quad a_{1} = 0, \ a_{3} < 0;$$

(K,0) is globally stable on
$$cl(S) \setminus \{(0,K)\}$$
.

<u>Proof</u>: In order to show that for all $X \in cl(S)$, $\tilde{\phi}(X,t)$ converges to a rest point of $\tilde{\phi}$ as t tends to $+\infty$, it suffices, according to the Poincare-Bendixson Theorem, to show that there is no periodic orbit nor rest points connecting orbit forming a loop in cl(S). Using Proposition 3.5 and simple geometric arguments, one could easily show that there are no rest points connecting orbits in cl(S).

On the other hand, since every periodic orbit must enclose a rest point, there can be no periodic orbit in cases (iii) – (viji). (Otherwise, the periodic orbit will have to enclose (K,O) or (O,K) contradicting the positive invariance of S .)

To show that there can be no periodic orbit enclosing the rest point (x_1^*, x_3^*) , one could use an index argument in case (ii) (since x_1^*, x_3^*) is a saddle in the case). As for case (i), let L_1 be the open line segment joining (K,0) and (x_1^*, x_3^*) (that is, end points are excluded) and let L_2 be the open line segment joining (0,K) and (x_1^*, x_3^*) . Also let T denote the open triangle with vertices (K,0), (x_1^*, x_3^*) , and (K,0). Then a straightforward calculation shows that the vector field defined by system (3.20) points into $S \setminus cl(T)$ on

 $L_1 U L_2$. For example, if $(x_1, x_3) \in L_1$, the dot product between the vector field at (x_1, x_3) and the outward normal (1,1+2c) of L_1 is given by:

$$\left(\frac{u^2}{(u+v)^2} - \frac{x_1}{K}\right)L + \left(\frac{v^2}{(u+v)^2} - \frac{x_3}{K}\right)L(1+2c) = \frac{2v(cv-u)L}{(u+v)^2}$$

which is negative. Thus, there can be no periodic orbit enclosing $\begin{pmatrix} x \\ 1 \end{pmatrix}$, $\begin{pmatrix} x \\ 3 \end{pmatrix}$. Hence, there is no periodic orbit in c1(S).

The rest of the assertions follow directly from the above and Proposition 3.5.

q.e.d.

3.5. Global Dynamics.

We will first describe the linearized stability properties and the local stability properties of the rest points of ϕ .

Proposition 3.7: The linearized stability properties of the rest points of ϕ are given by the following diagram:

		(بالمسموم المسلم الم		
		Rest Point		
Parameter Region		(K,0,0)	(0,0,K)	$(\mathbf{x}^{\star} \mathbf{x}^{\star} \mathbf{x}^{\star})$
gl	az	((,,0,,0))		(x_1, x_2, x_3)
+	+	-,-,-	-,-,-	-,-,+
-	-	-,-,+	-,-,+	-,-,-
+	-	-,-,-	·-,-,+	*
-	+	-,-,+	-,-,-	*
+	0	-,-,-	-,-,0	*
-	0	-,-,+	-,-,0	*
0	+	-,-,0	-,-,-	*
0	-	-,-,0	-,-,+	*

where "-,-,-" means that all three eigenvalues of the variational matrix at that rest point are negative, etc., as in Proposition 3.4.

<u>Proof</u>: This follows from Proposition 3.4 and the assumption (H6) in Section 3.1.

q.e.d.

Proposition 3.8: We have the following cases, depending on the parameter regions:

(i) $a_1 > 0$, $a_3 > 0$:

(K,0,0) and (0,0,K) are sinks (i.e. they are point attractors) (x_1^*, x_2^*, x_3^*) is a saddle with a 2-dimensional stable manifold and a 1-dimensional unstable manifold;

(ii)
$$a_1 < 0$$
, $a_3 < 0$:
 (x_1^*, x_2^*, x_3^*) is a sink, (K,0,0) is a saddle: its 2-dimensional
stable manifold $W^S(K,0,0)$ has trivial intersection with
 $c1(R_+^3)$, namely at H_{x_1} , and its 1-dimensional unstable mani-
fold $W^U(K,0,0)$ intersects $c1(R_+^3)$ in a whole branch (out of
two for the 1-dimensional unstable manifold), (0,0,K) is also
a saddle whose local phase portrait is similar to that of
(K,0,0);

(iii)
$$a_1 > 0$$
, $a_3 < 0$:

(K,0,0) <u>is a sink</u>, (0,0,K) <u>is a saddle whose local phase por-</u> trait is similar to that in case (ii);

(iv) $a_1 < 0$, $a_3 > 0$:

(0,0,K) <u>is a sink</u>, (K,0,0) <u>is a saddle whose local phase por-</u> trait is similar to that in case (ii);

 $(v) a_1 > 0, a_3 = 0$:

(K,0,0) is a sink, (0,0,K) is a saddle node: its strong stable manifold $W^{SS}(0,0,K)$, has trivial intersection with $c1(R_{+}^{3})$, mamely at H_{X3}, $c1(R_{+}^{3})$ is on the side of the strong stable manifold for which (0,0,K) is unstable;

(vi) $a_1 < 0$, $a_3 = 0$:

(K,0,0) is a saddle whose local phase portrait is similar to

that in case (ii), (0,0,K) is a saddle node: its strong stable manifold has trivial intersection with $cl(R_+^3)$, namely at H_{x_3} , $cl(R_+^3)$ is on the side of the strong stable manifold for which (0,0,K) is stable;

 $(vii) a_1 = 0, a_3 > 0$:

(0,0,K) is a sink, (K,0,0) is a saddle node whose local phase portrait is similar to that of (0,0,K) in case (v);

$$(viii) a_1 = 0, a_3 < 0:$$

(0,0,K) is a saddle whose local phase portrait is similar to that in case (ii), (K,0,0) is a saddle node whose local phase portrait is similar to that of (0,0,K) in case (vi).

<u>Proof</u>: The variational matrix of (3.1) at $X \in c1(R_+^3) \setminus \{(0,0,0)\}^{i}$ is given by

$$M(x_{1}, x_{2}, x_{3}) = \begin{bmatrix} M_{11} & M_{12} & M_{13} \\ M_{21} & M_{22} & M_{23} \\ M_{31} & M_{32} & M_{33} \end{bmatrix}$$
(3.31)

where

$$M_{12} = \frac{2f_1uv}{(u+v)^3} B(x) + \frac{u^2}{(u+v)^2} B'(x) - \frac{x_2+x_3}{x^2} D(x) - \frac{x_1}{x} D'(x) ,$$

$$M_{12} = \frac{f_2u(v-u)}{(u+v)^3} B(x) + \frac{u^2}{(u+v)^2} B'(x) + \frac{x_1}{x^2} D(x) - \frac{x_1}{x} D'(x) ;$$

$$\begin{split} \mathsf{M}_{13} &= -\frac{2f_{3}u^{2}}{(u+v)^{3}} \, \mathsf{B}(x) + \frac{u^{2}}{(u+v)^{2}} \, \mathsf{B}'(x) + \frac{x_{1}}{x^{2}} \, \mathsf{D}(x) - \frac{x_{1}}{x} \, \mathsf{D}'(x) \\ \mathsf{M}_{21} &= \frac{2f_{1}v(v-u)}{(u+v)^{3}} \, \mathsf{B}(x) + \frac{2uv}{(u+v)^{2}} \, \mathsf{B}'(x) + \frac{x_{2}}{x^{2}} \, \mathsf{D}(x) - \frac{x_{2}}{x} \, \mathsf{D}'(x) \\ \mathsf{M}_{22} &= \frac{f_{2}(u-v)^{2}}{(u+v)^{3}} \, \mathsf{B}(x) + \frac{2uv}{(u+v)^{2}} \, \mathsf{B}'(x) - \frac{x_{1}^{+}x_{3}}{x^{2}} \, \mathsf{D}(x) - \frac{x_{2}}{x} \, \mathsf{D}'(x) \\ \mathsf{M}_{23} &= \frac{2f_{3}u(u-v)}{(u+v)^{3}} \, \mathsf{B}(x) + \frac{2uv}{(u+v)^{2}} \, \mathsf{B}'(x) + \frac{x_{2}}{x^{2}} \, \mathsf{D}(x) - \frac{x_{2}}{x} \, \mathsf{D}'(x) \\ \mathsf{M}_{31}^{-} &= -\frac{2f_{1}v^{2}}{(u+v)^{3}} \, \mathsf{B}(x) + \frac{v^{2}}{(u+v)^{2}} \, \mathsf{B}'(x) + \frac{x_{3}}{x^{2}} \, \mathsf{D}(x) - \frac{x_{3}}{x} \, \mathsf{D}'(x) \\ \mathsf{M}_{32} &= -\frac{f_{2}v(u-v)}{(u+v)^{3}} \, \mathsf{B}(x) + \frac{v^{2}}{(u+v)^{2}} \, \mathsf{B}'(x) + \frac{x_{3}}{x^{2}} \, \mathsf{D}(x) - \frac{x_{3}}{x} \, \mathsf{D}'(x) \\ \mathsf{M}_{33} &= -\frac{2f_{3}uv}{(u+v)^{3}} \, \mathsf{B}(x) + \frac{v^{2}}{(u+v)^{2}} \, \mathsf{B}'(x) - \frac{x_{1}^{+}x_{2}}{x^{2}} \, \mathsf{D}(x) - \frac{x_{3}}{x} \, \mathsf{D}'(x) \\ \mathsf{M}_{33} &= -\frac{2f_{3}uv}{(u+v)^{3}} \, \mathsf{B}(x) + \frac{v^{2}}{(u+v)^{2}} \, \mathsf{B}'(x) - \frac{x_{1}^{+}x_{2}}{x^{2}} \, \mathsf{D}(x) - \frac{x_{3}}{x} \, \mathsf{D}'(x) \\ \mathsf{M}_{33} &= -\frac{2f_{3}uv}{(u+v)^{3}} \, \mathsf{B}(x) + \frac{v^{2}}{(u+v)^{2}} \, \mathsf{B}'(x) - \frac{x_{1}^{+}x_{2}}{x^{2}} \, \mathsf{D}(x) - \frac{x_{3}}{x} \, \mathsf{D}'(x) \\ \mathsf{M}_{33} &= -\frac{2f_{3}uv}{(u+v)^{3}} \, \mathsf{B}(x) + \frac{v^{2}}{(u+v)^{2}} \, \mathsf{B}'(x) - \frac{x_{1}^{+}x_{2}}{x^{2}} \, \mathsf{D}(x) - \frac{x_{3}}{x} \, \mathsf{D}'(x) \\ \mathsf{M}_{33} &= -\frac{2f_{3}uv}{(u+v)^{3}} \, \mathsf{B}(x) + \frac{v^{2}}{(u+v)^{2}} \, \mathsf{B}'(x) - \frac{x_{1}^{+}x_{2}}{x^{2}} \, \mathsf{D}(x) - \frac{x_{3}}{x} \, \mathsf{D}'(x) \\ \mathsf{M}_{33} &= -\frac{2f_{3}uv}{(u+v)^{3}} \, \mathsf{B}(x) + \frac{v^{2}}{(u+v)^{2}} \, \mathsf{B}'(x) - \frac{x_{1}^{+}x_{2}}{x^{2}} \, \mathsf{D}(x) - \frac{x_{3}}{x} \, \mathsf{D}'(x) \\ \mathsf{M}_{33} &= -\frac{2f_{3}uv}{(u+v)^{3}} \, \mathsf{B}(x) + \frac{v^{2}}{(u+v)^{2}} \, \mathsf{B}'(x) - \frac{x_{1}^{+}x_{2}}{x^{2}} \, \mathsf{D}(x) - \frac{x_{3}}{x} \, \mathsf{D}'(x) \\ \mathsf{M}_{33} &= -\frac{2f_{3}uv}{(u+v)^{3}} \, \mathsf{D}(x) + \frac{v^{2}}{(u+v)^{2}} \, \mathsf{D}'(x) - \frac{x_{1}^{+}x_{2}}{x^{2}} \, \mathsf{D}(x) - \frac{x_{3}}{x} \, \mathsf{D}'(x) \\ \mathsf{M}_{33} &= -\frac{2f_{3}uv}{(u+v)^{3}} \, \mathsf{D}'(x) + \frac{v$$

<u></u>

$$M(K,0,0)^{n} = \begin{bmatrix} -e & -e + \frac{a_{1}L}{f_{1}K} & -e + \frac{(f_{1}-2f_{3})L}{f_{1}K} \\ 0 & -\frac{a_{1}L}{f_{1}K} & \frac{2f_{3}L}{f_{1}K} \\ 0 & 0 & -\frac{L}{K} \end{bmatrix}$$
(3.32)

where

e = D'(K) - B'(K) > 0

(3.33)

۶,

١

P

by (H6) of Section 3.1. Hence the eigenvalues of M(K,0,0) are: -e,

$$-\frac{a_1L}{f_1K}$$
, and, $-\frac{L}{K}$. When $a_1 > 0$ (consequently, (K,0,0) is a saddle),
the 2-dimensional eigenspace corresponding to the two negative eigenvalues:
-e and $-\frac{L}{K}$ is spanned by the vectors (1,0,0) and (0, $-\frac{2f_3}{f_2}$, 1).
Since $-\frac{2f_3}{f_2} < 0$, the positive invariance of $cl(R_+^3)$ and the stable man-
ifold theorem (Theorem 2.6) imply that the 2-dimensional stable manifold
of (K,0,0) intersects $cl(R_+^3)$ at H_{X_1} only. Similarly, one can show -
that when $a_3 < 0$, the 2-dimensional stable manifold of (0,0,K) inter-
sects $cl(R_+^3)$ at H_{X_3} only. This and Proposition 3.7 settle the cases
(i) - (iv).

55

Next, consider the case when $a_1 = 0$. We will use the procedure described in Section 2.10 to show that (K,0,0) is a saddle node.

 $z_{1} = x_{1} - K$

Let

$$z_{2} = x_{2}$$

$$z_{3} = x_{3}$$

$$N = \begin{bmatrix} -1 & 1 & 0 \\ 1 & 0 & -\frac{2f_{3}}{f_{1}} \\ 0 & 0 & 1 \end{bmatrix},$$

and

 $Y = N^{-1}Z$. Then (3.1) takes the form (2.6). Moreover, a straightforward calculation shows that the n in (2.9) is equal to $\frac{a_3L}{2f_1K^2}$. This settles the cases (vii) and (viii). The case when $a_3 = 0$ is similar. q.e.d.

We can now describe the global dynamics of the flow ϕ on $cl(R_{+}^{3})^{-}.$

<u>Theorem 3.9</u>: For all $X \in cl(R_+^3)$, $\phi(X,t)$ <u>converges to a rest point of</u> ϕ <u>as</u> t <u>tends to</u> $+\infty$. <u>More precisely, we have the following cases</u> <u>depending on the parameters regions</u>:

(i)
$$a_1 = 0$$
, $a_3 = 0$:
 $cl(H(R_+^1))$ is the set of rest points of ϕ in
 $cl(R_+^3) \setminus \{(0,0,0)\}$. $cl(R_+^3) \setminus \{(0,0,0)\}$ is foliated by the
strong stable manifolds, $W^{SS}(H(c))$ ($c \in [0,+\infty]$), of these
rest points', where

$$W^{SS}(H(c)) \cap cl(R_{+}^{3}) = \{(x_{1}, x_{2}, x_{3}) \in cl(R_{+}^{3}) \setminus \{(0, 0, 0)\} :$$

$$x_{1} + \frac{1-c}{2} x_{2} - cx_{3} = 0\} \qquad (3.34)$$

<u>is 2-dimensional for all</u> $c \in (0, +\infty)$, $W^{SS}(H(0)) \cap cl(R_{+}^{3}) = W^{SS}(0,0,K) \cap cl(R_{+}^{3}) = H_{X_{3}}$, <u>and</u>, $W^{SS}(H(+\infty)) \cap cl(R_{+}^{3}) = W^{SS}(K,0,0) \cap cl(R_{+}^{3}) = H_{X_{1}}$ <u>are</u> <u>1-dimensional</u>.

(ii)
$$a_1 > 0$$
 , $a_3 > 0$:

 $c1(R_{+}^{3}) \setminus \{(0,0,0)\} \text{ is the disjoint union of } W^{S}(K,0,0),$ $W^{S}(0,0,K) \xrightarrow{\text{and}} W^{S}(x_{1}^{*},x_{2}^{*},x_{3}^{*}) \cdot W^{S}(K,0,0) \xrightarrow{\text{and}} W^{S}(0,0,K) \xrightarrow{\text{are}}$ $\underline{3-\text{dimensional}}, W^{S}(x_{1},x_{2},x_{3}) \xrightarrow{\text{is a 2-dimensional immersed sub-}}$ $\underline{\text{manifold of } c1(R_{+}^{3}) \setminus \{(0,0,0)\} \xrightarrow{\text{forming a separatrix surface}}$ $\underline{\text{which divides } c1(R_{+}^{3}) \setminus \{(0,0,0)\} \xrightarrow{\text{into the two regions of stab-}}$ $\underline{\text{iliy}}; W^{S}(K,0,0) \xrightarrow{\text{and}} W^{S}(0,0,K) \xrightarrow{\text{of the rest points}} (K,0,0)$ $\underline{\text{and}} (0,0,K);$

(iii)
$$a_1 < 0$$
, $a_3 < 0$:
 (x_1^*, x_2^*, x_3^*) is globally stable on $cl(R_+^3) \setminus cl(H_{x_1} \cup H_{x_3})$,
(K,0,0) is globally stable on H_{x_1} and (0,0,K) is globally
stable on H_{x_3} ;
(iv) $a_1 > 0$, $a_3 < 0$:

(K,0,0) is globally stable on
$$cl(R_+^3) \setminus cl(H_{x_3})$$
 and (0,0,K)
is globally stable on H_{x_3} ;

$$a_1 < 0$$
, $a_3 > 0$:
(0,0,K) is globally stable on $cl(R_+^3) \land cl(H_{x_1})$ and (K,0,0)
is globally stable on H_{x_1} ;

(vi)
$$a_1 > 0$$
, $a_3 = 0$:
(K,0,0) is globally stable on $cl(R_+^3) \land cl(H_{X_3})$ and (0,0,K)
is globally stable on H_v ;

`3

(vii)
$$a_1 < 0$$
, $a_3 = 0$:
(0,0,k) is globally stable on $cl(R_+^3) \land cl(H_{x_1})$ and (K,0,0)
is globally stable on H_{x_1} ;
(viii) $a_1 = 0$, $a_3 > 0$:
(0,0,K) is globally stable on $cl(R_+^3) \land cl(H_{x_1})$ and (K,0,0)
is globally stable on H_{x_1} ;
(ix) $a_1 = 0$, $a_3 < 0$:
(K,0,0) is globally stable on $cl(R_+^3) \land cl(H_{x_3})$ and (0,0,K)
is globally stable on H_{x_3} .
Proof: Case (i) is discussed in Freedman-Waltman (1978). It was shown
there that if we let $u_0 = u(0)$, $v_0 = v(0)$ and $c = u_0 / v_0$ then
 $u(t) = cv(t)$ for all $t \in R^1$. Furthermore, $x_1(t) - c^2x_3(t) + 0$, and
 $x_2(t) - 2cx_3(t) + 0$ as $t + +\infty$. Hence,
($x_1(t), x_2(t), x_3(t)$) + $\left(\frac{c^2k}{(1+c)^2}, \frac{2cK}{(1+c)^2}, \frac{K}{(1+c)^2}\right)$
as $t + +\infty$. (Note that the assumption "death is certain":
 $\int_0^{t\infty} \frac{D(x)}{x} dx = +\infty$ made in Freedman-Waltman (1978) is automatically satisfied under the hypotheses (H1) - (H6) in Section 3.1 because,
 $\lim_{x\to\infty} \frac{D(x)}{x} = \frac{1}{K} > 0$.) The assertions made under case (i) are just a
rephrase of these results in geometric language.

We will now restrict ourselves to the case when $(a_1,a_3) \neq (0,0)$. Recall the following properties of the omega-limit set, w(X), of a precompact non-negative orbit, $0^+(X)$, of a flow ϕ :

- (a) w(X) is compact, connected, invariant and chain recurrent(Proposition 2.5) and
- (b) $\phi(X,t)$ approaches w(X) as $t \rightarrow +\infty$.

For all $X \in cl(R_+^3) \setminus \{(0,0,0)\}$, $w(X) \in cl(U)$ by the Proposition 3.1. According to the Theorem 3.6, the only compact, connected, invariant, chain recurrent subsets of cl(U) are the singleton rest point sets. Thus $\phi(X,t)$ converges to a rest point in cl(U).

The assertions made under the cases (ii) - (ix) follow from the above observation and Proposition 3.8.

q.e.d.

59

<u>Remarks</u>: (i) It is clear that one could weaken the assumption (H3) in Section 3.1 of strictly increasing B and D somewhat without changing the statement of the Theorem 3.9 nor its proof.

(ii) One could also add in an equation for the carrying capacity K of the form: $\dot{K} = G(K)$, as in Beck (1982). If we impose enough conditions on G so that $K_0 > 0$ is globally stable over the positive K axis then the asymptotic behavior described in Theorem 3.9 remains valid with K replaced by K_0 .

3.6 Discussion.

21

The biological interpretation of the parameter regions in Theorem 3.9 are as follows:

- (i) $a_1 = 0$, $a_3 = 0$: selection values are cli equal,
- (ii) $a_1 > 0$, $a_3 > 0$: heterozygote inferiority,
- (iii) $a_1 < 0$, $a_3 < 0$: heterozygote superiority,
- .(iv) $a_1 > 0$, $a_3 < 0$: incomplete dominance,
 - (v) $a_1 < 0$, $a_3 > 0$: incomplete dominance,
- (vi) $a_1 > 0$, $a_3 = 0$: selection against the dominant allele a,
- (vii) $a_1 < 0$, $a_3 = 0$: selection against the recessive allele A,
- (viii) $a_1 = 0$, $a_3 > 0$: selection against the dominant allele A, and,

(ix) $a_1 = 0$, $a_3 < 0$: selection against the recessive allele a.

Thus the results stated in Theorem 3.9 are directly analogous to the discrete case (see Chapter 3 of Roughgarden (1979)). The idea of convergence to rest points for a one-locus, n-allele genetic model is not new. One could refer to Butler-Freedman-Waltman (1982) and Hadeler-Glas (1983) for models along similar lines, and to Losert-Akin (1984) for the Fisher-Wright-Haldane model. The usefulness of using a continuous model like system (3.1) lies in the possibility of adding other populations into the system in the form of differential equations. In the next chapter, the genotypic effects of predator-prey interactions will be studied using these ideas.

CHAPTER 4

A PREDATOR-PREY MODEL CONSISTING OF THREE PREY GENOTYPES WITH FERTILITY/VIABILITY DIFFERENCES

In this chapter a model of a predator-prey interaction, where the prey population consists of three genotypes and is modelled by system (3.1), is proposed. Sufficiency conditions leading to the evolution of pure strains are given. This extends results of Freedman-Waltman (1978, 1982). As well, conditions are derived which guarantee the persistence of all components of the system.

4.1 Introduction and the Model.

In this chapter, the model of the previous chapter is extended to include a predator. More specifically, the model is given by:

$$\dot{x}_{1} = \frac{u^{2}}{(u+v)^{2}} B(x) - \frac{x_{1}}{x} [D(x) + yP_{1}(x)]$$

$$\dot{x}_{2} = \frac{2uv}{(u+v)^{2}} B(x) - \frac{x_{2}}{x} [D(x) + yP_{2}(x)]$$

$$\dot{x}_{3} = \frac{v^{2}}{(u+v)^{2}} B(x) - \frac{x_{3}}{x} [D(x) + yP_{3}(x)]$$

$$\dot{y} = y[-s + k \sum_{i=1}^{3} \frac{x_{i}}{x} P_{i}(x)]$$
(4.1)

Ì

where x, given by (3.3), is the total prey population and y is the
predator population, u and v are given by (3.2) and s and k are assumed to be positive.

The genetic component of this model arises by considering a one-locus, two-allele problem as given by system (3.1). The ecological component arises by imbedding an intermediate predator-prey model with continuous natural birth and death processes in a model which accounts for different predator functional responses (given by $P_i(x)$ (i=1,2,3)) by the predator on the different prey genotypes x_i (i=1,2,3) . For a discussion of intermediate predator-prey models, see Chapter 4 of Freedman (1980) and the literature cited there.

Besides the assumptions (H1) - (H6) made in Section 3.1 on the birth and death rates, we make the following assumptions on the predator functional responses:

(H7) $P_i : cl(R_+^1) \rightarrow cl(R_+^1)$ are C^2 (i=1,2,3), (H8) $P_i(0) = 0$ (i=1,2,3) (H9) $P_i(x) > 0$ (i=1,2,3) for all $x \in cl(R_+^1)$.

Assumption (H7) is sufficient to ensure existence and uniqueness of initial value problems for $t \ge 0$. (H8) implies that there is no predation in the absence of the prey population. (H9) states that the predator functional responses are strictly increasing as functions of the number of prey.

The remainder of this chapter is organized as follows: In Section 4.2, we construct the flow on $cl(R_{+}^{4})$ defined by system (4.1)

63

¢

and list some of its properties. The boundary rest points of this flow and their stability are considered in Section 4.3. In Section 4.4, conditions leading to the evolution of pure strains and the existence of a globally stable boundary rest point are given. The classification of the parameter values into persistence and non-persistence cases is discussed in Section 4.5. We finish with a short discussion of the results in Section 4.6.

4.2 The Associated Flow.

In this section we will show that system (4.1) defines a continuous semi-flow on $cl(R^4)$ which becomes a smooth (C^1) local flow when restricted to R_+^4 . We will also show that it is A-dissipative and discuss its boundary invariant sets.

As in system (3.1), system (4.1) is not defined when x = 0. Let us redefine system (4.1) as follows. We define system (4.2) to be

$$\dot{x}_{1} = \frac{u^{2}}{(u+v)^{2}} B(x) - \frac{x_{1}}{x} [D(x) + yP_{1}(x)]$$

$$\dot{x}_{2} = \frac{2uv}{(u+v)^{2}} B(x) - \frac{x_{2}}{x} [D(x) + yP_{2}(x)], \quad x > 0$$

$$\dot{x}_{3} = \frac{v^{2}}{(u+v)^{2}} B(x) - \frac{x_{3}}{x} [D(x) + yP_{3}(x)]$$

$$\dot{y} = y[-s + k \sum_{i=1}^{3} \frac{x_{i}}{x} P_{i}(x)]$$

$$\dot{x}_{i} = 0 \quad (i = 1, 2, 3) \quad , \quad x = 0$$

$$(4.2)$$

 $\dot{y} = -sy$

or

$$\dot{x}_{i} = F_{i}(x_{1}, x_{2}, x_{3}, y) \quad (i = 1, 2, 3)$$
$$\dot{y} = F_{4}(x_{1}, x_{2}, x_{3}, y) = G(x_{1}, x_{2}, x_{3}, y)$$

Let

$$F = (F_1, F_2, F_3, F_4) .$$
 (4.3)

Clearly,
$$F_i \in Lip$$
 (i = 1,...,4) on $cl(R_+^4)$. For example,
consider the two points (x_1, x_2, x_3, y) and $(0, 0, 0, \overline{y})$ where
 $x = x_1 + x_2 + x_3 > 0$. Then

$$|G(x_{1}, x_{2}, x_{3}, y) - G(0, 0, 0, \overline{y})| \le s |y - \overline{y}| + ky \sum_{i=1}^{3} P_{i}(x)$$

$$\le s |y - \overline{y}| + ky \sum_{i=1}^{3} P'_{i}(z_{i})x$$

for some $0 \le z_i \le x$ (i = 1,2,3) and hence $G \in Lip$.

<u>Proposition 4.1</u>: System (4.2) defines a continuous semi-flow on $cl(R_{+}^{4})$ which becomes a C^{1} local flow when restricted to R_{+}^{4} . It is A-dissipative where

$$A = \{(x_1, x_2, x_3, y) \in cl(R_+^4) : x \le K \text{ and } kx + y \le \frac{k}{s} (M+sK)\}$$
(4.4)

and M = max {B(x) - D(x) : $x \in [0,K]$ }. The edges and faces H_{x_1} , H_{x_3} , H_y ,

 $\begin{array}{l} H_{x_{1},y}, H_{x_{3},y}, H_{x_{1},x_{2},x_{3}} & \underline{ are invariant under the flow and the vector field} \\ F, \underline{ defined in (4.3), points into } R_{+}^{4} & \underline{ on } b(R_{+}^{4}) \setminus cl(H_{x_{1},y} \cup H_{x_{2},y} \cup H_{x_{1},x_{2},x_{3}}). \end{array}$

<u>Proof</u>: Using Proposition (2.1) it is easily seen that $cl(R_+^4)$ is positively invariant. System (4.2) is A-dissipative follows from

$$\dot{x} = B(x) - D(x) - y \sum_{i=1}^{3} \frac{x_i}{x} P_i(x) \le B(x) - D(x)$$

$$kx + y = k[B(x) - D(x)] - sy$$

and Proposition 2.2. The rest of the assertions can be easily verified, similar to the proof of Proposition 3.1.

q.e.d.

<u>Remarks</u>: (i) The invariance of H_{x_1} can be interpreted as: if there were no "a" gamete nor predator to start with, then there would not be any "a" gamete nor predator for any other time. The invariance of the other boundary invariant sets can be interpreted similarly.

(ii) The global dynamics of system (4.1) on the invariant set $cl(H_{x_1,x_2,x_3})$ is given by Theorem 3.9. Its dynamics on $cl(H_{x_1,y})$ (i = 1,3) is given by the intermediate predator-prey model:

$$\dot{x}_{i} = B(x_{i}) - D(x_{i}) - yP_{i}(x_{i})$$

$$\dot{y} = y[-s + kP_{i}(x_{i})]$$
(4.5)

4.3 Boundary Rest Points and Their Stability.

In this section those rest points of system (4.1) lying on $b(R_{+}^{4})$ and their stability properties will be considered. Clearly, $E_{0}(0,0,0,0)$ is an unstable rest point (because $\dot{y}\Big|_{\substack{x=0 \ y>0}} < 0$ and $\dot{x}\Big|_{\substack{0 < x < 1 \ y=0}} > 0$). In fact, we have the following proposition.

<u>Proposition 4.2</u>: $E_0 \notin w(x_1, x_2, x_3, y)$ for any $(x_1, x_2, x_3, y) \in R_+^4$. <u>Proof</u>: Since in a neighbourhood of E_0 in $cl(R_+^4)$, we have,

$$\dot{x} \ge B(x) - D(x) - yP(x) \ge k_1 x$$

 $\dot{y} \le y[-s+kP(x)] \le -k_2 y$ for some $k_1, k_2 > 0$,

and the flow on $cl(H_y)$ is given by y = -sy, where $P(x) = max \{P_i(x) : 1 \le i \le 3\}$, the result follows immediately from Proposition 2.15.

<u>Corollary 4.3</u>: Let $(0,0,0,y) \in cl(H_y)$. Then $(0,0,0,y) \notin w(x_1,x_2,x_3,y)$ for any $(x_1,x_2,x_3,y) \in R_+^4$.

<u>Proof</u>: Suppose not, since $w(x_1, x_2, x_3, y)$ is compact, connected and invariant, the global stability of E_0 on $cl(H_y)$ implies that $E_0 \in w(x_1, x_2, x_3, y)$ contradicting Proposition 4.2.

q.e.d.

Clearly, $E_{1}(K,0,0,0)$ (resp. $E_{2}(0,0,K,0)$) is the only rest point in $H_{x_{1}}$ (resp. $H_{x_{3}}$). Let $b_{i} = -s + kP_{i}(K)$ (i = 1,2,3). (4.6)

A necessary and sufficient condition for $H_{x_1,y}$ (resp. $H_{x_3,y}$) to contain a rest point is b_1 (resp. b_3) > 0. Moreover, (H9) in Section 4.1 implies that these rest points, if they exist, are unique. Let us denote them by $E_3(\overline{x_1},0,0,\overline{y_1})$ and $E_4(0,0,\overline{x_3},\overline{y_3})$.

Rest points lying in H_{x_1,x_2,x_3} are of the form $E_5(x_1^*,x_2^*,x_3^*,0) = (H(c),0)$ where H(c) is given by (3.10). According to Proposition 3.3, in the case when $a_1 = a_3 = 0$, (H(c),0) is a rest point for all $c \in (0,+\infty)$. Hence there is a curve of rest points. In the other cases, E_5 exists if and only if $a_1a_3 > 0$; and if it exists, then it is unique and is given by (H(c),0) with $c = a_3/a_1$.

The linearized stability of E_h (h = 1,...,5) is governed by the sign of the real parts of the eigenvalues of the variational matrix $M = [M_{ij}]_{1 \le i, j \le 4}$ of system (4.1) evaluated at E_h , where $M_{ij} = \frac{\partial F_i}{\partial z_j}$, $z_i = x_i$ (i = 1,2,3), and, $z_4 = y$.

Recall the definition of a_i (i = 1,3), L, and, e from (3.16), (3.22), and, (3.33) in Chapter 3.

Clearly,

68

á

$$M(E_{1}) = \begin{bmatrix} -e & -e + \frac{a_{1}L}{f_{1}K} & -e + \frac{(f_{1}-2f_{3})L}{f_{1}K} & -P_{1}(K)_{p} \\ 0 & -\frac{a_{1}L}{f_{1}K} & \frac{2f_{3}L}{f_{1}K} & 0 \\ 0 & 0 & -\frac{L}{K} & 0 \\ 0 & 0 & 0 & b_{1} \end{bmatrix}$$
(4.7)

69

$$\begin{split} \mathsf{M}(\mathsf{E}_1) & \text{has eigenvalues: -e, } -\frac{a_1\mathsf{L}}{f_1\mathsf{K}}, -\frac{\mathsf{L}}{\mathsf{K}}, \text{and, } \mathsf{b}_1 & \text{Since two of} \\ \text{these, -e and } \sqrt{\frac{\mathsf{L}}{\mathsf{K}}} & \text{are negative, } \mathsf{E}_1 & \text{has a stable manifold of at} \\ \text{least two dimensions. The actual dimensions of } \mathsf{W}^\mathsf{S}(\mathsf{E}_1) & \text{and } \mathsf{W}^\mathsf{U}(\mathsf{E}_1) \\ \text{depend on the signs of } a_1 & \text{and } b_1 & \text{Analogous statements can be made} \\ \text{for. } \mathsf{E}_2 & \text{.} \end{split}$$

 $M(E_3)$, is given by

$$\begin{split} \mathsf{M}_{11} &= \mathsf{B}'(\overline{x}_1) - \mathsf{D}'(\overline{x}_1) - \overline{y}_1 \mathsf{P}_1(\overline{x}_1) \quad , \\ \mathsf{M}_{12} &= \mathsf{B}'(\overline{x}_1) - \mathsf{D}'(\overline{x}_1) - \frac{f_2 \mathsf{B}(\overline{x}_1)}{f_1 \overline{x}_1} + \frac{\mathsf{D}(\overline{x}_1)}{\overline{x}_1} - \overline{y}_1 \mathsf{P}_1'(\overline{x}_1) + \frac{\overline{y}_1 \mathsf{P}_1(\overline{x}_1)}{\overline{x}_1} \, , \\ &= \mathsf{M}_{13} = \mathsf{B}'(\overline{x}_1) - \mathsf{D}'(\overline{x}_1) - \frac{2f_3 \mathsf{B}(\overline{x}_1)}{f_1 \overline{x}_1} + \frac{\mathsf{D}(\overline{x}_1)}{\overline{x}_1} - \overline{y}_1 \mathsf{P}_1'(\overline{x}_1) + \frac{\overline{y}_1 \mathsf{P}_1(\overline{x}_1)}{\overline{x}_1} \, , \\ \mathsf{M}_{14} &= -\mathsf{P}_1(\overline{x}_1) \, , \, \mathsf{M}_{21} = 0 \, , \, \mathsf{M}_{223} = \frac{f_2 \mathsf{B}(\overline{x}_1)}{f_1 \overline{x}_1} - \frac{\mathsf{D}(\overline{x}_1)}{\overline{x}_1} - \frac{\overline{y}_1 \mathsf{P}_2(\overline{x}_1)}{\overline{x}_1} \, , \\ \mathsf{M}_{23} &= \frac{2f_3 \mathsf{B}(\overline{x}_1)}{f_1 \overline{x}_2} \, , \, \, \mathsf{M}_{24} = 0 \, , \, \mathsf{M}_{31} = 0 \, , \, \mathsf{M}_{32} = 0 \, , \end{split}$$

 $\left\{ \right\}$

$$M_{33} = -\frac{D(\overline{x}_{1}) + \overline{y}_{1}P_{3}(\overline{x}_{1})}{\overline{x}_{1}} , \quad M_{34} = 0 , \quad M_{41} = k\overline{y}_{1}P_{1}'(\overline{x}_{1}) ,$$

$$M_{42} = k\overline{y}_{1} \left[P_{1}'(\overline{x}_{1}) - \frac{P_{1}(\overline{x}_{1})}{\overline{x}_{1}} + \frac{P_{2}(\overline{x}_{1})}{\overline{x}_{1}} \right] ,$$

$$M_{43} = k\overline{y}_{1} \left[P_{1}'(\overline{x}_{1}) - \frac{P_{1}(\overline{x}_{1})}{\overline{x}_{1}} + \frac{P_{3}(\overline{x}_{1})}{\overline{x}_{1}} \right] , \quad M_{44} = 0 . \qquad (4.8)$$

After interchaning the second and fourth row and columns respectively of $M(E_3)$ and calling the resulting matrix \overline{M} , we see that $\overline{M}_{31} = \overline{M}_{32} = \overline{M}_{41} = \overline{M}_{42} = \overline{M}_{43} = 0$. Analysis of this matrix shows that the eigenvalues corresponding to vectors lying in $H_{x_1,y}$ are such that the sign of their real parts is the same as the sign of M_{11} . The other two eigenvalues of $M(E_3)$ are given by M_{22} and M_{33} . Clearly $M_{33} < 0$ but M_{22} has undetermined sign. Analogous statements can be made regarding E_4 . For future reference, we denote $M_{22}(E_3)$ and $M_{22}(E_4)$ by d_1 and d_3 respectively. That is, we define

$$d_{i} = \frac{1}{\overline{x}_{i}} \left[\frac{f_{2}B(\overline{x}_{i})}{f_{i}} - D(\overline{x}_{i}) - \overline{y}_{i}P_{2}(\overline{x}_{i}) \right] \quad (i = 1,3) \quad . \quad (4.9)$$

The stability of $E_5(x_1^*, x_2^*, x_3^*, 0)$ on H_{x_1, x_2, x_3} is given by Proposition 3.7. To compute the stability of E_5 in the y-direction we note that

Э

$$d = -s + \frac{k}{K} \sum_{i=1}^{3} x_{i}^{*} P_{i}(K)$$
 (4.10)

70 د

ß

is an eigenvalue of $M(E_5)$ corresponding to eigenvectors in y-direction, and hence E_5 is asymptotically stable or unstable in the ydirection as d is negative or positive. The case when $f_1 = f_2 = f_3$ is of special interest. For each $E_5 = (H(c), 0)$ ($c \in (0, +\infty)$), $M(E_5)$ has one zero and two negative eigenvalues corresponding to eigenvectors on H_{x_1, x_2, x_3} . As for the y-direction, we have the following proposition.

Proposition 4.4: Let $f_1 = f_2 = f_3$ and $b_1, b_3 > 0$. If

$$P_2(K) > \frac{s}{k} - \sqrt{(P_1(K) - \frac{s}{k})(P_3(K) - \frac{s}{k})}$$
 (4.11)

then for all
$$(x_{10}, x_{20}, y_{30}, y_0) \in \mathbb{R}^4_+$$
, lim $y(t) > 0$.

0

A STATE

<u>Proof</u>: Let $h_i = P_i(K) - \frac{s}{k}$ (i = 1,2,3). $b_i > 0$ (i = 1,3) implies that $h_i > 0$ (i = 1,3). Moreover, $h_2 > -\sqrt{h_1 h_3}$ by (4.11). According to Lemma 4.5 (see below), we have,

$$d = -s + \frac{k}{(1+c)^2} [c^2 P_1(K) + 2c P_2(K) + P_3(K)]$$
$$= \frac{1}{(1+c)^2} [c^2 h_1 + 2c h_2 + h_3] > 0$$

for all $c \in [0, +\infty]$, where d is defined in (4.10). Hence the Hardy-Weinberg manifold, $cl(H(R_+^1))$ is normally hyperbolic. Suppose that $\lim_{t \to +\infty} y(t) = 0$. Using an argument similar to the Butler-McGehee lemma (Lemma 2.14) and its extension (Proposition 2.16), one shows that $w = w(x_{10}, x_{20}, x_{30}, y_0)$ must intersect $cl(H_{x_1, x_2, x_3})$ at a point other than $cl(H(R_+^1))$. The global dynamics of system (4.1) on $cl(H_{x_1, x_2, x_3})$ as given in case (i) of Theorem 3.9 implies that $E_0 \in w$ or w is unbounded. In either case we have a contradiction.

q.e.d.

<u>Remark</u>: For $b_1, b_3 > 0$, (4.11) is satisfied if $b_2 > 0$.

Lemma 4.5: Let $h_1, h_2, h_3 \in \mathbb{R}$. Then

÷.

$$I(c) = \frac{1}{(1+c)^2} [c^2h_1 + 2ch_2 + h_3] > 0 \quad for all \quad (c \in [0, +\infty)]$$

if and only if $h_1, h_3 > 0$ and $h_2 > -\sqrt{h_1h_3}$.

<u>Proof</u>: Since $I(0) = h_3$ and $I(+\infty) = h_1$, the conditions $h_1 > 0$ and $h_3 > 0$ are clearly necessary. Now for $c \in (0, +\infty)$, I(c) > 0 if and only if $c^2h_1 + 2ch_2 + h_3 > 0$, and,

$$c^{2}h_{1} + 2ch_{2} + h_{3} = (c\sqrt{h_{1}} - \sqrt{h_{3}})^{2} + 2c(h_{2} + \sqrt{h_{1}h_{3}}).$$

This completes the proof of the lemma.

q.e.d.

4.4 Evolution to Pure Strains.

In this section, criteria will be given which guarantee the disappearance of one of the gamete types. As well, sufficiency conditions ensuring E_3 or E_4 to be globally stable over R_+^4 will be considered.

From the definition of u and v in (3.2),

$$\dot{u} = u \left[\frac{f_1 u + f_2 v}{(u+v)^2} B(x) - \frac{1}{x} D(x) \right] - \frac{y}{x} [f_1 x_1 P_1(x) + \frac{1}{2} f_2 x_2 P_2(x)]$$
(4.12)

$$\dot{v} = v \left[\frac{f_3 v^4 + f_2 u}{(u+v)^2} B(x) - \frac{1}{x} D(x) \right] - \frac{y}{x} \left[f_3 x_3 P_3(x) + \frac{1}{2} f_2 x_2 P_2(x) \right]$$

Therefore,

. Ĉŝ

$$\frac{\dot{v}}{v} - \frac{\dot{u}}{u} = -\frac{a_1 u - a_3 v}{(u + v)^2} B(x) - \frac{y}{xuv} \left[\frac{1}{2} f_1 f_2 (P_2(x) - P_1(x)) x_1 x_2 + f_1 f_3 (P_3(x) - P_1(x)) x_1 x_3 + \frac{1}{2} f_2 f_3 (P_3(x) - P_2(x)) x_2 x_3\right] . \quad (4.13)$$

Theorem 4.6: Assume that

(i)
$$f_1 = f_2 = f_3$$
,
(ii) $P_1(x) \le P_2(x) \le P_3(x)$ for all $x \in [0,K]$,
iii) $-s + kP_1(K) > 0$, and,

(iv) there exist
$$k_1, k_3 \ge 0$$
 not both zero such that
 $P_2(x) - P_1(x) \ge k_1 x$ and $P_3(x) - P_2(x) \ge k_3 x$ for all. $x \in [0, K]$.

Then for all
$$(x_{10}, x_{20}, x_{30}, y_0) \in \mathbb{R}^4_+$$
, $x_2(t)$, $x_3(t) \rightarrow 0$ as $t \rightarrow +\infty$

Proof: The proof proceeds in four steps.

<u>Step 1</u>. <u>lim</u> y(t) > 0. (ii) and (iii) implies $b_i > 0$ (i=1,2,3). Hence this follows from Proposition 4.4.

<u>Step 2</u>. One can assume that $x(t) \le K$ for all $t \ge 0$. This is because system (4.1) is A-dissipative, where A is given in (4.4), and that there are no invariant sets in $\{(x_1, x_2, x_3, y) \in cl(R_+^4) : x = K\}$ other than those that lie in $cl(H_{x_1, x_2, x_3})$.

<u>Step 3</u>. Points of the form $(\hat{x}_1, \hat{x}_2, \hat{x}_3, \hat{y})$ where $\hat{x}_2 > 0$, and, $\hat{x}_1 = 0$ or $\hat{x}_3 = 0$ are not in $w(x_{10}, x_{20}, x_{30}, x_{40})$. Consider the case when $\hat{x}_1 = 0$ and $\hat{x}_2 > 0$. Then $F_1(\hat{x}_1, \hat{x}_1, \hat{x}_2, \hat{y}_3) > 0$ and we can apply Proposition 2.13. Similarly we can deal with the case when $\hat{x}_3 = 0$ and $\hat{x}_2 > 0$.

Step 4. Since $a_1 = a_3 = 0$, according to (4.13), we have,

$$\frac{\dot{v}}{v} - \frac{\dot{u}}{u} \leq -\frac{y}{uv} \cdot [\frac{1}{2} k_1 x_1 x_2 + (k_1 + k_3) x_1 x_3 + \frac{1}{2} k_3 x_2 x_3]$$
$$\leq -[\frac{k_1}{v} x_1 y + \frac{k_3}{u} x_3 y] \leq -\frac{1}{K} [k_1 x_1 y + k_3 x_3 y]$$

since $0 \le u, v \le K$.

Case 1: $k_1 > 0$. Then, by integrating,

$$v(t) \leq \frac{v_0}{u_0} u(t) \exp \left[-\frac{k_1}{K} \int_0^t x_1(\tau) y(\tau) d\tau\right]$$
 (4.14)

If the last integral diverges, then $\lim_{t\to+\infty} v(t) = 0$. Otherwise, $(x_1y)(t)$ is in $L^1[0,+\infty)$. Since $\frac{d}{dt}[x_1y(t)]$ is bounded, $(x_1y)(t)$ is uniformly continuous. As uniformly continuous $L^1[0,+\infty)$ functions have limit zero when $t \to +\infty$, this shows that $\lim_{t\to+\infty} (x_1y)(t) = 0$. $\lim_{t\to+\infty} y(t) > 0$ implies $\lim_{t\to+\infty} x_1(t) = 0$ and hence $\lim_{t\to+\infty} x_2(t) = 0$, by $\lim_{t\to+\infty} u(t) = 0$ and consequently $\lim_{t\to+\infty} v(t) = 0$, by (4.14). But this contradicts Corollary 4.3.

Case 2: $k_3 > 0$. This is similar to Case 1.

q.e.d.

<u>Remark</u>: The case when $P_2 = P_3$ and $k_1 > 0$ was studied in Freedman-Waltman (1978) and the case when $P_1 = P_2$ and $k_3 > 0$ was studied in Freedman-Waltman (1982).

<u>Corollary 4.7</u>: If, in addition, we assume that E_3 exists and is globally stable (resp. globally exponentially stable) on $H_{x_1,y}$, then it is globally attracting (resp. globally stable) on R_+^4 .

<u>Proof</u>: The global attractivity of E_3 follows from Theorem 4.6, <u>lim</u> y(t) > 0 and a chain recurrence argument. In the case when E_3 $t \rightarrow t \infty$ is exponentially stable on $H_{x_1,y}$ (i.e. the two eigenvalues of $M(E_3)$ corresponding to eigenvectors on $H_{x_1,y}$ have negative real parts), the dimension of $W^{SS}(E_3)$ is at least three. Therefore $W^C(E_3)$ must be at most 1-dimensional and, according the Theorem 4.6, E_3 is stable on $W^C(E_3) \cap R_+^4$. The stability of E_3 on R_+^4 then follows from the Palis-Takens linearization theorem (Theorem 2.10).

<u>Remark</u>: Conditions for E_3 to be globally stable on $H_{x_1,y}$ can be found in Hsu (1978) and Cheng-Hsu-Lin (1981).

Theorem 4.8: Assume that

(i)
$$f_1 \ge f_2 \ge f_3$$
 and they are not all equal, and,
(ii) $P_1(x) \le P_2(x) \le P_3(x)$ for all $x \in [0, K']$, for some $K' > K$.
Then for all $(x_{10}, x_{20}, x_{30}, y_0) \in R_+^4$, $x_2(t), x_3(t) \to 0$ as $t \to +\infty$.

<u>Proof</u>: Again, since system (4.1) is A-dissipative, we can assume that $x(t) \le K'$ for all $t \ge 0$. Since $a_1 \ge 0$ and $a_3 \le 0$, from (4.13), we have

$$\frac{\dot{v}}{v} - \frac{\dot{u}}{u} \le -\frac{a_1 u - a_3 v}{(u+v)^2} B(x) \qquad (4.15)$$

If $a_1 > 0$ (and $a_3 \le 0$), then from (4.15),

$$\frac{v}{v} - \frac{u}{u} \leq - \frac{a_1 u}{(u+v)^2} B(x)$$

q.e.d.

Integration yields:

ĸ.

$$v(t) \leq \frac{v_0}{u_0} u(t) \exp \left[- \int_0^t \frac{a_1 u(\tau)}{(u(\tau) + v(\tau))^2} B(x(\tau)) d\tau \right] .$$

If $\lim_{t \to +\infty} u(t) = 0$, then $\lim_{t \to +\infty} v(t) = 0$ which in turn implies $\lim_{t \to +\infty} x(t) = 0$ contradicting Corollary 4.3. Therefore $\lim_{t \to +\infty} u(t) = \hat{u} > 0$. This implies the integral

$$\int_{0}^{+\infty} \frac{a_1 u(\tau)}{(u(\tau) + v(\tau))^2} B(x(\tau)) d\tau = +\infty \text{ and hence } \lim_{t \to +\infty} v(t) = 0$$

Suppose now that $a_1 = 0$ and hence $a_3 < 0$. Then from (4.15)

$$\frac{\dot{v}}{v} - \frac{\dot{u}}{u} \leq \frac{a_3 v}{(u+v)^2} B(x)$$

and

$$v(t) \leq \frac{v_0}{u_0} u(t) \exp\left[\int_0^t \frac{a_3 v(\tau)}{(u(\tau)+v(\tau))^2} B(x(\tau)) d\tau\right]$$

If this last integral diverges, then again $\lim_{t\to+\infty} v(t) = 0$. Suppose not, i.e. suppose $z(t) = \frac{v(t)}{(u(t)+v(t))^2} B(x(t))$ is in $L^1[0,+\infty)$. By Corollary 4.3, x(t) is bounded away from zero. Therefore, $\frac{d}{dt} z(t)$ is bounded and so z(t) is uniformly continuous. Thus $\lim_{t\to+\infty} z(t) = 0$. Since $\lim_{t\to+\infty} B(x(t)) > 0$, we have, $\lim_{t\to+\infty} v(t) = 0$. <u>Corollary 4.9</u>: If, in addition, we assume E_3 exists and is globally stable (resp. globally exponentially stable) on $H_{x_1,y}$, then it is globally attracting (resp. globally stable) on R_+^4 .

<u>Proof</u>: Let $(x_{10}, x_{20}, x_{30}, y_0) \in R_+$. Then $w = w(x_{10}, x_{20}, x_{30}, y) \in Cl(H_{x_1, y})$, by Theorem 4.8. We first show that $E_1 \notin w$. Suppose not, i.e. suppose $E_1 \in w$, then by Proposition 2.16, $w \cap W^{CS}(E_1)$ contains a point other than E_1 . Since $W^{CS}(E_1) \cap cl(R_+^4) \in cl(H_{x_1, x_2, x_3})$, this means that w contains a point in $cl(H_{x_1})$ other than E_1 , which is a contradiction. Therefore $E_1 \notin w$. The global dynamics on $cl(H_{x_1, y})$ and the chain recurrence of w imply that $w = \{E_3\}$. Hence $E_3 \in S$ globally attracting on R_+^4 .

The proof of the stability of E_3 on R_+^4 when it is exponentially stable on H_1 is similar to that given in the proof of Corollary 4.7.

q.e.d.

<u>Remark</u>: In Theorems 4.6 and 4.8, one could interchange the roles of the gamete types A and a and obtain conditions for which $x_1(t), x_2(t) \neq 0$ as $t \neq +\infty$. The same observation applies to Corollaries 4.7 and 4.9.

4.5 Persistence and Non-Persistence Results.

In this section, conditions leading to the persistence and nonpersistence of system (4.1) will be considered.

In the case when all selective values are equal, we have the following theorem for persistence.

<u>Theorem 4.10:</u> Let $f_1 = f_2 = f_3$. Assume that both E_3 and E_4 exist and are globally stable on $H_{x_1,y}$ and $H_{x_3,y}$ respectively. If $d_1, d_3 > 0$ and (4.11) holds, then system (4.1) is persistent.

<u>Proof</u>: Let $(x_{10}, x_{20}, x_{30}, y) \in \mathbb{R}_{+}^{4}$ and $w = w(x_{10}, x_{20}, x_{30}, y)$. The existence of E_{3} and E_{4} implies $b_{1}, b_{3} > 0$. Proposition 4.5 shows that $w \cap cl(H_{x_{1}, x_{2}, x_{3}}) = \phi$. Next we show that $E_{3} \neq w$. Since $d_{1} > 0$, dim $W^{U}(E_{1}) = 1$ and $W^{CS}(E_{3}) \cap cl(\mathbb{R}_{+}^{4}) = cl(H_{x_{1}, y})$. By proposition 2.16, $w \cap W^{CS}(E_{3})$ contains a point other than E_{3} and hence w contains a point in $cl(H_{x_{1}, y})$ other than E_{3} . But this is impossible because of the global dynamics of system (4.1) on $cl(H_{x_{1}, y})$ and that $E_{0}, E_{1} \neq w$. Similarly, we can show that $E_{4} \neq w$. Knowing these facts, it is then easy to show that $w \cap b(\mathbb{R}_{+}^{4}) = \phi$.

<u>Remark</u>: Since $B(\overline{x}_i) - D(\overline{x}_i) - \overline{y}_i P_i(\overline{x}_i) = 0$, in the case when $f_1 = f_2 = f_3$, the condition $d_i > 0$ (i = 1,3) simplifies to $P_2(\overline{x}_i) > P_i(\overline{x}_i)$ (i = 1,3). 3× -2

q.e.d.

In the case when the selective values are different, we have the following theorem.

<u>Theorem 4.11</u>. Assume that E_3 (resp. E_4) is globally exponentially <u>stable on</u> $H_{x_1,y}$ (resp. on $H_{x_3,y}$) if it exists. We have the following table for persistence and non-persistence of system (4.1):

	a ₁	a3	bj	^b 3	dl	d ₃	d	·P/NP			
la	-	-	-*	-			-	NP			
15	-	-	-	-			+	Р			
2a	1	-	-	+	-			NP			
2b	-	-	-	+			-	NP			
2c	- ****	-	-	+	+		+	Р			
3a		-	+	-		. I		NP			
3b		-	+	-			-	NP			
3c	-	-	+	-		+	+	Р			
4a	-	-	+	+	, -		•	NP			
4b	-	- ^{r.}	+	+		-		NP			
4c	-	-	+	+		4		NP			
4d	-	-	+	+	+	, +	+	Р			
5	-	•+	-	-				NP			
6a	-	+	-	+		-		NP			
6b	ф -	+	-	+		+		Р			
7	-	+	+	-				NP			
8a	-	+	+	+	-			NP			

80

[8b	-	+	+	+		-	NP
	8c	- •	+ ,	er +	+	+	+	Р
	9	+	-	-	-			NP
	10	+	-	-	+			NP
	11a	+	-	+ *	-	-		NP
	11b	+	-	+	-	+	ņ	Р
	12a	+	· -	+	.+	-		NP
	12b	+	-	+	+		-	NP
	12c	+	-	+	+	+	+ ·	Р
	13	+	+	-	-			NP
<u>ک</u>	14	+	+	-	+			NP
	15	+	+	+	-			NP
	16a	+	+	+	+	-		小NP
	16b	+	+	+	+		-	.NP
	16ç	+	+	+	+	+	+	Р

where a "+" (resp. a "-") means that the corresponding parameter is positive (resp. negative), a blank means that the corresponding parameter value is irrelevant, and, P (resp. NP) stands for persistence (resp. non-persistence) of system (4.1).

<u>Proof</u>: We will illustrate the proof by proving the non-persistence case (1a) and the persistence case (1b). Under case (1), E_3 and E_4 do not exist, E_1 (resp. E_2) is globally stable on $H_{x_1,y}$ (resp. $H_{x_3,y}$), E_5

81

 $\sum_{i=1}^{n}$

is unique and is globally stable on H_{x_1,x_2,x_3} (case (iii) of Theorem 3.9). In case (la), d < 0. Therefore E_5 is asymptotically stable on R_+^4 and hence system (4.1) is non-persistent. In case (lb), it is easily seen that $E_0, E_1, E_2, E_5 \notin w(x_{10}, x_{20}, x_{30}, y_0)$ for all $(x_{10}, x_{20}, x_{30}, y_0) \in R_+^4$. (This type of argument has been used in the proof of Corollary 4.9 and will not be reproduced here.) Therefore the global dynamics on $b(R_+^4)$ implies that $w \cap b(R_+^4) = \phi$ as desired. q.e.d.

4.6 Discussion.

4

In this chapter we have considered a predator-prey model where the prey population consists of three genotypes with fertility / viability differences, and with different predator functional responses. We have given conditions for persistence and non-persistence of the prey genotypes.

The results in Section 4 can be interpreted as: if a prey gamete type has selective disadvantages as well as predation disadvantages, then it will die out eventually. As the results in Section 5 show, the ordering of the selective values and of the predation functional responses are necessary for the results in Section 4 to hold.

As part of the conditions for persistence in Theorem 4.11, we have shown that in some cases where the subsystem modelling the prey population growth exhibits non-persistence, the total system exhibits persistence. This may be viewed as predator regulated survival among prey genotypes, which otherwise, due to fertility/viability differences,

would become extinct.

 \mathcal{O}

In Freedman-Waltman (1978, 1980), conditions were given under the assumption of no fertility / viability differences, for two of the genotypes to become extinct. Here we have generalized these conditions to include the case of fertility differences. We have also given conditions changing the outcome to one of persistence, this time due to fertility / viability differences.

5

ðة'

CHAPTER 5

A TWO-LOCUS, TWO-ALLELE MODEL

In this chapter, the model considered in Chapter 3 for one locus, two allele will be extended to two loci. The fertility/viability of the genotypes will be taken to be equal. It is shown that solutions converge to rest points on the Hardy-Weinberg manifold.

5.1 Introduction and the Model.

Consider a two-locus, two-allele problem where A, a denote the two allele types at the first locus and B,b denote the two allele types at the second locus. There are four gamete types and ten geno-If the number of gametes of type AB; Ab, aB and ab types. are denoted by $u_1^{-}, u_2^{-}, u_3^{-}$ and u_4^{-} respectively, then it is natural to denote the number of organisms of genotype AB/AB, AB/Ab, AB/aB, AB/ab, Ab/Ab, Ab/aB, Ab/ab, aB/aB, aB/ab and ab/ab by x_{11} , x_{12} , x_{13} , x_{14} , x_{22} , x_{23} , x_{24} , x_{33} , x_{34} , x_{44} respectively. In the two-locus, twoallele model proposed here, the genotypes AB/ab and Ab/aB will not be distinguished. x_{11} to denote the number of We use organisms of the genotype AB/AB (another notation: AABB) and use 😔 x_{12} , x_{13} , x_{21} , x_{22} , x_{23} , x_{31} , x_{32} and x_{33} respectively for that of the genotypes AABb , AAbb , AaBB , AaBb $(AB/a\ b\ and\ Ab/aB)$, Aabb , aaBB , aaBb and aabb.

The model we wish to consider in this chapter is of the form:

$$\dot{x}_{11} = \frac{u_1^2}{x^2} \quad B(x) - \frac{x_{11}}{x} \quad D(x)$$

$$\dot{x}_{12} = \frac{2u_1u_2}{x^2} \quad B(x) - \frac{x_{12}}{x} \quad D(x)$$

$$\dot{x}_{13} = \frac{u_2^2}{x^2} \quad B(x) - \frac{x_{13}}{x} \quad D(x)$$

$$\dot{x}_{21} = \frac{2u_1u_3}{x^2} \quad B(x) - \frac{x_{21}}{x} \quad D(x)$$

$$\dot{x}_{22} = \frac{2(u_1u_4 + u_2u_3)}{x^2} \quad B(x) - \frac{x_{22}}{x} \quad D(x)$$

$$\dot{x}_{23} = \frac{2u_2u_4}{x^2} \quad B(x) - \frac{x_{23}}{x} \quad D(x)$$

$$\dot{x}_{31} = \frac{u_3^2}{x^2} \quad B(x) - \frac{x_{31}}{x} \quad D(x)$$

$$\dot{x}_{32} = \frac{2u_3u_4}{x^2} \quad B(x) - \frac{x_{32}}{x} \quad D(x)$$

$$\dot{x}_{33} = \frac{u_4^2}{x^2} \quad B(x) - \frac{x_{33}}{x} \quad D(x)$$

$$u_1 = x_{11} + \frac{1}{2} x_{12} + \frac{1}{2} x_{21} + \frac{1}{4} x_{22}$$

$$u_2 = x_{13} + \frac{1}{2} x_{12} + \frac{1}{2} x_{32} + \frac{1}{4} x_{22}$$

$$u_3 = x_{31} + \frac{1}{2} x_{21} + \frac{1}{2} x_{32} + \frac{1}{4} x_{22}$$
(5.2)

.

· .·

ï,

where

a) M

Ø

85

 $u_4 = x_{33} + \frac{1}{2} x_{23} + \frac{1}{2} x_{32} + \frac{1}{4} x_{22}$

and

 $x = x_{11} + \cdots + x_{33}$

For a derivation of system (5.1), see Appendix 2. The pirth and death functions (B and D) are assumed to satisfy the same hypotheses as those for system (3.1), namely, (H1) - (H6) of Section 3.1. System (5.1) is a no fertility / viability difference generalization of system (3.1) to two loci. It can also be considered as a continuous version of the wellknown model of two loci for non-overlapping generations (see Chapter 8 of Roughgarden (1979)). For the discrete model of two loci (with random mating, no selection, linked or unlinked loci), it is known that the Hardy-Weinberg equilibrium relation is attained gradually over at number of generations (see, for example, Section 2.6 of Crow-Kimura (1970)). An analogous result is true for system (5.1). In contrast to this, the recent work of Akin, Akin (1979, 1982, 1983), shows that in the Kimura model for two loci with selection and recombination (see p. 197 of Crow-Kimura, 1970), it is possible for stable periodic solutions to exist. Other related work of interest are: Nagylaki-Crow (1974), Karlin (1975), Karlin-Carmelli (1975a, 1975b), and Nagylaki (1977).

The rest of this chapter is organized as follows. In Section 5.2, the semi flow on $cl(R_+^9)$ defined by system (5.1) is described and some of its elementary invariance properties are considered. The rest points for system (5.1) are described in Section 5.3. It is shown that the rest point set (excluding the origin), forms a 2-dimensional surface. (with boundary) and has a global parametrization. In Section 5.4, we

86

(5.3)

prove the main result of this chapter: every solution of system (5.1) converges to a rest point as t tends to $+\infty$. It turns out that which rest point a solution converge to is predictable and is dependent only on the initial condition. Finally, Section 5.5 is a short discussion of the results.

5.2 The Associated Flow.

As with system (3.1), system (5.1) is not defined when $X = (x_{11}, \dots, x_{33}) = (0, \dots, 0)$. Let $F = (F_{11}, \dots, F_{33})$ denote the vector field defined by system (5.1) (i.e. the right hand side of (5.1)). Define $F = (0, \dots, 0)$ when $X = (0, \dots, 0)$. Then system (5.1) becomes

$$\dot{X} = F(X)$$
 (5.4)

87

Clearly $F: cl(R_+^9) \rightarrow R^9$ is locally Lipschitz. As in (3.5), we define the simplex

$$U = \{X \in R^9_+ : x = K\}$$

and let

$$c1(U) = \{X \in c1(R_{+}^{9}) : x = K\}$$
 (5.5)

be the closure of U in R^9 . We will summarize some of the properties of system (5.4) in the following proposition.

<u>Proposition 5.1</u>: <u>System (5.4)</u> defines a continuous semi-flow on $c1(R_+^9)$ which becomes a smooth (c^1) local flow when restricted to R_+^9 . The $\frac{\text{closed simplex cl(U) defined in (5.5) is positively invariant and is}}{\text{globally stable on cl(R_+^9) \ {(0,...,0)} . (0,...,0) is a repelling}} \\ \frac{\text{rest point. }}{\text{Hx}_{11}}, \frac{\text{Hx}_{13}}{\text{Hx}_{31}}, \frac{\text{Hx}_{33}}{\text{Hx}_{33}}, \frac{\text{Hx}_{11}, x_{12}, x_{13}}{\text{Hx}_{11}, x_{12}, x_{13}}, \frac{\text{Hx}_{11}, x_{21}, x_{31}}{\text{Hx}_{31}, x_{32}, x_{33}}, \frac{\text{and}}{\text{Hx}_{11}, x_{32}, x_{33}}, \frac{\text{are invariant.}^{\circ}}{\text{The vector field}} F \\ \frac{\text{points into }}{\text{Points into }} R_+^9 \text{ on } b(R_+^9) \ (\text{cl(Hx}_{11}, x_{12}, x_{13}, U, H_{11}, x_{21}, x_{31}, U, H_{11}, x_{32}, x_{33}, U, H_{11}, x_{32}, x_{33}, U, H_{11}, x_{32}, x_{33}, U, H_{11}, x_{31}, x_{31}, x_{32}, x_{33}, U, H_{11}, x_{31}, x_{31}, x_{32}, x_{33}, U, H_{11}, x_{31}, x_{32}, x_{33}, U, H_{11}, x_{31}$

<u>Proof</u>: The positive invariance of $cl(R_+^9)$ can be easily shown using Proposition 2.1. As with system (3,1);

$$\dot{x} = B(x) - D(x)$$
 (5.6)

Hence, for all x(0) > 0,

.

$$x(t) \rightarrow K$$

as $t \rightarrow +\infty$. The positive invariance and global stability properties of cl(U) follow from (H5), (H6) of Section 3.1 and (5.7). The rest of the assertions can be proved as in Proposition 3.1.

(5.7)

q.e.d.

5.3 The Rest Point Set.

In this section the set of rest points for system (5.1) will be described. By Proposition 5.1, all rest points of (5.1) other than $(0, \dots, 0)$ must lie in cl(U). First, we consider the interior rest points.

¢

Proposition 5.2: Define $H : R_+^2 \rightarrow U$ by

$$H(c_{1},c_{2}) = (H_{11}(c_{1},c_{2}),\cdots,H_{33}(c_{1},c_{2})) \quad for \quad (c_{1},c_{2}) \in \mathbb{R}^{2}_{+}$$

where

$$H_{11}(c_{1},c_{2}) = c_{1}^{2}c_{2}^{2} \frac{K}{(1+c_{1})^{2}(1+c_{2})^{2}}$$

$$H_{12}(c_{1},c_{2}) = 2c_{1}^{2}c_{2} \frac{K}{(1+c_{1})^{2}(1+c_{2})^{2}}$$

$$H_{13}(c_{1},c_{2}) = c_{1}^{2} \frac{K}{(1+c_{1})^{2}(1+c_{2})^{2}}$$

$$H_{21}(c_{1},c_{2}) = 2c_{1}c_{2}^{2} \frac{K}{(1+c_{1})^{2}(1+c_{2})^{2}}$$

$$H_{22}(c_{1},c_{2}) = 4c_{1}c_{2} \frac{K}{(1+c_{1})^{2}(1+c_{2})^{2}}$$

$$H_{23}(c_{1},c_{2}) = 2c_{1} \frac{K}{(1+c_{1})^{2}(1+c_{2})^{2}}$$

$$H_{31}(c_{1},c_{2}) = c_{2}^{2} \frac{K}{(1+c_{1})^{2}(1+c_{2})^{2}}$$

89

$$H_{32}(c_{1},c_{2}) = 2c_{2} \frac{K}{(1+c_{1})^{2}(1+c_{2})^{2}}$$
(5.8)

$$H_{33}(c_{1},c_{2}) = \frac{K}{(1+c_{1})^{2}(1+c_{2})^{2}}$$
(5.8)
Then $H(R_{+}^{2})$ is the set of all rest points of system (5.1) in R_{+}^{9} .
Proof: First we show that all points of the form $H(c_{1},c_{2})$ where
 $(c_{1},c_{2}) \in R_{+}^{2}$ are rest points of system (5.1). Let $H(c_{1},c_{2}) = X^{*} = (x_{11}^{*},\cdots,x_{33}^{*})$. Clearly $X^{*} \in R_{+}^{9}$. Define
 $u_{1}^{*} = x_{11}^{*} + \frac{1}{2}x_{12}^{*} + \frac{1}{2}x_{21}^{*} + \frac{1}{4}x_{22}^{*}$
 $u_{2}^{*} = x_{13}^{*} + \frac{1}{2}x_{21}^{*} + \frac{1}{2}x_{32}^{*} + \frac{1}{4}x_{22}^{*}$
 $u_{3}^{*} = x_{31}^{*} + \frac{1}{2}x_{23}^{*} + \frac{1}{2}x_{32}^{*} + \frac{1}{4}x_{22}^{*}$
 $u_{4}^{*} = x_{33}^{*} + \frac{1}{2}x_{23}^{*} + \frac{1}{2}x_{32}^{*} + \frac{1}{4}x_{22}^{*}$
and
 $x^{*} = x_{11}^{*} + \cdots + x_{33}^{*}$.
Then
 $u_{1}^{*} = \frac{c_{1}c_{2}K}{(1+c_{1})(1+c_{2})}$
 $u_{3}^{*} = \frac{c_{2}K}{(1+c_{1})(1+c_{2})}$
 $u_{4}^{*} = \frac{K}{(1+c_{1})(1+c_{2})}$

$$F_{11}(X^{*}) = \frac{u_{1}^{*2}}{x^{*2}} B(x^{*}) - \frac{x_{11}^{*}}{x^{*}} D(x^{*}) = (u_{1}^{*2} - x_{11}^{*}K) \frac{B(K)}{K^{2}} = 0 \quad . \quad \square$$

Similarly, one shows that

$$F_{12}(X^*) = \cdots = F_{33}(X^*) = 0$$

and hence X^* is a rest point.

Next, we show that every interior rest point of system (5.1) is of the form $H(c_1,c_2)$ for some $(c_1,c_2) \in R_+^2$. Let $X^* = (x_{11}, \cdots, x_{33}^*)$ be a rest point of system (5.1) with $x_{ij}^* > 0$ (i,j = 1,2,3). Define $u_1^*, u_2^*, u_3^*, u_4^*$, and x^* as in (5.9). Clearly $x^* = K$. Also define

$$d_{ij} = \frac{x_{ij}^{*}}{x_{33}^{*}} \qquad (i,j = 1,2,3) \qquad (5.10)$$

$$r_{i} = \frac{u_{i}^{*}}{x_{33}^{*}} \qquad (i = 1,\cdots,4) \qquad (5.11)$$

Then

and

$$r_1^2 x_{33}^* = d_{11} K$$
 (i)

$$2r_1r_2 x_{33}^{*} = d_{12} K$$
 (ii)

$$r_2^2 x_{33}^* = d_{13}^K$$
 (iii)

$$2r_1r_3 x_{33}^* = d_{21}K$$
 (iv)

 $(2r_1r_4 + 2r_2r_3) x_{33}^* = d_{22}K$ (v)

$$2r_{2}r_{4} x_{33}^{*} = d_{23}K \qquad (vi)$$

$$r_{3}^{2} x_{33}^{*} = d_{31}K \qquad (vii)$$

$$2r_{3}r_{4} x_{33}^{*} = d_{32}K \qquad (viii)$$

$$r_{4}^{2} x_{33}^{*} = d_{32}K \qquad (viii)$$

$$r_{4}^{2} x_{33}^{*} = d_{32}K \qquad (viii)$$

$$r_{4}^{2} x_{33}^{*} = d_{33}K = K \qquad (ix)$$
Define $c_{1}, c_{2} > 0$ by
$$c_{1} = \frac{x_{23}}{2x_{33}^{*}} \quad \text{and} \quad c_{2} = \frac{x_{32}}{2x_{33}^{*}} \quad .$$
Then $2c_{1} = d_{23}$ and $2c_{2} = d_{32}$.
$$\frac{(ii)}{(i)} : \frac{2r_{2}}{r_{1}} = \frac{d_{12}}{d_{11}} \quad \text{and} \quad \frac{(ii)}{(iii)} : \frac{2r_{1}}{r_{2}} = \frac{d_{12}}{d_{13}} \quad \text{give} \quad \frac{(ii)}{(i)} : \frac{(ii)}{(iii)} : \frac{d_{12}}{d_{33}} = d_{32}$$

$$\frac{d_{12}}{d_{12}} = 4d_{11}d_{13} \quad . \quad \frac{(viii)}{(vii)} : \frac{2r_{4}}{r_{3}} = \frac{d_{32}}{d_{31}} \quad \text{and} \quad \frac{(viii)}{(ix)} : \frac{2r_{3}}{r_{4}} = \frac{d_{32}}{d_{33}} = d_{32}$$

$$give \quad \frac{(viii)}{(vii)} \cdot \frac{(viii)}{(ix)} : d_{31} = c_{2}^{2} \quad . \quad (iv)$$

$$\frac{2r_{1}}{r_{3}} = \frac{d_{21}}{d_{31}} \quad give \quad \frac{(iv)}{(i)} : \frac{(iv)}{(vii)} : d_{21} = 2c_{2}\sqrt{d_{11}} \quad . \quad \frac{(vi)}{(iii)} : \frac{2r_{4}}{r_{2}} = \frac{d}{d_{33}}$$

$$and \quad \frac{(vi)}{(ix)} : \frac{2r_{2}}{r_{4}} = \frac{d_{23}}{d_{33}} = d_{23} \quad give \quad \frac{(vi)}{(ii)} : \frac{(vi)}{(ix)} : d_{13} = c_{1}^{2} \quad Thus,$$

$$d_{12} = 2c_{1}\sqrt{d_{11}} \quad .$$

92

^d32

d_23 \overline{d}_{13}

Since

$$\frac{r_2}{r_1} = \frac{c_1}{\sqrt{d_{11}}}$$
, $\frac{r_3}{r_1} = \frac{c_2}{\sqrt{d_{11}}}$, $\frac{r_4}{r_1} = \frac{1}{\sqrt{d_{11}}}$, and,

WT.

ŝ).

$$\frac{(v)}{(1)}: \frac{2r_4}{r_1} + \frac{2r_2r_3}{r_1^2} = \frac{d_{22}}{d_{11}}, \text{ therefore}'$$

Substituting all of the above into the equation $x^* = K$, we have,

$$(d_{11} + 2c_1\sqrt{d_{11}} + c^2 + 2c_2\sqrt{d_{11}} + 2c_1c_2 + 2\sqrt{d_{11}} + 2c + c_2^2 + 2c_2 + 1)x_{33}^* = K$$

which in turn implies that

$$x_{33}^{\star} = \frac{1}{(\sqrt{d_{11}} + c_1 + c_2 + 1)^2}$$

Substituting this into (i) and by noting $r_1 = d_{11} + \frac{1}{2} d_{12} + \frac{1}{2} d_{21} + \frac{1}{4} d_{22}$, we have, $d_{11} = c_1^2 c_2^2$. From this, one can easily show that $X^* = H(c_1, c_2)$.

Next, we observe that the boundary rest points of system (5.1) must lie on $cl(H_{x_{11},x_{12},x_{13}})$, $cl(H_{x_{11},x_{21},x_{31}})$, $cl(H_{x_{31},x_{32},x_{33}})$ or $cl(H_{x_{13},x_{23},x_{33}})$. From this we have the following proposition.

<u>Proposition 5.3</u>: $cl(H(R_{+}^{2}))$ is the set of rest points of system (5.1). in $cl(R_{+}^{9}) \setminus \{(0, \dots, 0)\}$.

<u>Proof</u>: This follows from Proposition 5.2, the above observation and case (i) of Proposition 3.3. Note that, for example, $cl(H(R_{+}^{2})) \cap$ $cl(H_{x_{11},x_{21},x_{31}})$ is the Hardy-Weinberg parabolic segment given in

93

q.e.d.

Proposition 3.3.

<u>Remark</u>: As in Section 3.3, $c1(H(R_+^2)) = H([0,+\infty][0,+\infty])$ is called the <u>Hardy-Weinberg</u> manifold. In this case it is a 2-dimensional manifold (surface) with boundary.

5.4 Global Convergence to the Hardy-Weinberg Manifold.

In this section we will show that every solution X(t) of system (5.1) with X(0) ϵ cl(R_+^9) \ {(0,...,0)} converges to a rest point in cl(H(R_+^2)) as t $\rightarrow +\infty$.

First we define some auxiliary quantities. Define

$$\begin{cases} x_{AA} = x_{11} + x_{12} + x_{13} \\ x_{Aa} = x_{21} + x_{22} + x_{23} \\ x_{aa} = x_{31} + x_{32} + x_{33} \\ x_{aa} = x_{31} + x_{32} +$$

$$x_{BB} = x_{11} + x_{21} + x_{31}$$

$$x_{Bb} = x_{12} + x_{22} + x_{32}$$

$$x_{bb} = x_{13} + x_{23} + x_{33}$$
(5.13)

$$u_A = x_{AA} + \frac{1}{2} x_{Aa}$$
 $u_a = x_{aa} + \frac{1}{2} x_{Aa}$, (5.14)

q.e.d.

(5.12)

6.2

Ì.

$$u_{B} = x_{BB} + \frac{1}{2} x_{Bb} \qquad u_{b} = x_{bb} + \frac{1}{2} x_{Bb} \qquad (5.15)$$
Clearly
$$u_{A} + u_{a} = x_{AA} + x_{Aa} + x_{aa} = x \qquad (5.16)$$

$$u_{B} + u_{b} = x_{BB} + x_{Bb} + x_{bb} = x \qquad (5.16)$$

$$u_{B} + u_{b} = x_{BB} + x_{Bb} + x_{bb} = x \qquad (5.16)$$

$$u_{A} = u_{1} + u_{2} \qquad u_{a} = u_{3} + u_{4} \qquad (5.17)$$

$$u_{B} = u_{1} + u_{3} \qquad u_{b} = u_{2} + u_{4} \qquad (5.17)$$

$$u_{B} = u_{1} + u_{3} \qquad u_{b} = u_{2} + u_{4} \qquad (5.17)$$

$$(1) \quad u_{A}(t) = c_{1}u_{a}(t) \qquad (u_{B}(t)) = c_{2}u_{b}(t) \qquad (5.18)$$

$$(1) \quad (u_{A}(t), u_{a}(t)) + \left(\frac{c_{1}K}{1+c_{1}}, \frac{K}{1+c_{1}}\right) \qquad (5.19)$$

 $\mathbf{C} \geq$

and

1

(

$$\underbrace{\operatorname{and}}_{(u_{B}(t), u_{b}(t))} \xrightarrow{(1+c_{2})} \left(\underbrace{\stackrel{k}{1+c_{2}}}_{1+c_{2}}, \frac{k}{1+c_{2}} \right)$$
(5.20)

$$as t \rightarrow +\infty$$
, and,
 $c_{-}^{2}K = 2c_{+}K$

iii)
$$(x_{AA}(t), x_{Aa}(t), x_{aa}(t)) \rightarrow \left(\frac{c_1^K}{(1+c_1)^2}, \frac{2c_1^K}{(1+c_1)^2}, \frac{K}{(1+c_1)^2}\right)$$
 (5.21)
 $(x_{BB}(t), x_{Bb}(t), x_{bb}(t)) \rightarrow \left(\frac{c_2^2 K}{(1+c_2)^2}, \frac{2c_2^K}{(1+c_2)^2}, \frac{K}{(1+c_2)^2}\right)$ (5.22)

as +∞. t→

£-

X

$$\dot{x}_{AA} = \frac{u_A^2}{x^2} B(x) - \frac{x_{AA}}{x} D(x)$$

$$\dot{x}_{Aa} = \frac{2u_A u_a}{x^2} B(x) - \frac{x_{Aa}}{x} D(x)$$
(5.23)
$$\dot{x}_{Aa} = \frac{u_a^2}{x^2} B(x) - \frac{x_{aa}}{x} D(x)$$

$$\dot{x}_{BB} = \frac{u_B^2}{x^2} B(x) - \frac{x_{BB}}{x} D(x)$$

$$\dot{x}_{Bb} = \frac{2u_B u_b}{x^2} B(x) - \frac{x_{Bb}}{x} D(x)$$
(5.24)
$$\dot{x}_{bb} = \frac{u_b^2}{x^2} B(x) - \frac{x_{bb}}{x} D(x)$$

Ð

A

96

÷.

and

R.S

Using a result in Freedman-Waltman (1978) (mentioned in the proof of case (i) of Theorem 3.9), we know that (i) holds and that

$$\begin{aligned} x_{AA}(t) &- c_1^2 x_{aa}(t) \neq 0 , & x_{Aa}(t) - 2c_1 x_{aa}(t) \neq 0 \\ & & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ &$$

ί.

as $t \rightarrow +\infty$. (ii) and (iii) follow from (5.7), (5.16) and (5.25).

- n

Remark: Proposition 5.4 and (5.17) show that

$$u_{1}(t) = c_{2}u_{b}(t) - u_{a}(t) + u_{4}(t) = c_{1}u_{a}(t) - u_{b}(t) + u_{4}(t)$$

$$u_{2}(t) = u_{b}(t) - u_{4}(t)$$

$$u_{3}(t) = u_{a}(t) - u_{4}(t)$$
(5.26)

for all $t \ge 0$, where

$$c_1 = \frac{u_A(0)}{u_a(0)}$$
 and $c_2 = \frac{u_B(0)}{u_b(0)}$

Proposition 5.5. Let $c_1 = \frac{u_A(0)}{u_A(0)} \frac{\text{and}}{\text{and}} c_2 = \frac{u_B(0)}{u_B(0)} \cdot \frac{\text{Then}}{\text{Then}}$

$$(u_1(t), u_2(t), u_3(t), u_4(t)) \rightarrow$$

 $\left(\frac{c_{1}c_{2}K}{(1+c_{1})(1+c_{2})}, \frac{c_{1}K}{(1+c_{1})(1+c_{2})}, \frac{c_{2}K}{(1+c_{1})(1+c_{2})}, \frac{K}{(1+c_{1})(1+c_{2})}\right) (5.27)$

as $t \rightarrow +\infty$.

Proof: The proof is divided into three steps.

Step 1: u₁, u₂, u₃, u₄ satisfy the following system of differential equations.

$$\dot{u}_{1} = (u_{1}^{2} + u_{1}u_{2} + u_{1}u_{3} + \frac{1}{2}u_{1}u_{4} + \frac{1}{2}u_{2}u_{3}) \frac{B(x)}{x^{2}} - u_{1} \frac{D(x)}{x}$$
$$\dot{u}_{2} = (u_{2}^{2} + u_{1}u_{2} + u_{2}u_{4} + \frac{1}{2}u_{1}u_{4} + \frac{1}{2}u_{2}u_{3}) \frac{B(x)}{x^{2}} - u_{2} \frac{D(x)}{x}$$

97[°]

$$\dot{u}_{3} = (u_{3}^{2} + u_{1}u_{3} + u_{3}u_{4} + \frac{1}{2}u_{1}u_{4} + \frac{1}{2}u_{2}u_{3}) \frac{B(x)}{x^{2}} - u_{3}\frac{D(x)}{x}$$
$$\dot{u}_{4} = (u_{4}^{2} + u_{2}u_{4} + u_{3}u_{4} + \frac{1}{2}u_{1}u_{4} + \frac{1}{2}u_{2}u_{3}) \frac{B(x)}{x^{2}} - u_{4}\frac{D(x)}{x} , \quad (5.28)$$

where $\dot{x} = u_1 + u_2 + u_3 + u_4$. This follows directly from (5.1), (5.2), and, (5.3).

Step 2:

$$u_{1}(t)-c_{2}u_{2}(t) = (u_{10}-c_{1}u_{20}) \exp\left[\int_{0}^{t} \left(\frac{(1+c_{1})u_{a}(\tau)B(x(\tau))}{2x^{2}(\tau)} - \frac{D(x(\tau))}{x(\tau)}\right)d\tau\right]$$

$$u_{3}(t)-c_{2}u_{4}(t) = (u_{30}-c_{2}u_{40}) \exp\left[\int_{0}^{t} \left(\frac{(1+c_{1})u_{a}(\tau)B(x(\tau))}{2x^{2}(\tau)} - \frac{D(x(\tau))}{x(\tau)}\right)d\tau\right]$$

$$u_{1}(t)-c_{1}u_{3}(t) = (u_{10}-c_{1}u_{30}) \exp\left[\int_{0}^{t} \left(\frac{(1+c_{1})u_{a}(\tau)B(x(\tau))}{2x^{2}(\tau)} - \frac{D(x(\tau))}{x(\tau)}\right)d\tau\right]$$

$$u_{2}(t)-c_{1}u_{4}(t) = (u_{20}-c_{1}u_{40}) \exp\left[\int_{0}^{t} \left(\frac{(1+c_{1})u_{a}(\tau)B(x(\tau))}{2x^{2}(\tau)} - \frac{D(x(\tau))}{x(\tau)}\right)d\tau\right]$$

$$(5.29)$$

This can be proved by writing down the linear differential equation that each of these functions satisfies. For example, by (5.26) and (5.28),

$$\dot{u}_{2} - c_{1}\dot{u}_{4} = (u_{2} - c_{1}u_{4}) \left[\frac{(1 + c_{2})u_{b}B(x)}{2x^{2}} - \frac{D(x)}{x} \right]$$
$$= (u_{2} - c_{1}u_{4}) \left[\frac{(1 + c_{1})u_{a}B(x)}{2x^{2}} - \frac{D(x)}{x} \right]$$
Step 3:

$$u_{1}(t) - c_{2}u_{2}(t) \rightarrow 0 \qquad u_{3}(t) - c_{2}u_{4}(t) \rightarrow 0$$

$$u_{1}(t) - c_{1}u_{3}(t) \rightarrow 0 \qquad u_{2}(t) - c_{1}u_{4}(t) \rightarrow 0 , \qquad (5.30)$$

as $t \to +\infty$. By (5.7), (5.19), and, (5.20),

$$\frac{(1+c_1)u_a(t)B(x(t))}{2x^2(t)} - \frac{D(x(t))}{x(t)} \to -\frac{B(K)}{2K} < 0$$

мſ

5

J

as $t \rightarrow +\infty$. Therefore, the integral in (5.29) is divergent and consequently we have (5.30).

Using (5.30), it is then easy to see that (5.27) holds because

$$u_1 + u_2 + u_3 + u_4 = x$$
 (5.31)

q.e.d.

6.

ή

99

Theorem 5.6: Let
$$c_1 = \frac{u_A(0)}{u_a(0)}$$
 and $c_2 = \frac{u_B(0)}{u_b(0)}$. Then
($x_{11}(t), \dots, x_{33}(t)$) + H(c_1, c_2) (5.32)

as $t \rightarrow +\infty$, where $H(c_1, c_2)$ is defined in (5.8).

Proof: To show (5.32), it suffices to show

$$x_{11}(t) - c_{1}^{2}c_{2}^{2}x_{33}(t) , x_{12}(t) - 2c_{1}^{2}c_{2}x_{33}(t) , x_{13}(t) - c_{1}^{2}x_{33}(t) ,$$

$$x_{21}(t) - 2c_{1}c_{2}^{2}x_{33}(t) , x_{22}(t) - 4c_{1}c_{2}x_{33}(t) , x_{23}(t) - 2c_{1}x_{33}(t) ,$$

$$x_{31}(t) - c_{2}^{2}x_{33}(t) , x_{32}(t) - 2c_{2}x_{33}(t) \neq 0$$
(5.33)

as $t \rightarrow +\infty$. We illustrate this by showing $z(t) \equiv x_{32}(t) - 2c_{12}^{2}x_{33}(t) \rightarrow 0$ as $t \rightarrow +\infty$. Clearly z(t) satisfies the linear differential equation

$$\dot{z}(t) + Q_1(t)z(t) = Q_2(t)$$

where $Q_1(t) = \frac{D(x(t))}{x(t)}$, $Q_2(t) = 2u_4(t)[u_3(t) - c_2u_4(t)] \frac{B(x(t))}{x(t)}$. Solving this equation yields

$$z(t) = z(0) \exp \left[-\int_0^{t} Q_1(\tau) d\tau\right] + \int_0^{t} \exp \left[-\int_s^{t} Q_1(\tau) d\tau\right] Q_2(s) ds$$

We will show that the first and second terms tend to 0 as t tends to $+\infty$.

Since
$$\lim_{t \to +\infty} Q_1(t) = \frac{D(K)}{K}$$
, by (3.7), therefore

$$\int_{0}^{+\infty} Q_{1}(t)dt = +\infty , \text{ and hence, } \lim_{t \to +\infty} \exp\left[-\int_{0}^{t} Q_{1}(\tau)d\tau\right] = 0$$

On the other hand, by (5.29),

$$exp[-\int_{s}^{t} Q_{1}(\tau)d\tau] Q_{2}(s)ds = 2(u_{3}(0)-c_{2}u_{4}(0)) exp[-\int_{0}^{t} \frac{D(x(\tau))}{x(\tau)} d\tau]$$

•
$$\int_{0}^{t} u_{4}(s) \exp\left[\frac{1+c_{2}}{2}\int_{0}^{s} \frac{u_{4}(\tau)B(x(\tau))}{x^{2}(\tau)} d\tau\right] \frac{B(x(\tau))}{x^{2}(\tau)} ds$$
 (5.34)

To show that the L.H.S. of (3.34) tends to 0 as t tends to $+\infty$ it suffices, by L'Hospital's rule, to show that

$$\lim_{t \to +\infty} \frac{u_{4}(t) \exp\left[\frac{1+c_{2}}{2} \int_{0}^{t} \frac{u_{4}(\tau)B(x(\tau))}{x(\tau)} d\tau\right] \frac{B(x(t))}{x^{2}(t)} = 0 \quad (5.35)$$

Since

$$\begin{array}{c}
\left(\begin{array}{c} \lim_{t \to +\infty} u_4(t) = \frac{K}{(1+c_1)(1+c_2)} > 0, \\
\left(\begin{array}{c} \lim_{t \to +\infty} \frac{B(x(t))}{x(t)} = \frac{B(K)}{K} > 0, \text{ and, } \lim_{t \to +\infty} \frac{D(x(t))}{x(t)} = \frac{D(K)}{K} > 0, \\
\end{array}\right)$$

therefore, it suffices to show,

$$\cdot \lim_{t \to +\infty} \int_0^t \left[\frac{D(x(\tau)}{x(\tau)} - \frac{(1+c_2)u_4(\tau)B(x(\tau))}{x^2(\tau)} \right] d\tau = +\infty.$$

But this follows from

$$\lim_{t \to +\infty} \left[\frac{D(x(t))}{x(t)} - \frac{(1+c_2)u_4(t)B(x(t))}{x^2(t)} \right] = \frac{D(K)}{K} - \frac{1+c_2}{2} \cdot \frac{K}{1+c_2} \cdot \frac{B(K)}{K^2} = \frac{B(K)}{2K} > 0 .$$
q.e.d.*

<u>Remark</u>: Theorem 5.6 implies that $cl(R_{+}^{9}) \setminus \{(0, \dots, 0)\}$ is foliated by the strong stable manifolds $W^{SS}(H(c_1, c_2))$, $((c_1, c_2) \in [0, +\infty] \times [0, +\infty])$ of the rest points $H(c_1, c_2)$ on the Hardy-Weinberg manifold, $cl(H(R_{+}^{2}))$. For $0 < c_1, c_2 < +\infty$,

$$W^{SS}(H(c_1, c_2)) \cap cl(R_+^9) = \{(x_{11}, \dots, x_{33}) \in cl(R_+^9) \setminus \{(0, \dots, 0)\}$$

$$(x_{11}^{+}x_{12}^{+}x_{13}^{+}) + \frac{1-c_1}{2} (x_{21}^{+}x_{22}^{+}x_{23}^{+}) - c(x_{31}^{+}x_{32}^{+}x_{33}^{+}) = 0$$

$$(x_{11}^{+}x_{21}^{+}x_{31}^{+}) + \frac{1-c_2}{2} (x_{12}^{+}x_{22}^{+}x_{32}^{+}) - c(x_{13}^{+}x_{23}^{+}x_{33}^{+}) = 0$$

is 7-dimensional. For $c_1 = 0$, $0 < c_2 < +\infty$,

$$W^{SS}(H(0,c_2)) \cap cl(R_+^9) = \{(x_{11}, \cdots, x_{33}) \in cl(H_{x_{31}}, x_{32}, x_{33}) \setminus \{(0, \cdots, 0)\}:$$

$$x_{31} + \frac{1-c_2}{2} x_{32} - c_2 x_{33} = 0\}$$

is 2-dimensional. For $0 < c_1 < +\infty$, $c_2 = 0$,

$$W^{SS}(H(c_1,0)) \cap cl(R_{+}^{9}) = \{(x_{11}, \dots, x_{33}) \in cl(H_{x_{13}}, x_{23}, x_{33}) \setminus \{(0, \dots, 0)\}:$$
$$x_{13} + \frac{1-c_1}{2} x_{23} - c_1 x_{33} = 0\}$$

is 2-dimensional. For $c_1 = +\infty$, $0 < c_2 < +\infty$,

$$W^{SS}(H(_{0}^{+\infty},c_{2}^{-})) \cap c^{1}(R_{+}^{9}) = \{(x_{11},\cdots,x_{33}^{-}) \in c^{1}(H_{x_{11}},x_{12}^{-},x_{13}^{-}) \setminus \{(0,\cdots,0)\}: x_{11}^{-1} + \frac{1-c_{2}}{2} x_{12}^{-1} - c_{2}^{-}x_{13}^{-1} = 0\}$$

is 2-dimensional. For $0 < c_1 < +\infty$, $c_2 = +\infty$,

Ą

$$W^{SS}(H(c_1, +\infty)) \cap c1(R_+^9) = \{(x_{11}, \dots, x_{33}) \in c1(H_{x_{11}}, x_{21}, x_{31}) \setminus \{(0, \dots, 0)\}:$$
$$x_{11} + \frac{1 - c_1}{2} x_{21} - c_1 x_{31} = 0\}$$

ç

is 2-dimensional. Finally,

$$W^{SS}(H(0,0)) \cap cl(R_{+}^{9}) = H_{X_{33}}$$

103

÷

1

$$W^{SS}(H(+\infty,0)) \cap c1(R_{+}^{9}) = H_{X_{13}}$$

are 1-dimensional.

5.5 Discussion.

In this chapter we have shown that if $(x_{11}(t), \dots, x_{33}(t))$ is a solution of system (5.1) other than $(0, \dots, 0)$, then

$$x_{11}(t):x_{12}(t):x_{13}(t):x_{21}(t):x_{22}(t):x_{23}(t):x_{31}(t):x_{32}(t):x_{33}(t)$$

+ $c_1^2c_2^2:2c_1^2c_2:c_1^2:2c_1c_2^2:4c_1c_2:2c_1:c_2^2:2c_2:1$

at $t \rightarrow +\infty$, where

$$c_{1} = \frac{x_{11}(0) + x_{12}(0) + x_{13}(0) + \frac{1}{2}x_{21}(0) + \frac{1}{2}x_{22}(0) + \frac{1}{2}x_{23}(0)}{x_{31}(0) + x_{31}(0) + x_{33}(0) + \frac{1}{2}x_{21}(0) + \frac{1}{2}x_{22}(0) + \frac{1}{2}x_{23}(0)}$$

and

$$c_{2} = \frac{x_{11}(0) + x_{21}(0) + x_{31}(0) + \frac{1}{2}x_{12}(0) + \frac{1}{2}x_{22}(0) + \frac{1}{2}x_{32}(0)}{x_{13}(0) + x_{23}(0) + x_{33}(0) + \frac{1}{2}x_{12}(0) + \frac{1}{2}x_{22}(0) + \frac{1}{2}x_{32}(0)}$$

This is directly analogous to the discrete case as discussed in Section 2.6 of Crow-Kimura (1970). It should be mentioned that model (5.1) is a first attempt to extend the one-locus, two-allele model (3.1) to two loci. And as such, many of the standard two loci features (for example, selection, recombination, etc.) are not incorporated. In the next chapter, the effect of predator mediated selection will be studied using a predator-prey model in which the prey population is modelled by system (5.1). The selection comes from choosing different predator functional responses for the various prey genotypes.

CHAPTER 6

A PREDATOR-PREY MODEL CONSISTING OF NINE PREY GENOTYPES

In this chapter a model of predator-prey interaction, for which the prey population consists of nine genotypes corresponding to a two-locus, two-allele problem and modelled by system (5.1), is proposed. Some sufficiency conditions leading to the evolution of pure strains as well as to the persistence of all components of the system are given. These conditions serve to illustrate the similarity as well as differences between the one-locus and the two-locus models.

6.1 Introduction and the Model.

In this chapter, the model of the previous chapter is extended to include a predator. More specifically, the model to be studied is given by:

$$\dot{x}_{11} = \frac{u_1^2}{x^2} B(x) - \frac{x_{11}}{x} [D(x) + yP_{11}(x)]$$

$$\dot{x}_{12} = \frac{2u_1u_2}{x^2} B(x) - \frac{x_{12}}{x} [D(x) + yP_{12}(x)]$$

$$\dot{x}_{13} = \frac{u_2^2}{x^2} B(x) - \frac{x_{13}}{x} [D(x) + yP_{13}(x)]$$

$$\dot{x}_{21} = \frac{2u_1u_3}{x^2} B(x) - \frac{x_{21}}{x} [D(x) + yP_{21}(x)]$$

$$\dot{x}_{22} = \frac{2(u_1u_4 + u_2u_3)}{x_2} B(x) - \frac{x_{22}}{x} [D(x) + yP_{22}(x)]$$
(6.1)

$$\dot{x}_{23} = \frac{2u_2u_4}{x^2} B(x) - \frac{x_{23}}{x} [D(x) + yP_{23}(x)]$$

$$\dot{x}_{31} = \frac{u_3^2}{x^2} B(x) - \frac{x_{31}}{x} [D(x) + yP_{31}^{x}(x)]$$

$$\dot{x}_{32} = \frac{2u_3u_4}{x^2} B(x) - \frac{x_{32}}{x} [D(x) + yP_{32}(x)]$$
$$\dot{x}_{33} = \frac{u_4^2}{x^2} B(x) - \frac{x_{33}}{x} [D(x) + yP_{33}(x)]$$

$$\dot{y} = y[-s + k \sum_{i,j=1}^{3} \frac{x_{ij}}{x} P_{ij}(x)]$$

where u_1 , u_2 , u_3 , u_4 are given by (5.2), x is given by (5.3), and, s,k > 0. As in the case of systems (4.1) and (5.1), x_{ij} (i,j=1,2,3) are the nine genotypes of the prey population and y denotes the predator population. The birth and death functions (B and D) are assumed to satisfy the assumptions (H1) - (H6) Section 3.1. The predator functional responses P_{ij} (i,j=1,2,3) are assumed to satisfy (H7) - (H9) of Section 4.1, namely,

(H7) $P_{ij} : cl(R_{+}^{l}) \rightarrow cl(R_{+}^{l})$ (i,j=1,2,3) are C^{l} (H8) $P_{ij}(0) = 0$ (i,j=1,2,3) (H9) $P'_{ij}(x) > 0$ (i,j=1,2,3) for $x \in cl(R_{+}^{l})$.

In Section 6.2, we construct theiflow defined by (6.1) and study its boundedness and invariance properties. Some of the boundary rest points of system (6.1) and their stability properties are considered in Section 6.3. In Section 6.4, some sufficiency conditions leading to the evolution of pure strains as well as the global stability of a boundary rest point will be given. In Section 6.5, some sufficiency conditions which guarantee the persistence of the system will be discussed. We finish with a short discussion of the results in Section 6.6.

6.2 The Associated Flow.

Q

As in the case of system (4.1), system (6.1) is not defined when x = 0. When x = 0, we define system (6.1) to be

> $\dot{x}_{ij} = 0$ (6.2) $\dot{y} = -sy$

and write (6.1) and (6.2) as

$$\dot{x}_{ij} = F_{ij}(x_{11}, \dots, x_{33}, y)$$

$$\dot{y} = G(x_{11}, \dots, x_{33}, y) .$$
(6.3)

We have the following analogue of Proposition 4.1:

Proposition 6.1: F_{ij} (i,j=1,2,3), $G \in Lip \ on \ cl(R_+^{10}) \ and$ F_{ij} (i,j=1,2,3), $G \in C^1 \ on \ R_+^{10}$. Thus system (6.3) defines a continuous semi-flow on $cl(R_+^{10})$ which becomes a smooth (C^1) local flow when restricted on R_{+}^{10} . It is A-dissipative with A = { $(x_{11}, \dots, x_{33}, y) \in c1(R_{+}^{10})$: $x \le K$ and $kx + y \leq \frac{k}{s} (M + sK)$ ' 🔨 (6.4) <u>where</u> $M = \max \{B(x) - D(x) : x \in [0,K]\}$. $H_{x_{11}}, H_{x_{13}}, H_{x_{31}}, H_{x_{33}}, H_{y}$, H_{x11}, y, H_{x13}, y, H_{x31}, y, H_{x33}, y, H_{x11}, x₁₂, x₁₃, H_{x11}, x₂₁, x₃₁, H_{x13}, x₂₃, x₃₃, H_{x31}, x₃₂, x₃₃, H_{x11}, x₁₂, x₁₃, y, H_{x11}, x₂₁, x₃₁, y, H_{x13}, x₂₃, x₃₃, y, $H_{x_{31},x_{32},x_{33},y}$, and $H_{x_{11}}, \dots, x_{33}$ are invariant. The vector field $(f_{11}, \dots, F_{33}, G)$ points into R_{+}^{10} on $b(R_{+}^{10}) \setminus V$, where $V = c1(H_{x_{11},x_{12},x_{13},y}, U, H_{x_{11},x_{21},x_{31},y}, U, H_{x_{13},x_{23},x_{33},y})$ $\begin{bmatrix} U & H \\ x_{31}, x_{32}, x_{33}, y \end{bmatrix} \begin{bmatrix} U & H \\ x_{11}, \cdots, x_{33} \end{bmatrix}$. (6.5)

<u>Proof</u>: To show that $F_{11} \in \text{Lip on } c1(R_+^{10})$, it suffices to consider two points in $c1(R_+^{10})$ of the form $(x_{11}, \dots, x_{33}, y)$ and $(0, \dots, 0, \overline{y})$ where $x = x_{11} + \dots + x_{33} > 0$. Then

$$\begin{aligned} \left| F_{11}(x_{11}, \cdots, x_{33}, y) - F_{11}(0, \cdots, 0, \overline{y}) \right| &\leq \frac{u_1^2}{x^2} B(x) + \frac{x_{11}}{x} [D(x) + yP_{11}(x)] \\ &\leq B(x) + D(x) + yP_{11}(x) \\ &\leq [B'(z_1) + D'(z_2) + yP'_{11}(z_3)] |x| \end{aligned}$$

for some $0 \le z_1, z_2, z_3 \le x$. Therefore $F_{11} \in Lip$ on $cl(R_+^{10})$ and similarly we show that $F_{12}, \dots, F_{33}, G \in Lip$.

The positive invariance of $cl(R_{+}^{10})$ can be easily verified using Proposition 2.1. Hence system (6.3) defines a continuous local semi-flow on $cl(R_{+}^{10})$ which becomes a C^{1} local flow when restricted to R_{+}^{10} .

To show that system (6.3) is A-dissipative for A defined in (6.4), it suffices to observe that

 $\dot{x} \leq B(x) - D(x)$ $\dot{x} + \dot{y} \leq k[B(x) - D(x)] - sy$

and use Proposition 2.2.⁷ This also shows that system (6.1) defines a continuous semi-flow on $cl(R_+^{10})$.

The invariance of the sets listed is clear. It is also easy to show that the vector field $(F_{11}, \dots, F_{33}, G)$ points into R^{10} on $b(R_{+}^{10}) \setminus V$. For example, for $(X, y) = (x_{11}, \dots, x_{33}, y) \in b(R_{+}^{10})$ with $x_{11} = 0$, $F_{11}(X, y) = \frac{u_1^2}{x^2} B(x) \ge 0$ and $F_{11}(X, y) = 0$ if and only if $u_1 = 0$ (i.e. $x_{11} = x_{12} = x_{21} = x_{22}$).

5

6

6.3 Some Boundary Rest Points and Their Stability.

4

In this section some of the rest points of system (6.1) which lie on $b(R_{+}^{10})$ will be considered. According to Proposition 6.1, these rest points must lie on V. Clearly, $E_0(0, \dots, 0)$ is a rest point and by Proposition 5.3, $E = \{(H(c_1, c_2), 0) : (c_1, c_2) \in [0, +\infty] \times [0, +\infty]\}$ is the set of rest points of system (6.1) in $cl(H_{x_{11}}, \dots, x_{33}) \setminus \{E_0\}$. Let

 $b_{ij} = -s + k P_{ij}(K)$ (i,j=1,3) . (6.6)

Then $H_{x_{ij},y_{a}}(i,j=1,3)$ contains a rest point if and only if $b_{ij} > 0$ (i,j=1,3). If they exist these rest points are unique. Let us denote these rest points by E_i (i=1,2,3,4) if they exist. That is, let $E_1 = (\bar{x}_{11},0,\cdots,0,\bar{y}_{11})$, $E_2 = (0,0,\bar{x}_{13},0,\cdots,0,\bar{y}_{13})$, $E_3 = (0,\cdots,0,\bar{x}_{31},0,0,\bar{y}_{31})$, and $E_4 = (0,\cdots,0,\bar{x}_{33},\bar{y}_{33})$. The invariant sets $H_{x_{11},x_{21},x_{31},y}, H_{x_{11},x_{12},x_{13},y}, H_{x_{13},x_{23},x_{33},y}$, and $H_{x_{31},x_{32},x_{33},y}$ may or may not contain rest points. Some conditions are known which guarantee the existence or non-existence of these rest points (see Freedman-So-Waltman (1984)). In this thesis we will not use these conditions.

The rest point E_0 is unstable. Indeed we have the following Proposition.

<u>Proposition 6.2</u>: $E_0 \neq w(x_{11}, \dots, x_{33}, y)$ for all $(x_{11}, \dots, x_{33}, y) \in R_+^{10}$.

<u>Proof</u>: This follows from Proposition 2.15 because in a small enough neighbourhood of E_0 , we have,

$$\dot{\mathbf{x}} \ge \mathbf{B}(\mathbf{x}) - \mathbf{D}(\mathbf{x}) - \mathbf{yP}(\mathbf{x}) \ge \mathbf{k}_1 \mathbf{x}$$
 for some $\mathbf{k}_1 > 0$

$$\mathbf{\dot{y}} \leq \mathbf{y}[-\mathbf{s} + \mathbf{k}\mathbf{P}(\mathbf{x})] \leq -\mathbf{k}_2 \mathbf{y}$$
 for some $\mathbf{k}_2 > 0$

where $P(x) = \max \{P_{ij}(x) : i, j = 1, 2, 3\}$, and the flow on $cl(H_y)$ is given by $\dot{y} = -s\dot{y}$.

Corollary 6.3:
$$(0, \dots, 0, y) \notin w(x_{11}, \dots, x_{33}, y)$$
 for all $(x_{11}, \dots, x_{33}, y) \in R_{+}^{10}$.

<u>Proof</u>: This follows from Proposition 6.2 and the global stability of E_0 on the invariant set $cl(H_y)$.

q.e.d.

The stability properties of each $(H(c_1,c_2),0)$ in $c1(X_{x_{11}},\cdots,x_{33}) \setminus \{E_0\}$ is determined by Theorem 5.6. To compute the stability of $(H(c_1,c_2),0)$ in the y-direction we note that

$$d(c_{1},c_{2}) = -s + \frac{k}{(1+c_{1})^{2}(1+c_{2})^{2}} [c_{1}^{2}c_{2}^{2}P_{11}(K) + 2c_{1}^{2}c_{2}P_{12}(K) + c_{1}^{2}P_{13}(K) + 2c_{1}c_{2}^{2}P_{21}(K) + 4c_{1}c_{2}P_{22}(K) + 2c_{1}P_{23}(K) + c_{2}^{2}P_{31}(K) + c_{2}P_{32}(K) + P_{33}(K)]$$
(6.6)

is the eigenvalue of the variational matrix of system (6.1) at $(H(c_1,c_2),0)$ that corresponds to eigenvectors that have a non-zero y

component. A condition for the persistence of y is given by the following proposition.

$$\frac{\text{Proposition 6.4: If } d(c_1,c_2) > 0 \quad \text{for all } c_1,c_2 \in [0,+\infty] \text{, then}}{\lim_{t \to +\infty} y(t) > 0 \quad \text{for all } (x_{110},\cdots,x_{330},y_0) \in \mathbb{R}^{10}_+ \text{.}}$$

<u>Proof</u>: The proof is similar to that of Proposition 4.4. Again we know that E is repelling in the y-direction and therefore in order for $\lim_{t \to +\infty} y(t) = 0$, the omega-limit set $w(x_{110}, \dots, x_{330}, y_0)$ of $(x_{110}, \dots, x_{330}, y_0)$ must intersect the stable manifold, $W^{S}(E)$, of the invariant set E at a point other than E itself. Now $W^{S}(E) \cap c1(R_{+}^{10})$ $\subset c1(H_{x_{11}}, \dots, x_{33})$. The global dynamics on $c1(H_{x_{11}}, \dots, x_{33})$ as given by Theorem 5.6 implies that either w contains E_0 or w is unbounded. In either case, we have a contradiction.

q.e.d.

<u>Remark</u>. Unlike as in Lemma 4.5, we are unable to obtain a necessary and sufficient condition on $P_{ij}(K)$ so that $d(c_1,c_2) > 0$ for all $c_1, c_2 \in [0, +\infty]$. However, from the definition of $d(c_1,c_2)$ given in (4.7), it is clear that $d(c_1,c_2) \leq 0$ for all $c_1, c_2 \in [0, +\infty]$ if $P_{ij}(K) \geq \frac{s}{k}$ for all i, j = 1, 2, 3.

The linear stability of E_i (i = 1,2,3,4) is governed by the signs of the real parts of the eigenvalues of the variational matrix $M(E_i)$ (i = 1,2,3,4) of system (6.1) evaluated at E_i (i = 1,2,3,4). For illustration, we will present $M(E_1)$ below. To simplify notations, we use

j.

 $M_{ij,hk}$ to denote $\frac{\partial F_{ij}}{\partial x_{hk}}(E_1)$, $M_{y,ij}$ to denote $\frac{\partial G}{\partial x_{ij}}(E_1)$, etc. and we move y to the second row.

	× ₁₁	у	¹ × ₁₂	×13	×21	× ₂₂	× ₂₃	×31	× ₃₂	×33
× ₁₁	?	-ve	?	?	?	?	?	?	?	?
У	+ve	0	?	?	?	?	?	?	?	?
×12	0	0	?	+ve	0	+ve	+ve	0	0,.	0
×13	0	0	0	~ve	0	0	0	0	0	0
×21	0	0	0	0	?	+ve	0	+ve	+ve	0
× ₂₂	0	0	0	0.	0	?	+ve	0	+ve	+ve
x ₂₃	0	0	0	0	0	0	-ve	0	0	0
× ₃₁	0	0	0	0	0	0	0	-ve		0
× ₃₂	0	۰ ۵	0	0	0	0	0	0	-ve	0
×33	0	0	0	0	0	0	0.	0	0	-ve

where +ve (resp. -ve,0,?) means that the entry is positive (resp. is negative, zero, has indeterminate sign). Furthermore, the non-zero entries are given by:

$$M_{11,11} = B'(\overline{x}_{11}) - D'(\overline{x}_{11}) - \overline{y}_{11}P_{11}'(\overline{x}_{11})$$

$$M_{11,y} = -P_{11}(\overline{x}_{11}) < 0$$

$$M_{11,12} = M_{11,21} = B'(\overline{x}_{11}) - D'(\overline{x}_{11}) - \overline{y}_{11}P_{11}'(\overline{x}_{11})$$

$$M_{11,22} = -\frac{1}{2\overline{x}_{11}}B(\overline{x}_{11}) + B'(\overline{x}_{11}) - D'(\overline{x}_{11}) - \overline{y}_{11}P_{11}'(\overline{x}_{11})$$

$$\begin{split} \mathsf{M}_{11}, \mathbf{ij} &= -\frac{1}{\overline{x_{11}}} \; \mathsf{B}(\overline{x_{11}}) + \mathsf{B}'(\overline{x_{11}}) - \mathsf{D}'(\overline{x_{11}}) - \overline{y_1}/\mathsf{P}_{11}'(\overline{x_{11}}) \\ (\text{for } (\mathbf{i}, \mathbf{j}) &= (\mathbf{1}, \mathbf{3}), (2, \mathbf{3}), (3, 1), (3, 2) \text{ or } (3, 3)) \\ \mathsf{M}_{y,11} &= \mathbf{k} \overline{y_{11}} \; \mathsf{P}_{11}'(\overline{x_{11}}) > 0 \\ \mathsf{M}_{y,1j} &= \frac{\mathbf{k} \overline{y_{11}}}{\overline{x_{11}}} \; \mathsf{L} \overline{x_{11}} \mathsf{P}_{11}'(\overline{x_{11}}) + \mathsf{P}_{1j}(\overline{x_{11}}) - \mathsf{P}_{11}(\overline{x_{11}})] \\ \mathsf{M}_{12,12} &\equiv \frac{1}{\overline{x_{11}}} \; \mathsf{L} \mathsf{B}(\overline{x_{11}}) - \mathsf{D}(\overline{x_{11}}) - \overline{y_{11}} \mathsf{P}_{12}(\overline{x_{11}})] = \frac{\overline{y_{11}}}{\overline{x_{11}}} \; \mathsf{L} \frac{\mathsf{S}}{\mathsf{K}} - \mathsf{P}_{12}(\overline{x_{11}})] \\ \mathsf{M}_{12,12} &\equiv \frac{1}{\overline{x_{11}}} \; \mathsf{B}(\overline{x_{11}}) > 0 \\ \mathsf{M}_{12,22} &= \frac{1}{\overline{x_{11}}} \; \mathsf{B}(\overline{x_{11}}) > 0 \\ \mathsf{M}_{12,22} &= \frac{1}{\overline{x_{11}}} \; \mathsf{B}(\overline{x_{11}}) > 0 \\ \mathsf{M}_{13,13} &= \frac{1}{\overline{x_{11}}} \; \mathsf{L} (\mathsf{D}(\overline{x_{11}}) + \overline{y_{11}} \mathsf{P}_{13}(\overline{x_{11}}))] < 0 \\ \mathsf{M}_{21,21} &= \frac{1}{\overline{x_{11}}} \; \mathsf{L} (\mathsf{B}(\overline{x_{11}}) - \mathsf{D}(\overline{x_{11}}) - \overline{y_{11}} \mathsf{P}_{21}(\overline{x_{11}})] = \frac{\overline{y_{11}}}{\overline{x_{11}}} \; \mathsf{L} \frac{\mathsf{S}}{\mathsf{K}} - \mathsf{P}_{21}(\overline{x_{11}})] \\ \mathsf{M}_{21,22} &= \frac{1}{\overline{x_{11}}} \; \mathsf{B}(\overline{x_{11}}) > 0 \\ \mathsf{M}_{21,31} &= \frac{1}{\overline{x_{11}}} \; \mathsf{B}(\overline{x_{11}}) > 0 \\ \mathsf{M}_{21,32} &= \frac{1}{\overline{x_{11}}} \; \mathsf{B}(\overline{x_{11}}) > 0 \\ \mathsf{M}_{21,32} &= \frac{1}{\overline{x_{11}}} \; \mathsf{B}(\overline{x_{11}}) > 0 \\ \mathsf{M}_{21,32} &= \frac{1}{\overline{x_{11}}} \; \mathsf{B}(\overline{x_{11}}) > 0 \end{split}$$

$$M_{22,22} = \frac{1}{\overline{x_{11}}} \left[\frac{1}{2} B(\overline{x_{11}}) - D(\overline{x_{11}}) - \overline{y_{11}} P_{22}(\overline{x_{11}}) \right]$$

$$= \frac{1}{\overline{x_{11}}} \left[\overline{y_{11}} \left(\frac{s}{k} - P_{22}(\overline{x_{11}}) \right) - \frac{1}{2} B(\overline{x_{11}}) \right]$$

$$M_{22,23} = \frac{1}{\overline{x_{11}}} B(\overline{x_{11}}) > 0$$

$$M_{22,32} = \frac{1}{\overline{x_{11}}} B(\overline{x_{11}}) > 0$$

$$M_{22,33} = \frac{2}{\overline{x_{11}}} B(\overline{x_{11}}) > 0$$

$$M_{23,23} = -\frac{1}{\overline{x_{11}}} \left[D(\overline{x_{11}}) + \overline{y_{11}} P_{23}(\overline{x_{11}}) \right] < 0$$

$$M_{31,31} = -\frac{1}{\overline{x_{11}}} \left[D(\overline{x_{11}}) + \overline{y_{11}} P_{31}(\overline{x_{11}}) \right] < 0$$

$$M_{32,32} = -\frac{1}{\overline{x_{11}}} \left[D(\overline{x_{11}}) + \overline{y_{11}} P_{32}(\overline{x_{11}}) \right] < 0$$

$$M_{33,33} = -\frac{1}{\overline{x_{11}}} \left[D(\overline{x_{11}}) + \overline{y_{11}} P_{33}(\overline{x_{11}}) \right] < 0$$

\$r:

()

Therefore the $M(E_1)$ has five negative eigenvalues: $M_{13,13}$, $M_{23,23}$, $M_{31,31}$, $M_{32,32}$, $M_{33,33}$, three real eigenvalues with indeterminate signs: $M_{12,12}$, $M_{21,21}$, $M_{22,22}$, and, the sign of the real parts of the remaining two eigenvalues (corresponding to eigenvectors lying on $H_{x_{11},y}$) is the same as that of the sign of $M_{11,11}$.

115

]

6.4 Some Conditions Which Lead to Evolution of Pure Strains.

In this section, conditions fleading to the global stability of boundary rest points will be considered. Due to the complexity of the problem, we are forced to focus on a small number of cases. In our analysis, we will need to know the global dynamics of system (6.1) on the boundary invariant set V. The assumptions are essentially those in Theorem 4.6 applied to each of the four 4-dimensional boundary invariant sets $H_{x_{11},x_{12},x_{13},y}$, $H_{x_{11},x_{21},x_{31},y}$, $H_{x_{13},x_{23},x_{33}}^{and}$, and $H_{x_{31},x_{32},x_{33}}^{and}$. The added assumptions for this section and the next are as follows.

- (HT) The predator y persists. Mathematically speaking, this means that, according to Proposition 6.4, we require $d(c_1,c_2) > 0$ for all $c_1, c_2 \in [0,+\infty]$.
- (H2) E_1 (resp. E_2, E_3, E_4) is globally exponentially stable on $H_{x_{11}, y}$ (resp. $H_{x_{13}, y}$, $H_{x_{31}, y}$, $H_{x_{33}, y}$),
- (H3) The four sets of predator functional responses: $P_{11}/P_{12}/P_{13}$ $P_{11}/P_{21}/P_{31}$, $P_{13}/P_{23}/P_{33}$, and, $P_{31}/P_{32}/P_{33}$ are ordered with the heterozygote one between the two homozygotes ones. There are 16 possibilities. By interchanging the roles of the allele types A (resp. B) and a (resp. b), if necessary, we are left with 3 cases to consider:
 - (1) $P_{11} \leq P_{12} \leq P_{13}$, $P_{11} \leq P_{21} \leq P_{31}$, $P_{13} \leq P_{23} \leq P_{33}$, $P_{31} \leq P_{32} \leq P_{33}$.

(2)
$$P_{11} \leq P_{12} \leq P_{13}$$
, $P_{11} \leq P_{21} \leq P_{31}$, $P_{33} \leq P_{23} \leq P_{13}$,
 $P_{33} \leq P_{32} \leq P_{31}$, and
(3) $P_{11} \leq P_{12} \leq P_{13}$, $P_{11} \leq P_{21} \leq P_{31}$, $P_{13} \leq P_{23} \leq P_{33}$,
 $P_{33} \leq P_{32} \leq P_{31}$.

The case when $P_{11} \leq P_{12} \leq P_{13} \leq P_{23} \leq P_{33} \leq P_{23} \leq P_{31} \leq P_{21} \leq P_{11}$ is excluded because we require at least one of the inequalities to be strict.

In each of the above three cases (1) - (3), we require the differences of the appropriate P_{ij} 's be bounded below by a non-negative linear function of x on [0,K]. For example $P_{11} \leq P_{12} \leq P_{13}$ in case (1) should read: there exist $k_{11,12}$, $k_{12,13} \geq 0$, not both zero, such that $P_{12}(x) - P_{11}(x) \geq k_{11,12}x$ and $P_{13}(x) - P_{12}(x) \geq k_{12,13}x$ for $x \in [0,K]$.

The following is a theorem which provides conditions under which the rest point E_1 is globally stable.

Theorem 6.5: Suppose (HT) and (HZ) hold. Furthermore assume that

- (i) $P_{11} \leq P_{12} \leq P_{13} \leq P_{21} \leq P_{22} \leq P_{23} \leq P_{31} \leq P_{32} \leq P_{33}$,
- (ii) there exist $e_{12}, e_{13}, e_{21}, e_{22}, e_{23}, e_{31}, e_{32}, e_{33} \neq 0$ such that

$$\begin{split} & \mathsf{P}_{12}(x) - \mathsf{P}_{11}(x) \ge \mathsf{e}_{12}x \quad , \quad \mathsf{P}_{13}(x) - \mathsf{P}_{12}(x) \ge \mathsf{e}_{13}x \quad , \\ & \mathsf{P}_{21}(x) - \mathsf{P}_{13}(x) \ge \mathsf{e}_{21}x \quad , \quad \mathsf{P}_{22}(x) - \mathsf{P}_{21}(x) \ge \mathsf{e}_{22}x \quad , \\ & \mathsf{P}_{23}(x) - \mathsf{P}_{22}(x) \ge \mathsf{e}_{23}x \quad , \quad \mathsf{P}_{31}(x) - \mathsf{P}_{23}(x) \ge \mathsf{e}_{31}x \quad , \end{split}$$

$$P_{32}(x) - P_{31}(x) \ge e_{32}x , P_{33}(x) - P_{32}(x) \ge e_{33}x$$

$$for all x \in [0,K], and,$$

$$(iii) \overline{y}_{13}[\frac{s}{k} - P_{22}(\overline{x}_{13})] \neq \frac{1}{2} B(\overline{x}_{13}) ,$$

$$\overline{y}_{31}[\frac{s}{k} - P_{22}(\overline{x}_{31})] \neq \frac{1}{2} B(\overline{x}_{31}) , and$$

$$\overline{y}_{33}[\frac{s}{k} - P_{22}(\overline{x}_{33})] \neq \frac{1}{2} B(\overline{x}_{33}) ,$$

$$then E_{1} is globally stable on R_{+}^{10}.$$

<u>Proof</u>: (HZ), (ii), and (iii) imply that the rest points E_i (i = 1,2,3,4) are hyperbolic. In fact by studying the eigenvalues of the variational matrix at each of these rest points, as given in Section 6.3, we know, dim $W^{u}(E_1) = 0$, dim $W^{u}(E_2) \ge 1$, dim $W^{u}(E_3) \ge 1$, dim $W^{u}(E_4) \ge 2$. Therefore E_1 is asymptotically stable and $W^{U}(E_1)$ (i = 2,3,4) must have non-empty intersection with $R^{10} \setminus c1(R_+^{10})$. By Proposition 2.11, $W^{s}(E_{i}) \cap cl(R_{i}^{10})$ must lie in $b(R_{i}^{10})$. Theorem 4.6 tells us that $W^{s}(E_{4}) \cap cl(R_{+}^{10}) \in cl(H_{X_{33},y})$, $W^{s}(E_{3}) \cap cl(R_{+}^{10}) \in cl(H_{X_{31},X_{32},X_{33},y})$, and $W^{s}(E_{2}) \cap cl(R_{+}^{10}) \subset cl(H_{x_{13},x_{23},x_{33},y}^{\circ})$. From this one can easily show that E_1 is the only point in $b(R_1^{10})$ that can lie in $w = w(x_{11}, \dots, x_{33}, y)$ for any $(x_{11}, \dots, x_{33}, y)$ in R_{+}^{10} . Moreover, if $E_1 \in w$, then the orbit converges to E_1 as $t \rightarrow +\infty$. Using REDUCE 2 (a programming language for algebraic manipulations), one shows that $\frac{u_a}{u_a} - \frac{u_A}{u_A} \le -\frac{y}{u_A u_a} [e_{33}x_{12}x_{33} + other non-negative terms].$ Therefore, as in the proof of Theorem 4.6,

σ

$$u_{a}(t) \leq \frac{u_{a}(0)}{u_{A}(0)} u_{A}(t) \exp \left[-\frac{e_{33}}{\kappa^{2}} \int_{0}^{t} (x_{12}x_{33}y)(\tau)d\tau\right]$$

119

If the integral diverges, then $\lim_{t \to +\infty} u_a(t) = 0$ so that $w \cap b(R_+^{10}) \neq \phi$ and hence the orbit converges to E_1 . Now suppose $z(t) = (x_{12}x_{33}y)(t) \in L^1[0,+\infty)$, then $\lim_{t \to +\infty} z(t) = 0$ and hence $\lim_{t \to +\infty} x_{12}(t) = 0$ or $\lim_{t \to +\infty} x_{33}(t) = 0$ (since $\lim_{t \to +\infty} y(t) > 0$ by $(H\overline{1})$). This again implies $w \cap b(R_+^{10}) \neq \phi$ and so the orbit converges to E_1 as before. This shows that E_1 is globally attracting and it completes the proof of the theorem. q.e.d.

<u>Remark</u>: The assumption (iii) in the above theorem is technical and is necessary in order for the rest points E_i (i = 1,2,3,4) to be hyperbolic. This assumption is "generically" satisfied.

6.5 <u>Some Persistence Results</u>.

In this section we will consider conditions which lead to the persistence of system (6.1). Besides the conditions $(HT) - (H\overline{3})$ made in the previous section, we will also assume that the rest points E_i (i = 1,2,3,4) are hyperbolic. This amounts to requiring the $k_{ij,hl}$'s in (H\overline{3}) be positive and that $\overline{y}_{ij}[\frac{s}{k} - P_{22}(\overline{x}_{ij})] - \frac{1}{2}B(\overline{x}_{ij})$ (i,j=1,3) be non-zero.

Theorem 6.6: Under the above assumptions, system (2.1) is persistent if

(i) $\overline{y}_{11}\left[\frac{s}{k} - P_{22}(\overline{x}_{11})\right] - \frac{1}{2}B(\overline{x}_{11}) > 0$ holds under cases (1) and

(ii)
$$\overline{y}_{11}[\frac{s}{k} - P_{22}(\overline{x}_{11})] - \frac{1}{2}B(\overline{x}_{11}) > 0$$
, and
 $\overline{y}_{33}[\frac{s}{k} - P_{22}(\overline{x}_{33})] - \frac{1}{2}B(\overline{x}_{33}) > 0$

hold under case (2) of $(H\overline{3})$ in Section 6.4.

Proof: We will only discuss case (1) of (H3) in Section 6.4. The other two cases are similar. Let $(x_{11}, \dots, x_{33}, y) \in R_+^{10}$ and let $w = w(x_{11}, \dots, x_{33}, y)$. Our aim is to show $w \cap b(R_+^{10}) = \phi$, or equivalently $w \cap V = \phi$, where V was given in (6.5). Recall that $\lim_{t\to +\infty} y(t) > 0$ by (H1). First we will show that $E_4 \notin w$. Suppose not, that is, suppose $E_4 \in W$. Since dim $W^u(E_4) \ge 2$, therefore $W^u(E_4) \cap$ $(R_+^{10} \setminus cl(R_+^{10})) \neq \phi$, and hence, by Proposition 2.11, $W^{s}(\dot{E}_4) \cap R_+^{10} = \phi$. It follows from the global dynamics on $cl(H_{x_{13},x_{23},x_{33},y})$ and $c1(H_{x_{31},x_{32},x_{33},y})$ as given in Theorem 6.1 that $W^{S}(E_{4}) \cap c1(R_{+}^{10}) \subset C^{1}(R_{+}^{10})$ $cl(H_{X_{33}},y)$. The Butler-McGehee lemma then implies that w contains a point in $cl(H_{x_{33},y})$ other than E_4 which is a contradiction. Next we show that $E_3 \notin w$. As in the case of E_4 , $W^{u}(E_{3}) \geq 1$ implies $W^{u}(E_{3}) \cap (R_{+}^{10} \setminus cl(R_{+}^{10})) \neq \phi$ and therefore, by Proposition 2.11, $W^{S}(E_{3}) \cap R_{+}^{10} = \phi$. The global dynamics on $cl(H_{x_{11},x_{21},x_{31},y})$ and $cl(H_{x_{31},x_{32},x_{33},y})$ imply that $W^{s}(E_{3}) \cap cl(R_{+}^{10})$ $(c)(H_{x_{31},x_{32},x_{33},y})$. The Butler-McGehee lemma says that w must contain a point in $c1(H_{x_{31},x_{32},x_{33},y})$ other than E_3 , which is a contradiction.

Hence $E_3 \notin w$. Similarly $E_2 \notin w$. One could repeat the above argument for E_1 because $M_{22}(E_1) > 0$ by assumption. Hence $E_i \notin w$ (i = 1,2,3,4) and $w \cap V = \phi$ follows from the global dynamics on V.

6.6 Discussion.

Due to the complexity of (6.1), our analysis for this system is not as complete as the other models (3.1), (4.1) and (5.1), that we studied in the previous chapters. However, Theorem 6.6 points to a way of making the system persistent which otherwise would not be by adjusting the predator functional response of the double heterozygote (AaBb). This result indicates a situation for which a two-locus problem cannot be analyzed as the "sum" effect of two one-locus problems.

CHAPTER 7

CONCLUDING DISCUSSION & REMARKS

'In this thesis we study four continuous models in population genetics/ecology both for the one-locus, two-allele problem and for the two-locus, two-allele problem.

In Chapter 3, a continuous model (system (3.1)) for a onelocus, two-allele problem with non-linear birth (B) and death (D) processes as well as fertility/viability differences is considered. The population x under consideration is divided into three subpopulations x_1 , x_2 , x_3 corresponding to the three different genotypes AA, Aa and aa with fertility/viabilities denoted by f_1 , f_2 and f_3 . The main result for this model is Theorem 3.9 which provides a complete picture of the global dynamics of system (3.1). This theorem can be summarized as follows.

(i) When $f_1 = f_2 = f_3$ (no selection), all solutions converge to some equilibrium. More precisely, if $(x_1(t), x_2(t), x_3(t))$ is a solution, then it converges to the equilibrium: $\frac{K}{(1+c)^2}(c^2, 2c, 1)$ where c is determined by initial conditions, namely $c = \frac{x_1(0) + \frac{1}{2}x_2(0)}{x_2(0) + \frac{1}{2}x_2(0)}$. Hence,

all solutions converge to a polymorphism (positive interior equilibrium). Which equilibrium it converges to depends on initial conditions. Phrased in another way, this result also says that the proportion $x_1(x): x_2(t): x_3(t)$ tends to $c^2: 2c: 1$ as t tends to positive infinity. This is the analogue of the Hardy-Weinberg equilibrium relation for discrete one-

locus, two-allele models with random mating and no selection.

¢

٢

(ii) When $f_1, f_3 > f_2$ (homozygous advantage), the positive $x_1 - x_2 \cdot x_3$ cone is divided into two regions by a separatrix (dividing) surface. Solutions initiating from one region will converge to the boundary equilibrium $(x_1, x_2, x_3) = (K, 0, 0)$ whereas solutions initiating from the other region will converge to the boundary equilibrium (0, 0, K). Solutions initiating on the separatrix surface converge to the fique unstable polymorphism: $\frac{K}{(1+c)^2}(c^2, 2c, 1)$. Unlike the case when there is no selection, c is independent of initial conditions in this case. Instead, it is determined by the fertilities f_1 , f_2 and f_3 , namely,

 $c = \frac{f_3 - f_2}{f_1 - f_2}$. Hence, under "most" initial conditions, one of the gamete types becomes extinct. Which one will actually become extinct depends on initial conditions.

(iii) When $f_2 > f_1$, f_3 (heterozygous advantage), all (positive) solutions converge to the unique stable polymorphism: $\frac{K}{(1+c)^2}(c^2,2c,1)$. Hence all three genotypes coexist. As in the case of homozygous advantage, c is independent of initial conditions and is dependent solely on fertility differences, namely, $c = \frac{f_2 - f_3}{f_2 - f_1}$.

(iv) When $f_1 > f_2 > f_3$ (selection for allele A which has incomplete dominance over allele a) or when $f_1 > f_2 = f_3$ (selection for the recessive allele A), or when $f_1 = f_2 > f_3$ (selection for the dominant allele A), all solutions converge to the boundary equilibrium (K,0,0). Hence, the gamete type a becomes extinct.

}

(v) When $f_3 > f_2 > f_1$ (selection for allele a which has incomplete dominance over allele A), or when $f_3 > f_2 = f_1$ (selection for the recessive allele a), or when $f_3 = f_2 > f_1$ (selection for the dominant allele a), all solutions converge to the boundary equilibrium (0,0,K). Hence, the gamete type A becomes extinct.

124

All of the above results have their analogue with discrete models.

In Chapter 4, the model in Chapter 3 is extended to include a predator y (system (4.1)). The prey population x is modelled by (3.1). For the prey population, there are two types of selection forces at work. First, there is the one studied in Chapter 3, namely, selection due to fertility/viability differences. The other type is selection due to differential predator functional responses. (Recall that the predator functional response of x_i (i = 1,2,3) was denoted by P_i (i = 1,2,3) .) Of particular interest are the following two questions.

(1) Under what conditions will one of the prey gamete types become extinct?

(2) When will all the prey genotypes coexist with the predator?

The concept of coexistence is formulated in terms of the definition of persistence: if $x_i(0) > 0$ (i=1,2,3) and y(0) > 0, then lim inf $x_i(t) > 0$ (i=1,2,3) and lim inf y(t) > 0. In the case of $t \rightarrow +\infty$ no selection due to fertility / viability differences, the answer to question (1) is provided by Theorem 4.6. This theorem (modulo a number of technicalities) can be summarized as follows.

Ċ

 \bigcirc

If the predator functional responses of the prey genotypes are ordered: $P_1 < P_2 < P_3$ (hence x_1 can be thought of as the one which is the most difficult to catch) and the predator y can survive on x_1 alone (hence it will also be able to survive on the other two prey genotypes alone), then gamete type a becomes extinct.

This result can be interpretated as saying that the predator exhausted the two prey genotypes x_2 and x_3 which are easier to catch and lives on the prey genotype x_1 alone. In the case when there is selection due to fertility / viability differences, the answer to question (1) is provided by Theorem 4.8. In summary this theorem states the following. (

If the fertilities are ordered: $f_1 \ge f_2 \ge f_3$ and not all requal (hence x_1 has fertility advantage) and the predator functional responses are also ordered: $P_1 \le P_2 \le P_3$ (hence x_1 has predation advantage), then gamete a becomes extinct.

Note that in this theorem, it is not required that the predator y must survive. In fact it can also go extinct along with x_2 and x_3 . This result is intuitively clear in view of the previous results for no selection due to differential predator functional responses and for no selection due to fertility / viability differences.

The answer to question (2) in the case of no selection due to fertility/viability differences is provided by Theorem 4.10. In summary, this theorem states the following.

d.

We assume that

(i) the subsystem with only x_1 and y has a globally stable positive interior equilibrium $E_3 = (\overline{x}_1, 0, 0, \overline{y}_1)$ (consequently $P_1(K) > \frac{s}{k}$) and the subsystem with only x_3 and y has a globally stable positive interior equilibrium $E_4 = (0, 0, \overline{x}_3, \overline{y}_3)$ (consequently $P_3(K) > \frac{s}{k}$),

(ii)
$$P_2(\bar{x}_1) < \frac{s}{k}$$
, $P_2(\bar{x}_3) < \frac{s}{k}$, and,

(iii)
$$P_2(K) > \frac{s}{k} - \sqrt{(P_1(K) - \frac{s}{k})(P_3(K) - \frac{s}{k})}$$

then system (4.1) is persistent.

Recall that s is the death rate of the predator y in the of prey and k is the conversion factor from prey biomass to edator biomass. Condition (iii) is a condition to guarantee y surives. It says that $P_2(K)$ should be large enough so that y will not go extinct. Condition (ii) is a condition to guarantee neither E_3 ynor E_4 are asymptotically stable. It says that $P_2(\overline{x_1})$ (resp. $P_2(\overline{x_3})$) should be small enough so that for solutions that start near E_3 (resp. E_4), the gamete type a (resp. A) will not become extinct. Note that conditions (ii) and (iii) are not mutually exclusive because $\overline{x_1} < K$ (1), 3) and P_2 is strictly increasing.

In the case when the fertilities / viabilities are all different, the answer to question (2) is provided by Theorem 4.11. In order to e tain the results of this theorem, let us recall the following assumptions and notations. It is assumed that if the subsystem consisting of x_1 and y only contains a positive interior equilibrium $E_3 = (\overline{x}_1, 0, 0, \overline{y}_1)$, then it is globally stable. Similarly, if the subsystem consisting of x_3 and y only contains a positive interior equilibrium $E_4 = (0, 0, \overline{x}_3, \overline{y}_3)$, then it is globally stable.

 $a_{i} = f_{i} - f_{2} \qquad (i = 1,3)$ $b_{i} = -s + kP_{i}(K) \qquad (i = 1,3)$ $d_{i} = \frac{1}{\overline{x}_{i}} \left[\frac{f_{2}B(\overline{x}_{i})}{f_{i}} - D(\overline{x}_{i}) - \overline{y}_{i}P_{2}(\overline{x}_{i}) \right] \qquad (i = 1,3)$

$$d = -s + \frac{k}{(1+c)^2} [c^2 P_1(K) + 2c P_2(K) + P_3(K)]$$

where $c = \frac{a_3}{a_1}$. Note that d_1 (resp. d_3) is defined only when E_3 (resp. E_4) exists and d is defined only when a_1 and a_3 are of the same sign. Moreover, $b_i > 0$ (i = 1,3) is the necessary and sufficient condition for the existence of E_i (i = 1,3). By noting $B(\overline{x}_i) - D(\overline{x}_i) - \overline{y}_i P_i(\overline{x}_i) = 0$, the meaning of d_i will become a little bit more trans-

parent if we rewrite it as $\frac{1}{\overline{x}_i} [\overline{y}_i (\frac{s}{k} - P_2(\overline{x}_i)) - \frac{a_i}{f_i} B(\overline{x}_i)]$.

Since there are thirty two persistence and non-persistence cases studied, we will only highlight the results of the theorem by discussing two of these cases here. (See Theorem 4.11 for the labelling of the cases.)

127-

(I) Persistence due to predation.

In case (8c): $a_1 < 0$, $a_3 > 0$, $b_1 > 0$, $b_3 > 0$, $d_1 > 0$, $d_3 > 0$, system (4.1) is persistent. Since $a_1 < 0$ and $a_3 > 0$, we know that gamete A (consequently x_1 and x_2) will go extinct in the absence of the predator y. b_1 , $b_3 > 0$ says that y can survive on either x_1 or x_3 alone. $d_1 > 0$ (i = 1,3) reduces to

 $P_2(\overline{x}_i) < \frac{s}{k} - \frac{a_i}{f_i \overline{y}_i} B(\overline{x}_i)$. Hence the persistence of the system is due to P_i (i = 1,2,3).

(II) Persistence due to fertility / viability differences.

In case (1b): $a_1 < 0$, $a_3 < 0$, $b_1 < 0$, $b_3 < 0$, d > 0, system (4.1) is persistent. In the absence of fertility/viability differences (that is, assume $a_1 = a_3 = 0$), the system is non-persistent, because $b_1 < 0$ (i = 1,3). However, if the heterozygote x_2 has selection advantage due to fertilities/viabilities and the predator functional response P_2 of x_2 is large enough to make d > 0, then the system is persistent.

In Chapter 5 the model studied in Chapter 3 is extended to a two-locus, two-allele model. Only the no selection case is considered. The whole population is again denoted by x and the subpopulations of the nine genotypes: AABB, AABb, AAbb, AaBB, AaBb, AaBb, aaBb, aaBb, aaBb, and, aabb are denoted by x_{11} , x_{12} , x_{13} , x_{21} , x_{22} , x_{23} , x_{31} , x_{32} , and, x_{33} respectively. The main result of this chapter is Theorem 5.6 which provides a complete picture of the global dynamics of system (5.1). It is an analogue of the Hardy-Weinberg equilibrium relation for a

discrete two-locus, two-allele with random mating and no selection model. In summary, this theorem states the following.

All solutions of system (5.1) converge to some equilibrium. More precisely, if $(x_{11}(t), x_{12}(t), x_{13}(t), x_{21}(t), x_{22}(t), x_{23}(t), x_{31}(t), x_{32}(t), x_{33}(t))$ is a solution of system (5.1), then it converges to the equilibrium: $\frac{K}{(1+c_1)^2(1+c_2)^2} (c_1^2c_2^2, 2c_1^2c_2, c_1^2, 2c_1c_2^2, 4c_1c_2, 2c_1, c_2^2, 2c_2, 1)$

as t tends to positive infinity. As in the one-locus case, c₁,c₂ depend on initial conditions, namely

$$c_{1} = \frac{[x_{11}(0) + x_{12}(0) + x_{13}(0)] + \frac{1}{2} [x_{21}(0) + x_{22}(0) + x_{23}(0)]}{[x_{31}(0) + x_{32}(0) + x_{33}(0)] + \frac{1}{2} [x_{21}(0) + x_{22}(0) + x_{23}(0)]}$$

and

3

2

$$c_{2} = \frac{[x_{11}(0) + x_{21}(0) + x_{31}(0)] + \frac{1}{2} [x_{12}(0) + x_{22}(0) + x_{32}(0)]}{[x_{13}(0) + x_{23}(0) + x_{33}(0)] + \frac{1}{2} [x_{12}(0) + x_{22}(0) + x_{32}(0)]}$$

Hence, the dynamics of system (5.1) is trivial. Rephrased in terms of proportions, this theorem also says that $x_{11}(x) : x_{12}(t) : x_{13}(t) : x_{21}(t) : x_{22}(t) : x_{23}(t) : x_{31}(t) : x_{32}(t) : x_{33}(t)$ tends to $c_1^2 c_2^2 : 2c_1^2 c_2 : c_1^2 : 2c_1 c_2^2 : 4c_1 c_2 : 2c_1 : c_2^2 : 2c_2 : 1$ as t tends to positive infinity, which is the more usual form in population genetics literature.

In Chapter 6, the model in Chapter 5 ds extended to include a predator y. The dynamics of the prey population is modelled by system (5.1). Again the two questions in Chapter 4 are raised. A partial answer

P. 1

to question (1) is provided by Theorem 6.5 which, modulo a number of technicalities, can be summarized as follows.

Assume that each of the four two-population subsystems constituting y and x_{ij} (i,j=1,3) contains a globally stable positive interior equilibrium. If there is a complete ordering of the predator functional responses: $P_{11} < P_{12} < P_{13} < P_{21} < P_{22} < P_{23} < P_{31} < P_{32} < P_{33}$, and, if the predator y can survive on the most difficult to catch prey genotype x_{11} , then the gamete types Ab, aB, and, ab go extinct. Hence only x_{11} and y survive.

This result is intuitively clear in view of the similar result for the one-locus case.

A partial answer to question (2) is provided by Theorem 6.6 which, modulo a number of technicalities, can be summarized as follows.

Assume that the predator y survives and that each of the four subsystems consisting of y and x_{11} (resp. x_{13} , x_{31} , x_{33}), contains a globally stable positive interior equilibrium $E_1 = (\overline{x}_{11}, 0, \dots, 0, \overline{y}_{11})$ (resp. $E_2 = (0, 0, \overline{x}_{13}, 0, \dots, 0, \overline{y}_{13})$, $E_3 = (0, \dots, 0, \overline{x}_{31}, 0, 0, \overline{y}_{31})$, $E_{33} = (0, \dots, 0, \overline{x}_{33}, \overline{y}_{33})$). Under each of

the following three cases,

(i)
$$P_{11} < P_{12} < P_{13} < P_{23} < P_{33}$$
, $P_{11} < P_{21} < P_{31} < P_{32} < P_{33}$
and $P_{22}(\overline{x}_{11}) < \frac{s}{k} - \frac{1}{2\overline{y}_{11}} B(\overline{x}_{11})$,

(ii)
$$P_{11} < P_{12} < P_{13}$$
, $P_{11} < P_{21} < P_{31}$, $P_{33} < P_{23} < P_{13}$
 $P_{33} < P_{32} < P_{31}$, $P_{22}(\overline{x}_{11}) < \frac{s}{k} - \frac{1}{2\overline{y}_{11}} B(\overline{x}_{11})$ and

$$\begin{array}{l} P_{22}(\overline{x}_{33}) < \frac{s}{k} - \frac{1}{2\overline{y}_{33}} B(\overline{x}_{33}) \ , \ \text{and}, \\ (\text{iii}) P_{11} < P_{12} < P_{13} < P_{23} < P_{33} < P_{32} < P_{31} \ , \ P_{11} < P_{21} < P_{31} \\ \text{and} \ P_{22}(\overline{x}_{11}) < \frac{s}{k} - \frac{1}{2\overline{y}_{11}} B(\overline{x}_{11}) \ , \end{array}$$

131

system (6.1) is persistent.

One could interpret these results as saying that if we consider the system as a one-locus problem, then the system is non-persistent; but as a two-locus problem, the predator functional response P_{22} of the double heterozygote x_{22} can change (by making x_{22} more difficult to catch) a non-persistent system to one of persistence.

BIBLIOGRAPHY

- Abraham, R., Robbin, J. (1967): Transversal mappings and flows. New York-Amsterdam: W.A. Benjamin Inc.
- Akin, E. (1982): Cycling in simple genetic systems. J. Math. Bio. 13, 305-324.
- Akin, E. (1983): Hopf bifurcation in the two locus genetic model. ^ (Memoirs of the A.M.S. 44 No. 284). Providence: Amer. Math. Soc.
- Akin, E., Hofbauer, J. (1982): Recurrence of the unfit. Math. Biosci. 61, 51-62.
- Andronov, A.A., Leontovich, E.A., Gordon, I.I., Maier, A.G. (1975): Qualitative theory of second-order dynamic systems. (Eng. Transl.) New York: John Wiley and Sons.
- Aronson, D.G., Weinberger, H.F. (1975): Nonlinear diffusion in population genetics, combustion and nerve propagation. In: Proceedings of the Tulane Program in partial differential equations and related topics. (Lecture Notes in Mathematics 446, pp. 5-49.) Berlin-Heidelberg-New York: Springer-Verlag.
- Aronson, D.G., Weinberger, H.F. (1978): Multidimensional nonlinear diffusion arising in population genetics. Advances in Math. 30, 33-76.
- Beck, K. (1982): A model of the population genetics of cystic fibrosis in the United States. Math. Biosci. 58, 243-257.
- Beck, K. (1984): Coevolution: mathematical analysis of host parasite interactions. J. Math. Bio. 19, 63-77.

- Beck, K., Keener, J.P., Ricciardi, P. (1982): Influence of infectious disease on the growth of a population with three genotypes. In: Population Biology, Proceedings, Edmonton (Freedman, H.I, Strobeck C., eds.). Lecture Notes in Biomathematics 52, pp. 416-422, Berlin-Heidelberg-New York: Springer-Verlag.
- Beck, K., Keener, J.P., Ricciardi, P. (1984): The effect of epidemics on genetic evolution. J. Math. Bio. 19, 79-94.
- Bhatia, N.P., Szegö, G.P. (1967): Dynamical systems: stability theory and applications. (Lecture Notes in Mathematics 35). Berlin-Heidelberg - New York: Springer-Verlag.
- Butler, G.J., Freedman, H.I., Waltman, P.E. (1982): Global dynamics of a selection model for the growth of a population with genotypic fertility differences. J. Math. Bio. 14, 25-35.
- Cheng, K.S., Hsu, S.B., Lin, S.S. (1981): Some results on global stability of a predator-prey system. J. Math. Bio. 12, 115-126.
- Chow, S.N., Hale, J.K. (1982): Methods of bifurcation theory. Berlin-Heidelberg - New York: Springer-Verlag.
- Christiansen, F.G., Fenchel, T.M. (1977): Theory of population in biological communities. (Ecological studies 20). Berlin-Heidelberg-New York: Springer-Verlag.
- Crow, J.F., Kimura, M. (1970): An introduction to population genetics theory. New York-Evanston-London: Harper & Row, Publ.
- Coddington, E.A., Levinson, N. (1955): Theory of ordinary differential equations. New York: McGraw Hill.
- Conley, C. (1975): An application of Wazewski's method to a non-linear boundary value problem which arises in population genetics. J. Math. Bio. 2, 241-249.
- Conley, C. (1978): Isolated invariant sets and the Morse index. (C.B.M.S. Res. Conf. Ser. Math. 38). Providence: Amer. Math. Soc.

Conley, C., Fife, P. (1982): Critical manifolds, travelling waves, and example from population genetics. J. Math. Bio. 14, 159-176.

- Coppel, W.A. (1965): Stability and asymptotic behavior of differential equations. Boston: D.C. Heath and Company.
- Cottin, L. (1983): Die genetische Algebra der einfachen plasmatischen Vererbung. Diplomarbeit. Universität Duisburg - Gesamthochschule.
- Ewens, W.J. (1979): Mathematical population genetics. (Bjomathematics 9). Berlin-Heidelberg - New York: Springer-Verlag.
- Freedman, H.I. (1976); Graphical stability, enrichment, and pest control by a natural enemy. Math. Biosci. 23, 127-149.
- Freedman, H.I. (1980): Deterministic mathematical models in population ecology. (Monographs and texbooks in pure and applied mathematics 57). New York-Basel: Marcel Dekker Inc.
- Freedman, H.I., Waltman, P.E. (1978): Predator influences on the growth of a population with three genotypes. J. Math. Bio. 6, 367-374.
- Freedman, H.I., Waltman, P.E. (1982): Predator influences on the growth of a population with three genotypes II. Rocky Mountain J. Math. 12 No. 4, 779-784.
- Freedman, H.I., Waltman, P.E. (1984): Persistence in models of three interacting predator-prey populations. Math. Biosci. 68, pp. 213-231.
- Freedman, H.I., So, J.W.H., Waltman, P. (1984): Predator influences on the growth of a population with three genotypes III. (Preprint).
- Gause, G.F. (1934): Struggle of existence. Baltimore: Williams and Williams.
- Gause, G.F., Smaragdova, N.P., Witt, A.A. (1936): Further studies of interaction between predators and prey. J. Anim. Ecol. 5, 1-18.

Gregorius, H.R., Ross, M.D. (1981): Selection in plant population of effectively infinite size: I. Realized genotypic fitness. Math. Biosci. 54, 291-307.

- 3

÷.Ŧ

Hadeler, K.P. (1981): Stable polynorphism in a selection model with mutation. SIAM J. Appl. Math. 41, 1-7.

- Hadeler, K.P., Glas, D. (1983): Quasimonotone systems and convergence to equilibrium in a population genetic model. J. Math. Anal. Appl. 95, 297-303.
- Hadeler, K,P., Liberman, U. (1975): Selection models with fertility differences. J. Math. Bio. 2, 19-32.
- Hale, J.K. (1969): Ordinary differential equations. (Pure and Applied Mathematics 21). New-York - London - Sydney - Toronto: Wiley-Interscience.
- Hartl, D.L. (1981): A primer of population genetics. Sunderland: Sinauer Associates Inc.

Hartman, P. (1982): Ordinary differential equations. (Second edition). Boston-Basel-Stuttgart: Birkhauser.

- Hirsch, M.W., Pugh, C.C., Shub, M. (1977): Invariant manifolds. (Lecture Notes in Mathematics 583). Berlin-Heidelberg-New-York: Springer-Verlag.
- Hoppensteadt, F.C. (1975): Analysis of a stable polymorphism arising in a selection-migration model in population genetics. J. Math. Bio. 2, 235-240.
- Hsu, S.B. (1978): On global stability of a predator-prey system. Math. Biosci. 39, 1-10.
- Hunt, F. (1980): On the persistence of spatially homogeneous solutions of a population genetics model with slow selection. Math. Biosci. 52, 185-206.

135

Irwin, M.C. (1980): Smooth dynamical systems. New York: Academic Press.

- Jacquard, A. (1974): The genetic structure of populations. Berlin-Heidelberg - New York: Springer-Verlag.
- Karlin, S. (1975): General two-locus selection models: some objectives, results and interpretations. Theo. Pop. Bio. 7, 364-398.
- Karlin, S. (1980): The number of stable equilibria for the classical onelocus multiallele selection model. J. Math. Bio. 9, 189-192.
- Karlin, S., Carmelli, D. (1975a): Some population genetic models combining artificial and natural selection pressures: II. Two-locus theory. Theo. Pop. Bio. 7, 123-148.
- Karlin, S., Carmelli, D. (1975b): Numerical studies on two-loci selection models with general viabilities. Theo. Pop. Bio. 7, 399-421.
- Kimura, M. (1958): On the change of population fitness by natural selection. Heredity 12, 145-167.
- Losert, V., Akin, E. (1983): Dynamics of games and genes: discrete versus continuous time. J. Math. Bio. 17, 241-251.
- LaSalle, J.P. (1976): The stability of dynamical systems. (Reg. Conf. Ser. Appl. Math. 25). Philadelphia: S.I.A.M.
- McGehee, R., Armstrong, R.A. (1977): Some mathematical problems concerning the ecological principle of competitive exclusion. J. Diff. Equ. 23, 30-52.
- Nagylaki, T. (1974): Continuous selection models with mutation and migration. Theo. Pop. Bio. 5, 284-295.
- Nagylaki, T. (1977): Selection in one- and two-locus systems. (Lecture Notes in Biomathematics 15). Berlin-Heidelberg-New York: Springer-Verlag.
- Nagylaki, T., Crow, J.F. (1974): Continuous selection models. J. Theo. Pop. Bio. 5, 257-283.

Nemytskii, V.V., Stepanov, V.V. (1960): Qualitative theory of differential equations. Princeton: Princeton Univ. Press.

- Palis, J., Takens, F. (1977): Toplogical equivalence of normally hyperbolic dynamical systems. Topology 16, pp. 335-345.
- Rosenweig, M.L., MacArthur, R.H. (1963): Graphical representation and stability conditions of predator-prey interactions. Amer. Nat. 47, 209-223.
- Roughgarden, J. (1979): Theory of population genetics and evolutionary ecology: an introduction. New York: MacMillan.
- Shahshahani, S. (1979): A new mathematical framework for the study of linkage and selection. (Memoir of the A.M.S. 17 No. 211). Providence: Amer. Math. Soc.
- Sell, G.R. (1971): Topological dynamics and ordinary differential equations. London: Van Nostrand Reinhold.
- So, J.W.-H. (1978): A study of predator-prey food chains. M.Sc. Thesis. University of Alberta.
- So, J.W.-H. (1978): The effect of evolution on the abundance of species in a food chain. Notices of the A.M.S. 25, #6, issue 188, A-622, 759-C11.
- So, J.W.-H. (1979): A note on the global stability and bifurcation phenomenon of a Lotka-Volterra food chain. J. Theo. Bio. 80, 185-187.
- So, J.W.-H. (1984): Analysis of a continuous one-locus, two-allele genetic model with genotypic fertility differences and non-linear birth and death processes. (Preprint.).
- So, J.W.-H., Freedman, H.I. (1984): Persistence and global stability in a predator-prey model consisting of three prey genotypes with genotypic fertility differences. (Preprint.)
- Waltman, P. (1984): Competition models in population biology. (Reg. Conf. Ser. Appl. Math. 45). Philadelphia: S.I.A.M.

Ĩ

1

APPENDIX 1

DERIVATION OF THE ONE-LOCUS TWO-ALLELE MODEL (3.1)

If we denote the two alleles at the locus by A and a, then there are three genotypes: AA, Aa, and, aa. The mating table looks like:

Mating Type	AA	Aa	aa	
AA × AA	1	-		
AA × Aa 🖉	`1/2	1/2	-	
AA × aa]	_	
Aa × Aa	1/4	1/2	1/4	
Aa×aa		1/2	1/2	
aa × aa	-	-	1	

There are two ways of arriving at the coefficients of B(x) in system (3.1).

(i) f_i's as fertilities:

 f_i (i = 1,2,3) is to be thought of as the number of gametes produced by each individual in the x_i population. For example, there will be $\frac{1}{2}f_2x_2$. A gametes and $\frac{1}{2}f_2x_2$ a gametes produced by the x_2 population. Thus, there are

÷.,

 $f_1x_1 + \frac{1}{2}f_2x_2 = u$ that many A gametes and $f_3x_3 + \frac{1}{2}f_2x_2 = v$ that many a gametes in the gamete pool. By random union of gametes, the proportion of AA genotype produced is: $(\frac{u}{u+v})(\frac{v}{u+v}) = \frac{u^2}{(u+v)^2}$. Similarly one can arrive at the other coefficients in system (3.1) this way.

The fraction $\frac{f_i x_i}{f_1 x_1 + f_2 x_2 + f_3 x_3}$ (i = 1,2,3) is to be thought of as the proportion of x_i that involves in random mating. Since the AA genotype is obtained from the mating types: AA×AA, AA×Aa, Aa×AA, and, Aa×Aa with probabilities: 1,1/2, 1/2, and, 1/4 respectively, we have

$$(\frac{f_{1}x_{1}}{f_{1}x_{1} + f_{2}x_{2} + f_{3}x_{3}})(\frac{f_{1}x_{1}}{f_{1}x_{1} + f_{2}x_{2} + f_{3}x_{3}})$$

$$+ \frac{1}{2}(\frac{f_{1}x_{1}}{f_{1}x_{1} + f_{2}x_{2} + f_{3}x_{3}})(\frac{f_{2}x_{2}}{f_{1}x_{1} + f_{2}x_{2} + f_{3}x_{3}})$$

$$+ \frac{1}{2}(\frac{f_{2}x_{2}}{f_{1}x_{1} + f_{2}x_{2} + f_{3}x_{3}})(\frac{f_{1}x_{1}}{f_{1}x_{1} + f_{2}x_{2} + f_{3}x_{3}})$$

$$+ \frac{1}{4}(\frac{f_{2}x_{2}}{f_{1}x_{1} + f_{2}x_{2} + f_{3}x_{3}})(\frac{f_{2}x_{2}}{f_{1}x_{1} + f_{2}x_{2} + f_{3}x_{3}}) = \frac{u^{2}}{(u+v)^{2}}$$

as the proportion of AA genotypes produced. The other two coefficients of B(x) in system (3.1) are arrived at in a similar way.

APPENDIX 2

DERIVATION OF THE TWO-LOCUS TWO-ALLELE MODEL (5.1)

The derivation is similar to the one-locus case in Appendix 1. a denote the two allele types at the first locus by A and a and the two allele types at the second locus by B and b. There are nine genotypes: AABB, AABb, AAbb, AaBB, AaBb, Aabb, aaBB, aaBb, and, aabb, the number of which we denote by $x_{11}, x_{12}, \dots, x_{33}$ respectively. Therefore the number of AB (resp. Ab, aB, ab) gametes in gamet, pool given by fu_1 (resp. fu_2, fu_3, u_4) where the u_1 's were defined in (5.2) and f is the common fertility of the genotypes. The total number of nametes in the gamete pool is given by $fu_1 + fu_2 + fu_3 + fu_4 = fx$. By necess union of gametes, the proportion of AABB genotype produced is: $(\frac{fu_1}{fx})(\frac{fu_1}{fx}) = \frac{u_1^2}{x^2}$. Similarly one can arrive at the other coefficients of B(x) in system (5.1) this way.

The alternative derivation given in Appendix 1 can also be carried out. Since it is lengthy but similar, we will not write it down here.